Institutional Investors and Corporate Governance

Amil Dasgupta Vyacheslav Fos Zacharias Sautner

This survey was prepared by invitation for Foundations and Trends in Finance. We are grateful to Zac Rolnik (Publisher), Sheridan Titman (Editor-in-Chief), Chester Spatt (Associate Editor), Julian Franks (Referee), Brian Bushee, Adrian Corum, Alex Edmans, Jeffrey N. Gordon, Wei Jiang, Doron Levit, Nadya Malenko, Rich Mathews, John Morley, Ed Rock, Martin Schmalz, Miriam Schwartz-Ziv, Amy Wilson, and participants in the ECGI Spotlight Webinar for constructive comments. We thank Marc Gischer for excellent research assistance and Catherine Perry for editorial help. We are grateful to Miguel Ferreira and Pedro Matos for sharing their data on institutional ownership around the world. Dasgupta acknowledges financial support from the ESRC via Research Grant ES/S016686/1. Data on institutional blockholders (1%-, 3%-, 5%-, or 10%-blocks, blocks by the largest five institutional blockholders, and Big Three blockholdings) are available upon request from Vyacheslav Fos (fos@bc.edu).

 \bigcirc Amil Dasgupta, Vyacheslav Fos and Zacharias Sautner 2021. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including \bigcirc notice, is given to the source.

Abstract

We provide a comprehensive overview of the role of institutional investors in corporate governance with three main components. First, we establish new stylized facts documenting the evolution and importance of institutional ownership. Second, we provide a detailed characterization of key aspects of the legal and regulatory setting within which institutional investors govern portfolio firms. Third, we synthesize the evolving response of the recent theoretical and empirical academic literature in finance to the emergence of institutional investors in corporate governance. We highlight how the defining aspect of institutional investors – the fact that they are financial intermediaries – differentiates them in their governance role from standard principal blockholders. Further, not all institutional investors are identical, and we pay close attention to heterogeneity amongst institutional investors.

Keywords: Institutional investors, corporate governance, exit, voice, shareholder activism, proxy voting advisors

JEL Classifications: G11, G15, G22, G23

Amil Dasgupta*

Professor of Finance London School of Economics and Political Science, Department of Finance Houghton Street London, WC2A 2AE, United Kingdom phone: +44 207 955 7458 e-mail: a.dasgupta@lse.ac.uk

Vyacheslav Fos

Associate Professor of Finance, Hillenbrand Family Faculty Fellow Boston College, Carroll School of Management Fulton Hall, 140 Commonwealth Avenue Chestnut Hill, MA 02467, United States phone: +1 617 552 1536 e-mail: fos@bc.edu

Zacharias Sautner

Professor of Finance Frankfurt School of Finance and Management, Finance Department Adickesallee 32-34 60322 Frankfurt am Main, Germany phone: +49 69 154008 755 e-mail: z.sautner@fs.de

*Corresponding Author

Institutional Investors and Corporate Governance*

Amil Dasgupta, Vyacheslav Fos, Zacharias Sautner

July 5, 2021

Forthcoming in Foundations and Trends in Finance

Abstract

We provide a comprehensive overview of the role of institutional investors in corporate governance with three main components. First, we establish new stylized facts documenting the evolution and importance of institutional ownership. Second, we provide a detailed characterization of key aspects of the legal and regulatory setting within which institutional investors govern portfolio firms. Third, we synthesize the evolving response of the recent theoretical and empirical academic literature in finance to the emergence of institutional investors in corporate governance. We highlight how the defining aspect of institutional investors – the fact that they are financial intermediaries – differentiates them in their governance role from standard principal blockholders. Further, not all institutional investors are identical, and we pay close attention to heterogeneity amongst institutional investors as blockholders.

Keywords: Institutional investors, corporate governance, exit, voice, shareholder activism, proxy voting advisors JEL Codes: G11, G15, G22, G23

* Amil Dasgupta (a.dasgupta@lse.ac.uk) is at the London School of Economics and Political Science, Vyacheslav Fos (fos@bc.edu) is at Boston College, and Zacharias Sautner (z.sautner@fs.de) is at the Frankfurt School of Finance & Management. This survey was prepared by invitation for *Foundations and Trends in Finance*. We are grateful to Zac Rolnik (Publisher), Sheridan Titman (Editor-in-Chief), Chester Spatt (Associate Editor), Julian Franks (Referee), Brian Bushee, Adrian Corum, Alex Edmans, Jeffrey N. Gordon, Wei Jiang, Doron Levit, Nadya Malenko, Rich Mathews, John Morley, Ed Rock, Martin Schmalz, Miriam Schwartz-Ziv, Amy Wilson, and participants in the ECGI Spotlight Webinar for constructive comments. We thank Marc Gischer for excellent research assistance and Catherine Perry for editorial help. We are grateful to Miguel Ferreira and Pedro Matos for sharing their data on institutional ownership around the world. Dasgupta acknowledges financial support from the ESRC via Research Grant ES/S016686/1. Data on institutional blockholders (1%-, 3%-, 5%-, or 10%-blocks, blocks by the largest five institutional blockholders, and Big Three blockholdings) are available upon request from Vyacheslav Fos (fos@bc.edu).

Outline

| 1. Introduction | 4 |
|--|--|
| 2. Stylized Facts on the Importance of Institutional Investors | 7 |
| 2.1 Importance of Institutional Investors in the US | 7 9 |
| 3. Legal Environment: Obligations, Ability, and Incentives of Institutional Investors to Govern | 12 |
| 3.1 Legal Obligations of Institutional Investors to Govern | 14 17 19 19 19 22 23 24 25 25 27 |
| 4. Theoretical Literature on Institutional Investors and Corporate Governance | 32 |
| 4.1 Classical Theoretical Literature on Blockholder Governance 4.2 Theories Centered on Institution-Specific Ability 4.3 Theories of Blockholders as Agents 4.3.1 Dual-Layered Agency Structures 4.3.2 Dual-Layered Agency Models and the Exit Mechanism 4.3.3 Dual-Layered Agency Models and Short-Termism 4.3.4 Dual-Layered Agency Models and Multiple (Interacting) Blockholders 4.3.5 Active and Passive Management 4.4 Theories on Institutional Investors and Proxy Voting Advisors | 34 36 36 38 39 40 41 |
| 5. Empirical Literature on Institutional Investors and Corporate Governance | 44 |
| 5.1 Institutional Investor Data 5.1.1 Archival Data on Institutional Investors 5.1.2 Case Study Data for Specific Institutional Investors | 45 |

| References | 77 |
|---|----|
| 6. Conclusions | 73 |
| 5.5.5 Conflicts of Interest | 73 |
| 5.5.4 Firm and Market Reactions to Voting Recommendations | 72 |
| 5.5.3 Determinants and Quality of Voting Recommendations | 71 |
| 5.5.2 Influence of Voting Recommendations | 69 |
| 5.5.1 Importance of Proxy Voting Advisors | 67 |
| 5.5 Proxy Voting Advisors and Institutional Investors | 67 |
| 5.4.7 Institutional Investor Ideology | 67 |
| 5.4.6 Institutional Investor Location | |
| 5.4.5 Interplay between Institutional Investors | 64 |
| 5.4.4 Passive versus Active Institutional Investors | |
| 5.4.3 Long-term versus Short-term Institutional Investors | |
| 5.4.2 Conflicted versus Non-Conflicted Institutional Investors | |
| 5.4.1 Institutional Investor Governance Incentives | |
| 5.4 Institutional Investor Heterogeneity | |
| 5.3.2 Evidence on Governance via Exit | |
| 5.3.1 Evidence on Governance via Voice | |
| 5.3 Institutional Investors as Blockholders: Classical Evidence on Voice and Exit | |
| 5.2.4 Unexpected Mutual Fund Outflows | |
| 5.2.3 Index Decompositions | |
| 5.2.2 Discontinuities in Voting Outcomes | |
| 5.2.1 Activism Campaign Announcements | |
| 5.2 Empirical Methods to Identify Governance Effects of Institutional Investors | |
| 5.1.3 Survey Data on Institutional Investors | |

1. Introduction

Whenever the ownership of public corporations is dispersed amongst many shareholders, blockholders – owners of non-trivial percentages of a firm's shares – become central to good corporate governance. In contrast to small shareholders, who have neither the incentive nor the capacity to effectively monitor management, blockholders are able to govern firms to the benefit of all. The governance role of blockholders today must be viewed in the backdrop of the explosive growth of the asset management industry in recent decades, which has led to the large-scale *intermediation* of equity ownership.

Table 1 shows that 50 years ago households *directly* owned almost 80% of US corporate equity. Such direct ownership has declined dramatically over the years, reducing by more than a half, so that today only 38.3% of US corporate equity is directly owned by households. The remainder is *indirectly* held via different asset managers – commonly referred to as *institutional investors*. The table shows that such institutional ownership is dominated by four major types of domestic investors: Mutual funds (20.8%), exchange traded funds (ETFs) (6.6%), public pension funds (5.3%), and private pension funds (5.4%). The remaining institutional investors include insurance companies (1.9%), which have become progressively smaller over time, and "other" unclassified domestic investors (5.1%), a catch-all category including hedge funds or the proprietary holdings of financial institutions. Finally, 16.4% is held by foreign institutional investors. This phenomenon is not limited to the US: below we provide evidence of a similarly significant growth of institutional ownership in other major economies around the world.

[Insert Table 1 here]

Institutional investors are different from the standard blockholders of the classical corporate governance literature in a number of ways. They are larger than most private investors, often subject to extensive regulations, and – perhaps most fundamentally – they differ from private blockholders because

¹ The measure of foreign ownership does not allow the separation between direct investments by households and institutional investors. However, the vast majority of this category likely originates from foreign hedge funds, pension funds, mutual funds, and sovereign wealth funds.

they invest other people's money. Given the preponderance of institutional investors in corporate equity ownership, it is important to understand the role they play in corporate governance.

In this review, we provide a comprehensive overview of the role of institutional investors in corporate governance. Our contribution has three main components. First, we trace the emergence of institutional investors as the modal concentrated owners of public firms in modern economies, using a wide variety of data sources to establish new stylized facts. Second, we provide a detailed characterization of key aspects of the legal and regulatory setting within which institutional investors operate with respect to the governance of their portfolio firms. Third, we synthesize the evolving response of the academic literature in finance to the emergence of institutional investors in corporate governance, attempting to link theoretical predictions to empirical findings.

Because of our focus on institutional investors as the holders of equity blocks we highlight the role that characteristics specific to this type of equity blockholder play in corporate governance. For example, we document how the defining aspect of institutional investors – the fact that they are financial intermediaries – differentiates them in their governance role from standard principal blockholders (e.g., individuals, families, and firms). As a result, we discuss how differences with respect to explicit and implicit incentives, organizational structures, and regulatory requirements shape their obligations, incentives and ability to govern. This focus leads us to highlight aspects of governance that are unique to institutional investors, for example, the institutions' voting processes, the role of proxy voting advisors, and conflicts of interests arising from business ties with portfolio firms.² Further, not all institutional investors are identical, and we pay close attention to *heterogeneity* amongst institutional investors as blockholders, arising because of differences in their incentives, size, investment horizons, preferred governance mechanisms, or regulatory constraints. We shine a light on heterogeneity across institution types by establishing stylized facts on ownership heterogeneity, and by synthesizing the emerging lessons from the theoretical and empirical literatures on the impact of such heterogeneity.

² Given our specific focus, we do not review the literature on the role of non-institutional blockholders, such as families or other corporations, which also play an important role in the governance of firms. While some aspects in our discussions also apply to these blockholders, notably the broader theories and the evidence on monitoring by institutional blockholders, most aspects in our review are specific to institutional blockholders.

Hence, our survey differs in two important ways from recent surveys that look at the role of blockholders *per se* in corporate governance (e.g., Edmans, 2014; Edmans and Holderness, 2017). First, we highlight governance issues specific to institutional investors, which can lead to substantial differences between the objectives of canonical blockholders who maximize the value of their blocks and the objectives of institutional investors. For instance, a labor union pension fund that manages pension accounts of a firm's employees might consider negative effects of a value increasing action on the employees and thus resist the action. Second, the main objective of a rising class of passive institutional investors is to not maximize the value of their investment but to rather track an index. For instance, with trillions of dollars in assets, the objective of most ETFs is to track the performance of a basket of securities. Investment companies managing ETFs are, however, required to vote in the best interest of their investment objectives.³

Though we do not focus on the role of non-institutional blockholders, such as families, wealthy individuals, or other corporations, we should note that these equity blockholders also play an important role in the governance of firms, both by themselves and in their interactions with institutional investors. Notably, in publicly traded firms Germany or even the US (outside of the largest firms), family ownership remains a significant governance factor. Villalonga *et al.* (2015) review the literature on corporate governance in family-owned firms.

A few other recent surveys cover related issues. Yermack (2010) surveys the literature on shareholder voting, while we in contrast delve in detail on institution-specific aspects of proxy voting. Brav *et al.* (2010) provides a survey of activist hedge funds specifically. While activist hedge funds feature prominently in our review, they are only one part of the much wider landscape of the role of institutional investors in corporate governance that is of interest to us. Schmalz (2018) surveys the emerging literature on common ownership, a specific topic tied to the existence of large institutional investors, which we only touch on briefly below. Institutional investors are sometimes under scrutiny for their role in environmental and social issues tied to corporate actions. In a recent review, Matos (2020) focuses specifically at this

³ Despite our different focus, we pick up on key themes highlighted in Edmans and Holderness (2017), namely the role of blockholder heterogeneity and evidence from institutional settings beyond the US.

aspect of the role of institutional investors. Finally, Franks (2020) provides a contemporaneous review of institutional ownership around the world and discusses topics within their governance role.

The remainder of this survey is structured as follows. Section 2 provides a series of new stylized facts on the evolution of institutional ownership and its heterogeneity in the US and outside of the US. Section 3 describes the legal and regulatory environment within which institutional investors operate, with a focus on the obligations, ability, and incentives of such investors to engage in the corporate governance of firms. Section 4 reviews the theoretical literature on institutional investors and corporate governance while Section 5 discusses the empirical literature. Section 6 concludes.

2. Stylized Facts on the Importance of Institutional Investors

In this section, we present new stylized facts about the emergence and evolution of institutional investors as blockholders in firms. In Section 2.1 we characterize institutional ownership in the US. We report statistics on the cross-sectional and time-series variation of concentrated institutional stakes, measured at different block sizes, and document how the distribution of such stakes varies across key types of institutions. Following our detailed characterization of the US data, in Section 2.2 we trace the evolution of institutional ownership in China, Japan, Germany, the UK, and India, the five largest economies other than the US.⁴

2.1 Importance of Institutional Investors in the US

2.1.1 Data on Institutional Blockholdings

We rely on two sources of data to produce stylized facts on block ownership by institutional investors in the US. First, we use quarterly stock ownership data from Schedule 13F filings, which are required by the SEC for investment managers with over \$100 million. Rather than utilizing the commonly used Thomson Reuters S34 database, we rely on the Schedule 13F dataset compiled by Backus *et al.* (2021). This dataset covers the years from 1999 to 2017 and arguably has better coverage than the

⁴ Our classification follows from the GDP data provided by the IMF World Economic Outlook Database, April 2019, available at <u>https://www.imf.org/external/pubs/ft/weo/2019/01/weodata/index.aspx</u>. We use the rankings from 2017, as that is the most recent year from which confirmed data is available for the full set of top-ten economies at the time of writing.

Thomson Reuters database (Backus *et al.*, 2021). In characterizing US institutional ownership, we pay particular attention to two key types of institutions, namely mutual funds (the largest single type of institutional investor) and hedge funds (arguably the type of institutional investor that is most active with respect to corporate governance). Further detailed discussion of the key role of these two types of institutions in corporate governance is provided throughout the paper, especially in Sections 3, 4, and 5.

To analyze hedge fund ownership, we merge the 13F dataset with the list of Schedule 13F filers identified as hedge funds by Agarwal *et al.* (2013). This list includes about 1,200 hedge fund management companies. Thus, we use the schedule 13F dataset to produce stylized facts on block ownership for all institutional shareholders as well as on blocks held by hedge funds in particular.

To analyze mutual fund ownership, in a separate analysis, we use the CRSP Survivor-Bias-Free US Mutual Fund Database, which includes a monthly history of each mutual fund's holdings. We use this database – rather than Schedule 13F filings – to construct blockholding for mutual funds because Thomson Reuters's institution classifications are often unreliable. Data on security prices and shares outstanding are obtained from the Center for Research in Security Prices (CRSP).

In our analysis, we focus on four institutional block sizes: 1%-blocks, 3%-blocks, 5%-blocks, and 10%-blocks. An institution is set to hold an X%-block if it holds *at least* X% of a security's shares outstanding. The literature has so far paid relatively little attention to blocks of 1% and 3%, focusing instead on 5%- or 10%-blocks. The reason is that crossing the 5% ownership threshold triggers a Schedule 13D or 13G disclosure, while owners of 10%-blocks become corporate insiders and are subject to the disclosure rules of the insider trading regulation of the *Securities Exchange Act of 1934* (Section 16). We explore blocks of 1% and 3% as such relatively small percentage blocks can require large capital commitments at firms with large market capitalizations. Further, such small blocks are important for several governance mechanisms relevant to institutional investors (e.g., such smaller blocks are arguably most relevant to governance via exit). Finally, given the emerging evidence that multiple institutional investors engage in parallel in target firms (see Section 5.4.5 for empirical evidence and Section 4.3.4 for theoretical foundations), such small blocks may collectively play a key role in achieving governance outcomes. Stylized facts about such smaller blocks therefore fill an important gap in the literature.

2.1.2 Results on Institutional Blockholdings

Table 2, Panel A, reports summary statistics on the number of institutional blockholders.⁵ The unit of observations in the panel is the security-reporting quarter. We find that the average (median) number of institutions with a 1%-block *in a given security* is 11.2 (9.0); when we consider 3%-blocks, these statistics are 4.3 and 3.0, respectively. These figures show that a typical firm has several blockholders with the potential to play an active governance role. When we consider large blocks, the average number of 5%-blocks (10%-blocks) is 2.4 (0.9), indicating that the average firm has between one and two large institutional blockholders, depending on the definition of "large."

[Insert Table 2 here]

In Table 2, Panel B, we report the same statistics for blocks held by hedge funds. For such institutions, the average number of 1%- and 3%-blocks is 2.6 and 1.2, respectively, indicating that the average firm has a hedge fund blockholder in its shareholder base. There is substantial variation in the number of hedge fund blocks across firms. For instance, whereas the 10th percentile for 5%-blocks is zero, the 90th percentile of 3%- and 5%-blocks is 3.0 and 2.0, suggesting that a meaningful fraction of firms has several large hedge fund blockholders.

In Table 2, Panels C and D, we turn our attention to mutual funds. While Panel C reports statistics at the mutual fund *company level*, implying that the figures are directly comparable to those in Panels A and B, Panel D instead considers blocks at the *mutual fund level*. The distinction between these two levels of measurement for mutual fund ownership is important: while portfolio investment and selling decisions are typically made at the fund level, corporate governance decisions (e.g., how to vote) are usually made at the mutual fund company level.

In Panel C, the average number of 1%- and 3%-blocks for mutual fund companies is 2.5 and 0.6. Hence, relative to hedge fund blockholders, firms have fewer mutual fund blockholders in their shareholder base. For instance, the average number of hedge fund 5%-blocks is 0.7, while the average number of mutual fund 5%-blocks is 0.3. The results also indicate that, on average, firm have both hedge

⁵ Data on institutional blockholders reported in the table (as well as data on blockholdings by the largest five institutional investors and the Big Three are available upon request from Vyacheslav Fos (<u>fos@bc.edu</u>).

fund and mutual fund blockholders in their shareholder base. For instance, the average number of 1%blocks for mutual fund companies is 2.5 and for hedge funds it is 2.6 – thus, it is likely that there is a considerable level of interaction between different types of institutional investors when important corporate governance decisions are at stake. The evidence in Panel D indicates that, on average, a firm has more than two 1% mutual fund blockholders. Further, a typical firm has one 3% hedge fund blockholder but no 3% mutual fund blockholder. This difference is consistent with investment restrictions that prevent individual mutual funds from investing large fractions of their capital in just one security.

We next present a sequence of figures that document the time trends in concentrated institutional ownership. In Figure 1, we document a steady increase in the fraction of firms with at least one X%-block held by an institutional investor, with X% denoting again a block of 1%, 3%, 5%, and 10%, respectively. The increase in institutional blockholdings is independent of how we define block sizes. For instance, the percentage of firms with at least one 3%-block owned by an institutional investor increased from 65% in 1999 to more than 90% in 2017. Similarly, the fraction of firms with at least one 5%-block rose from 50% in 1999 to almost 80% in 2017.

[Insert Figure 1 here]

Figure 2 shows time-series evolution of the fraction of firms with at least one block held by a hedge fund. Similar to the evidence on the population of institutional investors in the previous figure, hedge funds blocks became more frequent during last two decades. For example, the fraction of firms with at least one hedge fund 3%-block increased from about 25% in 1999 to almost 70% in 2017.

[Insert Figure 2 here]

Figure 3 shows the fraction of firms with at least one block held by a mutual fund. Interestingly, whereas the frequency of mutual fund blocks increased during the first half of the sample, the fraction of firms with mutual fund blocks has been stable during the second half. For instance, the fraction of firms with at least one 3%-block increased from 25% in 1999 to about 50% in 2009 and remained stable since then.

[Insert Figure 3 here]

Next, Figure 4 shows the percentage of shares outstanding owned by the five largest institutional investors in a firm. The figure shows that the fraction of shares owned by the top-five institutional shareholders has increased during our sample period. Specifically, between 1999 and 2017, the fraction of shares outstanding held by the five largest institutional investors increased from about 20% in 1999 to about 35% in 2017.

[Insert Figure 4 here]

Finally, Figure 5 shows the fraction of shares owned by the three big passive fund families ("Big Three"), that is, by BlackRock, State Street, and Vanguard. The figure shows that ownership by these institutions has increased strongly during our sample period. Specifically, the fraction of shares outstanding held by increased from about less than 1% in 1999 to more than 14% in 2017.

[Insert Figure 5 here]

2.2 Importance of Institutional Investors outside the US

We next turn to the evolution of institutional ownership in firms in China, Japan, Germany, the UK, and India, the five largest economies after the US. In Figure 6, we report the percentage institutional ownership for different country-years over the period 2005 to 2017 (from 2009 onwards for China due to data quality issues). The figure is constructed based on data from FactSet Ownership (previously FactSet/LionShares).

[Insert Figure 6 here]

A few important observations emerge from the figure. First, across all five countries, institutional ownership has strongly increased over the sample period, which is consistent with the US evidence in the previous section (China experienced some small decline in institutional ownership since about 2009). Second, the overall increase in institutional ownership was largest in the UK. Third, at the end of the sample period, institutional ownership was highest in the UK (42%), followed by Germany (29.4%), Japan and India (about 20% each), as well as China (about 10%).

In Figure 7, we report the evolution of *foreign* institutional ownership across the same five countries (we discuss evidence on the governance role of foreign investors in Section 5.4.6). The figure

shows that foreign institutional ownership also trended upwards over the sample period. Further, at the end of the sample period, foreign institutional ownership was highest among UK firms (27.4%), followed by firms in Germany (23%), Japan (14%), India (13%), and China (only 8%).

[Insert Figure 7 here]

Combining information from the previous two figures, Figure 8 reports the time-series development of the *fraction* of foreign institutional ownership (relative to total institutional ownership). An interesting observation that emerges is that the fraction of foreign ownership remained relatively constant in Japan and the UK. In contrast, Germany and especially China experienced a steady increase in the fraction of foreign institutional owners, while India saw a decline. Relative to total institutional ownership, foreign ownership at the end of the sample period amounts to 80% in Germany, 76% in China, 70% in Japan, and 65% in India and the UK, respectively.

[Insert Figure 8 here]

3. Legal Environment: Obligations, Ability, and Incentives of Institutional Investors to Govern

The governance role of institutional investors is affected by the legal framework under which they conduct their business activities, which determines their *obligation* and *ability* to govern portfolio firms. This legal framework also has implications for *incentive structures* permissible to institutional investors that determine their willingness to govern. In this section, we discuss these aspects of the institutional investor landscape. Further, we discuss some of the "how" of corporate governance actions by institutional investors – in particular, the manner in which institutional voice is determined.

Table 1 demonstrates that four types of institutions – mutual funds (20.8%), ETFs (6.6%), and public (5.3%) and private (5.4%) pension funds – represent the bulk of institutional ownership. Accordingly, we organize the discussion of the institutional setting around these four groups of investors. However, throughout we also devote special attention to hedge funds, because *activist* hedge funds – while representing a much smaller ownership stake in comparison to the four largest types of institutions – have wielded a disproportionate influence on governance in the recent two decades. Thus, understanding the

extent to which the legal framework applies to hedge funds is also of relevance. While we devote the greatest amount of space to the US, the largest market for institutional investors and the area in which the literature is farthest developed, we also provide a comparative perspective for some of the countries for which we reported stylized facts in Section 2.

Our focus will be both on what distinguishes across institutional investors and what is common amongst them. Different types of institutional investors are often governed by different legal frameworks, imposing different legal obligations on them. For example, in the US, mutual funds, closed end funds, and the vast majority of ETFs are organized as so-called registered investment companies and governed by the *Investment Companies Act of 1940* (ICA). Such investment companies employ portfolio managers to make investment decisions, and the latter are governed in the US by the *Investment Advisers Act of 1940* (IAA). In contrast, private pension funds are subject to the *Employee Retirement Income Security Act of 1974* (ERISA). While pension funds invest significantly in mutual funds (see Figure 9), ERISA Section 3, paragraph 21(B) states that the investment by a pension plan into an investment company governed by the ICA does not render any ERISA fiduciary obligation on the investment company or its investment adviser.⁶ Thus, private pension funds and mutual funds are governed by distinct laws. Public pension funds are subject to neither ERISA nor the ICA and IAA. Hedge funds are organized to be exempt from the ICA but, following the passage of the 2010 Dodd Frank Act, are subject to some of the requirements of the IAA.⁷ We discuss the governance obligations of institutional investors in Section 3.1.

[Insert Figure 9 here]

In contrast, commonality arises from their shared identity as equity blockholders. In the US, for example, commonality stems from the unified legal framework in the *Securities Exchange Act of 1934* (SEA) and in the relevant state laws determining the ability of such parties to govern firms. We discuss the governance ability of institutional investors in Section 3.2.

Incentives to govern again vary significantly across institutional investors. While the vast majority of mutual funds charge flat assets under management fees (Elton *et al.*, 2003), hedge funds often charge

⁶ Pension funds may either directly invest in a mutual fund product or hire an investment company to invest the money on its behalf and to vote the shares according to specific instructions.

⁷ Formally, the advisers of hedge funds (i.e., the hedge fund managers) are governed by the IAA, not the hedge funds themselves. Some advisers were already subject to the IAA before the Dodd-Frank Act.

complex fees involving performance fees and watermarks (Fung and Hsieh, 1999). Public pension plans, on the other hand, are typically overseen by boards that may be politically motivated (Andonov *et al.*, 2018; Woidtke; 2002) and may have different incentives as a result. We discuss governance incentives in Section 3.3.

3.1 Legal Obligations of Institutional Investors to Govern

3.1.1 Obligation to Govern in the US

The legal obligations of institutional investors with respect to the governance of their portfolio firms arise from a variety of sources. For US mutual funds and their investment advisers, the basic obligation to govern arises from state law that pertains to each fund's incorporation.⁸ As is succinctly described by the SEC (SEC, 2003a, p. 3):

"Mutual funds are formed as corporations or business trusts under state law and, [...], must be operated for the benefit of their shareholders. Because a mutual fund is the beneficial owner of its portfolio securities, the fund's board of directors, acting on the fund's behalf, has the right and the obligation to vote proxies relating to the fund's portfolio securities. As a practical matter, however, the board typically delegates this function to the fund's investment adviser as part of the adviser's general management of fund assets, subject to the board's continuing oversight. The investment adviser to a mutual fund is a fiduciary that owes the fund a duty of "utmost good faith, and full and fair disclosure." This fiduciary duty extends to all functions undertaken on the fund's behalf, including the voting of proxies relating to the fund's portfolio securities. An investment adviser voting proxies on behalf of a fund, therefore, must do so in a manner consistent with the best interests of the fund and its shareholders."

Thus, mutual funds and their investment advisers are obliged to use shareholder voice, that is, to vote their proxies in a manner that is deemed beneficial to the value of the funds. In recent decades, there has been a significant expansion of the obligations of mutual funds and their advisors with respect to the *transparent* and *unconflicted* use of shareholder voice. In particular, the obligations of mutual funds and their advisors of mutual funds and their advisors now extend, under the ICA and the IAA, to:

1. The public disclosure of proxy voting policies.

⁸ It has long been recognized that the relationship between the mutual funds and their investment advisors is highly intertwined, see, e.g., Schiffman (1976).

- 2. The reporting of votes actually cast either publicly (for mutual funds, under the ICA) or to clients (for investment advisors under the IAA).
- 3. The establishment of policies to manage conflicts of interest.

The regulatory push for the transparent and unconflicted use of shareholder voice started with pension funds. In 1988, in a letter responding to queries raised by the Chairman of the Retirement Board of Avon Products, Inc. (the so-called "Avon letter"), the US Department of Labor (DoL) opined for the first time that proxy votes were to be treated as ERISA plan assets and thus were captured under ERISA fiduciary obligations (DoL, 1988). The DoL followed this up by further correspondence two years later in the "Monks letter," sent to the then Chairman of ISS, and in 1994 in a legal interpretative bulletin (29 CFR 2509.94-2) that fleshed out ERISA obligations to vote, maintain voting records, and to have a proxy voting policy (DoL, 1990, 1994). While it appears the DoL did not enforce these obligations (McRitchie, 2014), the DoL's visible stance on the importance of proxy voting led to a number of representations (both public suggestions and formal petitions for rule making) to the SEC to make voting mandatory for investment companies governed by the ICA.

In 2003, in the wake of the *Sarbanes Oxley Act of 2002*, the SEC took action on this, making modifications to both the ICA (SEC, 2003a) and the IAA (SEC, 2003b) to enhance transparency in the proxy voting of mutual funds and their advisors. The basis for such amendments was that mutual funds and their advisers often had business ties with portfolio firms, which may have led them to vote their proxies in a manner that favored portfolio firms and their managers (who could otherwise withhold business) rather than their own fund shareholders. Under these amendments, mutual funds, ETFs, and their advisers are obliged to disclose their policies with regard to proxy voting.⁹ For mutual funds and ETFs, actual votes cast (either directly, or through an investment adviser) must also be publicly disclosed. Advisers do not have to publicly disclose their votes but must disclose them to their clients upon request. Further, both funds and advisers are required to develop and disclose policies that govern conflicts of interest.

⁹ For mutual funds and ETFs such policies may (and usually do) simply involve delegation of voting responsibility to their investment advisers. Advisers in turn are in principle able to have different policies for different client funds, though in practice, in the vast majority of cases mutual fund advisers apply a uniform policy across their client funds. In their large sample study involving all mutual fund proxy votes cast between 2003 and 2011, Cvijanovic *et al.* (2016) find that some 94% of their data involve family-level uniformity, i.e., the adviser who voted the shares followed the same policy across all funds advised.

While the initial regulatory push for transparent voting stemmed from the pension fund sector based on fiduciary obligations under ERISA, the 2003 ICA and IAA amendments have effectively rendered registered investment companies – mutual funds, ETFs, and their advisers – *more* obliged to the transparent exercise of shareholder voice than private pension funds. Indeed, the DoL clarified in a 2007 advisory letter that – since fiduciary duties under ERISA pertained *only* to taking actions that enhanced the economic value of plan investments – private pension funds must undertake a cost-benefit analysis ahead of voting their proxies (DoL, 2007). If the costs of becoming informed in order to vote and any cost arising as a consequence of voting is higher than the portfolio value added by voting, then it is the pension funds' trustees' fiduciary obligation *not* to vote.¹⁰

Further differences in legal obligations with respect to shareholder voice arise from the fact that *public* pension funds are not governed by ERISA, but instead only by state law (Black, 1992, Woidtke, 2002) and thus are not subject to ERISA fiduciary responsibilities.

Finally, hedge funds are usually organized as limited partnerships in which investors are the limited partners (LPs) and the general partners (GPs) manage the fund by acting as its investment advisers. Hedge funds avoid registration as investment companies (and thus regulation via the ICA) by utilizing two clauses that exempt pooled investment vehicles from registration under the ICA so long as they have either no more than 100 investors (clause 3(c)(1)) or only qualified purchasers (clause 3(c)(7)).¹¹ A further condition – common to both clauses – is that the fund only sells securities to such investors privately. However, since the passage of the *Dodd-Frank Act of 2010*, hedge fund advisers with more than \$150 million under management must register under the IAA, which then subjects them to all applicable provisions of the IAA, including imposing the IAA standard of fiduciary obligation on them (Champ, 2012).¹²

¹⁰ On a legal basis, registered investment companies may of course also still decide *not* to vote if it is in the best interest of their investors.

¹¹ Qualified purchasers are individuals with at least \$5 million in investments or companies with at least \$100 million in financial investments.

¹² In general, however, fiduciary obligation for hedge funds is complex due to their organization as limited partnerships and the flexibility permitted under relevant state limited partnership statutes. As Shadab (2009) puts it, *"the fiduciary duties of general partners are to a large extent waivable in the limited partnership agreement."* (Shadab, 2009, p. 248).

3.1.2 Obligation to Govern outside the US

Globally, the legal framework dictating the governance obligations of institutional investors vary significantly. In the UK, governance obligations are defined by a *Stewardship Code*, introduced in 2010 – one of the earliest examples of such stewardship codes – and revised in 2012 and 2020. Institutional investors and their "service providers" (e.g., proxy voting advisers) are required to abide by this code on a "comply or explain" basis – in effect this means that institutional investors must develop policies to either comply with the requirement of the code or explain on their webpages why they do not. Thus, in the UK – unlike in the US – there is less heterogeneity amongst mutual funds, hedge funds, and pension funds with regard to governance obligations. That said, since regulation is carried out under a "comply or explain" basis, there may be more *de facto* heterogeneity across institutional investors.

Interestingly, the UK code imposes some requirements not dissimilar to those of the ICA and IAA – institutional investors must vote proxies in their investors' best interest, must disclose voting policies, record proxy votes, and develop policies to deal with conflicts of interest. Importantly, the UK code requires institutional investors to "constructively engage" with portfolio firms, a condition without legal equivalent in the US.¹³

The UK Financial Reporting Council assess the reported stewardship activities to evaluate adherence to the code. However, it remains a challenge how to properly measure such adherence and especially the quality of engagement, as metrics such as the meeting frequency with portfolio firms contain little information on how deep the engagement is. Likewise, institutional investors may formally comply with the code by simply voting with proxy voting advisors' recommendations. Nevertheless, the UK *Stewardship Code* has been followed widely around the world, and by 2020 a total of 41 stewardship codes (and code revisions) were enacted in more than 20 countries (Katelouzou and Siems, 2020).

The UK code, and that of several other European countries, incorporate (and have influenced) the *EU Shareholder Rights Directive of 2017*, which modified the first *EU Shareholder Rights Directive of 2007*. The Directive was incorporated into national regulations during 2019 and 2020 and, as a result, reduced some aspects of cross-country heterogeneity in EU country governance obligations. However, important

¹³ Despite the lack of a legal requirement in the US, McCahery *et al.* (2016) provide evidence that institutional investors do engage behind the scenes with their investee companies.

differences remain (e.g., Germany requires no disclosure of actual proxy votes cast) and not all European countries impose purely code-based governance obligations. Globally, different types of institutional investors are often governed by different laws imposing different requirements. In India, for example, mutual funds and pension funds are governed by (distinct) laws, while insurers are governed by a code, leading to distinct obligations for different types of institutional investors. Table 3 summarizes key corporate governance obligations of institutional investors in China, Japan, Germany, the UK, and India.

[Insert Table 3 here]

In Germany, the obligations of institutional investors are determined by a combination of statutory requirements in the law and codes of conduct privately developed by the investment industry. Thus, different from the UK, Germany has no official code mandating stewardship activities. According to Ringe (2020), a reason for the country's reluctance to introduce a code are concerns that it would be unable to mandate engagement principles for foreign institutions, which constitute about 80% of German total institutional ownership (see Figure 8). The most important private code of conduct was introduced in 2003 by the German Investment Funds Association (BVI) and operates on a "comply or explain" basis. However, the code does not address important governance issues, such as the disclosures of voting records or of conflict of interest policies, and there is no monitoring of compliance with the code.

Similar to the UK, since 2014 Japan has a "comply or explain" stewardship code, revised in 2020, in order to enhance the way in which institutional investors engage with portfolio firms. The Japanese code explicitly states that stewardship activities should not be confined to voting and that engagement should be "constructive." Interestingly, one of the seven code principles explicitly asks institutional investors to develop and disclose a policy on how they manage conflicts of interest with portfolio firms – this seems particularly relevant in Japan, given that some Japanese institutional investors are criticized for too close business ties with their portfolio firms (Miyajima and Hoda, 2015).

18

3.2 Ability of Institutional Investors to Govern

3.2.1 Ability to Govern in the US

3.2.1.1 US Federal Proxy Access Rules

Institutional investors can actively govern portfolio firms by exercising shareholder voice. As shareholders, institutional investors are able to approach managers through a number of informal, "behind the scenes," channels to make suggestions and express approval or disapproval (for a discussion, see Shleifer and Vishny, 1986). However, to account for instances where managers may not agree with such shareholder suggestions, the effective use of voice requires that institutional investors be able to make *formal* proposals to corporate executives and have the ability to let other shareholders vote on such proposals. Such proposals are commonly referred to as proxy proposals. In the US, the ability of equity blockholders to govern investee firms via proxy proposal and the associated constraints are delineated by the *Securities Exchange Act* and the laws in the states in which portfolio firms are incorporated. These rules apply to all equity blockholders regardless of identity, so individuals, mutual funds, pension funds, and hedge funds, all face identical constraints. Thus, unlike with regard to governance *obligations*, there is commonality in the governance *ability* of US institutional investors in this regard.

The Securities Exchange Act provides two ways for blockholders to make proxy proposals. First, under Rule 14a-8, any blockholder owning 1% or \$2,000 in market value of a firm's shares for at least one year may include proposals on the firm's proxy statement. These are usually referred to as *shareholder proposals*. While this bar for proxy access is rather low, shareholder proposals must satisfy several conditions. The proposal may not relate to the firm's ordinary business operations, to director elections, and – crucially – they may not stand in conflict (or act as an alternative to) with any management proposal on the same proxy statement. Further, the blockholder is limited to a supporting statement of no more than 500 words. For normally scheduled annual general meetings, shareholder proposals must be filed 120 calendar days ahead of the issuance of the firm's proxy statement. Finally, the proposal also cannot conflict with state law. The net effect of all these restrictions, according to Black (1990) is that *"politically motivated shareholders can offer a wide variety of social responsibility proposals that most shareholders don't care about, while access is sharply limited for issues that affect the corporation as a profitmaking institution"* (Black, 1990, p. 541).

Given these limitations, as an alternative to shareholder proposals, institutional investors may sidestep the firm's proxy statement and solicit their own proxies, leading to what is commonly referred to as a "proxy fight." Fos (2017) shows that the majority of proxy fights are initiated by institutional investors. Avoiding the restrictions of Rule 14a-8 has the significant benefit of providing institutional investors with the ability to make a full case to shareholders and to address a much wider range of issues, including nominating an alternative slate of directors. Yet, it suffers from three important potential drawbacks.

First, a proxy fight is costly due to research, access, compliance, and potential legal costs (discussed below). Gantchev (2013) estimates that activist hedge fund campaigns ending in a proxy fight costs upward of \$10 million. Within limits, the reimbursement of a dissident blockholder's expenses is undertaken at the discretion of the target firm board. This means that unless the blockholder is able to replace the majority of the board in the course of the activism campaign, the target board is unlikely to reimburse expenses.

Second, soliciting proxies requires access to shareholder lists. Rule 14a-7 of the *Securities Exchange Act* obliges companies to disclose shareholder lists or mail dissident proxy material to shareholders directly at the dissident's expense. Black (1990) notes that *"Companies invariably mail the materials themselves to avoid disclosing shareholder names.... For shareholder lists, Rule 14a-7 can fairly be called a "nonaccess" rule."* (Black, 1990, p. 542).¹⁴

Third, the most contentious aspect of proxy access are legal and regulatory risk associated with disclosure and official vetting requirements. The key issue relates to the obligation for pre-filing and approval by the SEC of any materials related to a proxy solicitation, with a lead time of 10 business days before any communication with other shareholders. Further, the term "solicitation" is defined broadly, with Rule 14a-1(1) of the Securities Exchange Act considering as solicitation "communication to security holders under circumstances reasonably calculated to result in the procurement, withholding, or revocation

¹⁴ In a reform in 1992, discussed below, Rule 14a-7 was changed to impose further requirements on the target company. If the company chooses to mail the shareholder list, it has to disclose information on the number of shareholders management will solicit and provide an estimate of the costs incurred. Further, following the reform, shareholders launching a proxy fight may choose to have target company managers send communication to certain subsets of shareholders (Calio and Zahralddin, 1994).

of a proxy."¹⁵ Finally, solicitation communications are subject to stringent anti-fraud provisions under Securities Exchange Act Rule 14a-9.

Overall, these provisions hindered the ability of institutional investors to exercise shareholder voice (Black, 1990; Black, 1992; Roe, 1991). In particular, the pre-disclosure requirements made it difficult to challenge management proposals by soliciting proxies, since the delay induced by pre-filing and approval combined with the fact that management proposals were only announced on the proxy statement a month before the annual general meeting formed a nearly insuperable obstacle. Further, the rules rendered difficult any form of pre-voting communication amongst shareholders and also fostered legal risk in obtaining professional advice from proxy advisory services.

In response to these concerns, the SEC undertook a revision to the proxy rules in 1992. A key element of the changes involved the narrowing of the definition of a solicitation – starting in 1992, public communications and declarations by shareholders of how they intended to vote are no longer considered solicitations under the revised Rule 14a-1. While this removed an impediment to the coordination of shareholder opinion, a limitation of the change is that private communication was excluded (Hornstein, 1993). A second key element involved a provision under Rule 14a-2(b) that excluded from the various predisclosure requirements communication by (or on behalf of) any person not seeking the power to act as proxy (though they remained constrained by the anti-fraud provisions of Rule 14a-9). A consequence of the change is that it allowed for a degree of communication between independent shareholders (who did not seek to act as proxy for each other) during proxy fights. In terms of actual effects, Choi (2000) shows that the reform only increased the proxy process participation of shareholders such as unions and religious groups with specific agendas, and Bradley *et al.* (2010) shows that it facilitated communication and helped open-ending of closed-end funds.

An outcome of the 1992 reform to Rules 14a-1 and 14a-2(b) is that it facilitated hedge fund activists. These activists usually do not seek the power to act as proxy for other investors, preferring instead simply to persuade other investors to vote in a manner consistent with their activist agenda, thus

¹⁵ Black (1990) notes that the terms "communication" and "reasonably calculated" have been expansively interpreted by US courts so that the effective test is "whether the communication is "part of a continuous plan ending in solicitation and which prepares the way for its success"." (Black, 1990, p. 537). This effectively meant that any form of communication amongst shareholders with respect to how to vote would be considered a solicitation, and that even proxy advisory firms would be exposed to the relevant disclosure and vetting processes.

allowing them to benefit from the modifications to Rule 14a-2(b). Further, much of their persuasion is achieved via public means, for example, by active interventions during investor calls or via media campaigns. These interventions therefore benefit from the 1992 modifications to Rule 14a-1. Unsurprisingly, activism by hedge funds emerged as a major corporate governance force around 1994 (Brav *et al.*, 2008).

3.2.1.2 US Federal Rules for Proxy Advisory Firms

A further effect of the 1992 reform is that the modifications to Rule 14a-2(b) provided a broad exemption for proxy advisory firms in their operations. While the legal clarity may well have been welcome in 1992 – Black (1990) suggests that prior to 1992 advisory firms such as the ISS were simply ignoring disclosure requirements and thus taking legal risk - in more recent times the exploitation of this exemption has become more controversial. Following the 2003 modifications to the ICA and IAA that made voting, its disclosure, and the declaration and management of conflicts of interest mandatory for mutual funds and their advisers, the use of proxy advisory services has increased substantially. For example, Rock (2018) suggests that in underscoring the need to avoid conflicts of interest, the 2003 reforms to the ICA and IAA "...created a compliance challenge. By then indicating reliance on guidelines or a predetermined policy of voting based on "the recommendations of an independent third party," the SEC gave a boost to... the proxy advisory industry." (Rock 2018, p. 20-21). For these reasons, proxy voting advisors are of special relevance for institutional investors. Over time, concern has grown on at least three fronts. First, the proxy advisory industry is highly concentrated, with two players, ISS and Glass Lewis, dominating the market. Second, while advising shareholders how to vote, some proxy advisers also advise target management on how to influence shareholders. Finally, their recommendations are criticized for being opaque and (perhaps necessarily) subjective and sometimes "appear capricious, particularly on issues such as social responsibility" (Tett, 2019). These concerns have led the US Chamber of Commerce to lobby for additional regulation of proxy advisory services. We discuss evidence on proxy voting advisors in Section 5.5.

While new regulation is not yet forthcoming, in August 2019, the SEC issued new guidance (SEC, 2019a) with regard to the degree to which the provisions of the *Securities Exchange Act* apply to proxy advisory services. In particular, the SEC reiterated that the opinions of proxy advisory services – notwithstanding the 1992 14a-2(b) exemption from pre-filing and disclosure – continued to fall within the definition of solicitation under the *Securities Exchange Act*, and thus was subject to the stringent anti-fraud

provisions of Rule 14a-9. The SEC went on to make several specific recommendations with regard to *how* a proxy advisory firm may provide reassurance that it was not in violation of Rule 14a-9. This could include, for example, *"an explanation of the methodology used to formulate its voting advice on a particular matter (including any material deviations from the provider's publicly announced guidelines, policies, or standard methodologies for analyzing such matters)"* (SEC, 2019a, p. 12). Thus, the SEC's efforts appear to be aligned with generating greater transparency in the operations and recommendations of proxy advisory firms. More ominously for such firms, the SEC also noted that: *"the staff is also considering recommending that the Commission propose rule amendments to address proxy advisory firms' reliance on the proxy solicitation exemptions in Exchange Act Rule 14a-2(b)"* (SEC, 2019a, p. 3). Simultaneously with this opinion, the SEC issued a second guidance document (SEC, 2019b) that commented at length on the responsibilities of investment advisers with regard to fiduciary obligations in proxy voting. This includes a lengthy discussion of the considerations that investment advisers should undertake when employing the services of a proxy advisory firm to assist in fulfilling their fiduciary obligation to vote proxies in the best interest of their clients. Proxy reform remains an ongoing process.

3.2.1.3 US Federal Ownership Filing Requirements

To exercise influence via voice, it is key for institutional investors to have a significant number of shares. In this regard, a different section of the *Securities Exchange Act* imposes restrictions on their ability to govern (arguably more so than proxy rules). The relevant requirements arise from Section 13(d), under which any person that owns more than 5% of a firm's shares must file a Schedule 13D providing information about themselves, their ownership, and intentions with regard to the firm (within ten business days of crossing the ownership threshold). The same requirement is jointly imposed on groups of investors who collectively have an ownership of more than 5%. The SEC's definition of a group is *"two or more persons who agree to act together for the purpose of acquiring, holding, voting, or disposing of equity securities"* (Rule 13d-5).¹⁶ Anyone who owns more than 5% but does not wish to influence control of the firm may file instead on Schedule 13G.

¹⁶ At large investment managers such as BlackRock, the definition of a "beneficial owner" combines different legal entities within the investment manager together. Morley (2019) shows how this affects the possibilities for activism by large investors.

The disclosure under Section 13(d), and the applicability of such requirements to groups of investors, creates significant legal risk for institutional investors. Even without group disclosure requirements, the obligation for institutional investors to declare their intentions ex ante can lead to lawsuits by firm managers on the grounds of concealment of true intent (Macey and Netter, 1987), with attendant costs and risks to the 13D filer. The group filing requirement increases legal risks, because there may be scope to question whether some investors who acted similarly ex post (perhaps for independent reasons) had agreed informally to act together ex ante.¹⁷ In fact, Black (1990) alleges that 13D regulations are sometimes more restrictive than even the pre-1992 proxy solicitation rules – when shareholders could at most approach ten other shareholders to coordinate – when it comes to the exercise of voice. The definition of a group has legal implications beyond the 13D filing requirements – poison pills, for example, may be triggered when a group of shareholders crosses a certain ownership threshold. However, the legal environment is evolving with some notable changes. Notably, some recent court rulings in relation to 13D regulations provide greater latitude for shareholder coordination, stating that Section 13D does not require shareholders to disclose a group *until* they agreed to act together with respect to the voting of their shares – before that moment they can act together with respect to a target's corporate policy.¹⁸ Nevertheless, McCahery et al. (2016) document that concerns about legal risks related to such acting as a group are key impediments for institutional investors to jointly engage with target firms.

3.2.1.4 US State Law

Important further rules governing the ability to exercise voting rights originate from the laws in the US states in which portfolio firms of equity blockholders are incorporated. As more than half of US publicly traded firms are incorporated in the state of Delaware, the laws and courts of this state are most applicable. Key regulation governing the ability of institutional investors to engage in governance activities of Delaware firms are specified in the *Delaware General Corporation Law*. Thus, it is Delaware corporate that determines for instance what institutions can vote on, including the election of directors, mergers, and amendments to the Certificate of Incorporation – the same holds for other governance activities, such the ability to sue a firm or its directors. As the majority of institutional investors neither submit shareholder

¹⁷ Black (1990) provides examples of instances in which institutional investors have been subjected to lawsuits either alleging the existence of an undeclared 13D group or the delay in the revelation of a declared 13D group, and notes that courts have allowed circumstantial evidence to demonstrate the existence of such groups.

¹⁸ See CSX Corp. v. The Children's Investment Fund Management, 08-2899-cv (2d Cir. July 18, 2011).

proposals nor enter into proxy fights, the possibility to vote against management at the general meeting – notably, on the election of the board of directors in a proxy contest – is in turn a key channel for many institutions to govern their portfolio firms. The implications of this practice is that the state laws legislating specific voting issues are arguably more important for the ability to govern than the federal proxy voting regulations of Rule 14a-8.

3.2.2 Ability to Govern outside the US

3.2.2.1 Proxy Access outside the US

There is significant variation in the ability to govern globally. In the UK, proxy access is governed by the *Companies Act of 2006* and influenced by the *UK Corporate Governance Code*. Shareholders in the that country have significantly more power conditional on a proxy vote than in the US, because shareholder votes are binding on corporate executives, whereas in the US shareholder proposals are only advisory. Buchanan *et al.* (2012) also note other important sources of enhanced shareholder power in the UK relative to the US. In the UK, a group of shareholders who in aggregate hold 5% or more in a firm can call an extraordinary general meeting (EGM) according to Sections 303 to 306 of the *Companies Act of 2006*. The shareholders can then use the EGM to demand the resignation of the entire board of directors, and they only need a strict majority of shareholders voting to do so. Activist investors in turn may use the *threat* of an EGM when engaging with companies in order to achieve their objectives. An example where an institutional investor successfully called for an EGM and replaced the board is the case of Brazit. In this firm, two institutional investors (one of them the Hermes UK Focus Fund) engaged with the board to demand various changes. When private negotiations failed, the two investors called for an EGM at which the entire board was dismissed (see Becht *et al.*, 2009).

In general, UK director elections are determined by simple majority at the general meeting – in contrast to the US where plurality voting occurs – and institutional investors can vote down the re-election of directors at the general meeting. Further, the *UK Corporate Governance Code* requires directors of large (FTSE 350) listed firms to be reelected annually. Thus, UK regulation is arguably more shareholder friendly than in the US on issues such as board elections and the possibility of replacing the board.

To sponsor a proposal, shareholders in the UK need to own in aggregate 5% of the voting shares according to Sections 338 to 340 of the *Companies Act of 2006* (five times the US threshold). Alternatively,

a group of no less than 100 shareholder (each owning no less than £100 in market value) can sponsor a proposal, but this requirement imposes a significant coordination requirement – such shareholder resolutions can then be used to propose the election of a new director. Further, according to the *UK Corporate Governance Code*, if 20% or more shareholders vote against the board recommendation for a resolution at the general meeting, the firm should explain how it intends to consult shareholders in order to better understand the reasons behind the vote.

In Germany, institutional investors' abilities to govern firms are primarily regulated by the *Stock Corporation Act*. According to §122(1) of the act, shareholders owning together at least 5% can request an extraordinary shareholder meeting. Similarly, according to §122(2) shareholders who in aggregate own 5% or more (or at least EUR 500,000 of the share capital) can put an item on the agenda of the general meeting. An important governance tool for institutional investors is the threat to request a special audit according to §142 of the *Stock Corporation Act*, which then assesses whether compensation claims against the board can be raised. If a firm refuses the request of a special audit, investors owning at least 1% (or EUR 100,000) can go to court to request the appointment of a special auditor. A case in point is Volkswagen AG, where a court in 2020 appointed a special auditor in order to investigate the firm's emission scandal.

Another special governance feature in Germany is related to the country's two-tier board system. As a result of this structure, shareholders can only elect members of the supervisory board (with a simple majority), and this body in turn appoints the management board. However, shareholders can use the general meeting to pass a vote of no confidence against the management board. Though extremely rare in German corporate history and operating mostly as a threat, an example for the successful use of this governance channel is the case of Bayer AG – triggered by large legal costs associated with the acquisition of US firm Monsanto, a majority of investors at the general meeting in 2019 supported a no-confidence vote against the firm's management board.

In Japan, several points related to institutional investors' ability to govern are noteworthy. First, any shareholder owning 3% or more in a firm can call an EGM (Article 297 of the *Companies Act*), a hurdle that is lower than in the UK. Second, shareholders owning 1% or more can propose a resolution for the shareholder meeting according to Articles 303 and 305 of the *Companies Act*. Third, compared to many other countries, shareholders in Japanese firms have relatively strong rights with respect to shareholder proposals, as they can pass *binding* resolutions requiring the management to make or change major

business decisions. Despite these relatively strong formal rights for institutional investors, most engagements by institutions in Japanese firms occurs through more informal and private routes. Engagement by foreign activists seem particularly rare to date – despite the relatively large foreign institutional ownership in Japanese firms (Figure 8) – arguably due to cultural reasons and the difficulty of obtaining support from domestic institutional investors.

[Insert Table 4 here]

Table 4 summarizes key proxy access rules for the UK, Germany, and Japan, and the table also contains information for key regulations in China and India. Overall, the displayed information demonstrates that the ownership threshold for proxy access varies between 1% and 10%, with India requiring the largest equity stake. The lead time ahead of the general meeting for shareholders to make proposals known to management varies from 10 days to 45 days (India again requires the longest lead time). In terms of regulations, China is located somewhere in the middle across the five countries with respect to the minimum equity stakes required for proxy access (3% in China) and the lead times for shareholder proposals (10 days). The table also shows that there are differences across major economies with regard to deadlines for firms to respond to or publish shareholder proposals.

3.2.2.2 Ownership Filing Requirements outside the US

Globally, there exist a number of disclosure rules not dissimilar to the 13D regulation in the US. For China, Japan, Germany, the UK, and India, these rules are detailed in Table 5. Two points are noteworthy. First, an inspection of the table makes it clear that there are significant similarities in the disclosure rules between the UK and Germany. This is because they both trace their origins to a single 2004 EU directive, the so-called *EU Transparency Directive* (EU Directive 2004/109/EC). As is often the case with EU directives, member states have a degree of leeway in implementing these in national law, so that regulation need not be identical across countries.

Second, disclosure requirements stemming from the *EU Transparency Directive* are less onerous than those stemming from the *Securities Exchange Act* in the US. While the *EU Transparency Directive* simply requires disclosure of holdings and some information about the chain of ownership at 5% and at a number of thresholds thereafter, the 5% disclosure threshold in the US requires an explicit choice from blockholders – they have to decide whether to be passive (in which case they file a Schedule 13G) or active

(in which case they file a Schedule 13D); in the latter case, they are also required to provide further disclosure of their intentions.

[Insert Table 5 here]

A key related aspect is the question of what constitutes "acting in concert" by institutional investors outside the US, how this differs from the definition of a "group" for US Schedule 13D filings, and what the potential governance consequences of such rules are. The *UK Takeover Code* defines as acting concert the situation where "persons who, pursuant to an agreement or understanding (whether formal or informal), co-operate to obtain or consolidate control." In Germany, acting in concert is similarly defined by the *German Securities Acquisition and Takeover Act*. For the ability of institutional investors to govern it is key to note that mistakes in the filing requirements resulting from shareholders acting in concert (or mistakes in ownership filings in general) can lead to the full loss of the voting rights. Importantly, in the UK and Germany, shareholders acting in concert and jointly owning more than 30% of a target firm are required to make a mandatory takeover bid to the remaining shareholders.¹⁹ Hence, rules on acting in concert also act as significant impediments to coordinated shareholder activism outside the US. Yet, similar to the US, some recent court rulings in Germany increased the leeway that investors have when coordinating engagement with target firms.²⁰

3.3 Incentives of Institutional Investors to Govern

The impact of the law on incentives is more subtle than its impact on obligations and ability to govern: while in the latter two the impact is primarily direct, the impact on the former is likely to be indirect, working through trade-offs imposed on institutional investors and potentially interacting with the reactions of their clients to any behavior that arises in response to such trade-offs. A full understanding of the impact of regulation on governance behavior thus requires careful modeling and empirical analysis and extends beyond the domain of the legal literature. The academic literature in finance has engaged – theoretically and empirically – with several aspects of the impact of the (law-induced) incentives of institutional investors on their governance behavior. As a result, we take a two-stage approach: in this section, we focus on how the legal environment *fosters* particular trade-offs for asset managers and how

¹⁹ In contrast, there is no mandatory bid requirement under US federal or Delaware corporate law.

²⁰ See the Ruling II ZR 190/17 by the German Bundesgerichtshof in 2018.

such trade-offs may vary *across* different institutional investors, but leave a more detailed discussion of *how* such incentives affect governance behavior to Sections 4 and 5.

A key manner in which the legal framework ultimately affects incentives to govern is via constraints on fees that institutional investors can charge. In the US, the IAA forbids registered investment advisers from the charging of performance fees to clients based purely on fund profits unless these fees are symmetric in losses and gains relative to a benchmark (the so-called "fulcrum fee rule"), leading to the vast majority of mutual funds to charge flat assets under management fees (Elton *et al.*, 2003). However, as Shadab (2009) notes, there is a key exemption. The adviser may charge performance fees when advising a fund which is exempted from registration under the ICA under Clause 3(c)(7) or if all fund investors meet certain qualifying (net worth or investment) characteristics. Since hedge funds are almost universally organized to be exempt under Clause 3(c)(7), this means that even following the *Dodd-Frank Act*, hedge funds advisers are free to charge performance fees to their clients. Accordingly, the vast majority of registered investment companies charge flat fees designated as a percentage of assets under management (AUM fees), while hedge funds charge a combination of such fees with (much higher percentage) performance fees designated on performance over some benchmark or watermark.

The preponderance of AUM fees fosters rather directly the temptation to increase assets under management and thus induces registered investment companies to "compete for flow." The asset pricing literature persuasively documents that mutual funds respond in their trading and risk-taking behavior to such flow incentives (Brown *et al.*, 1996; Chevalier and Ellison, 1997). The corporate governance literature, discussed in Section 4, show that such flow incentives may weaken the incentives of mutual funds to govern via exit and voice (Dasgupta and Piacentino, 2015; Song, 2017). Recent papers nevertheless also demonstrate how flow incentives can generate subtle positive governance effects by inducing mutual funds to support activist hedge funds in their engagement efforts (Brav *et al.*, 2021; Cvijanovic *et al.*, 2019).

The IAA's prohibition of asymmetric performance fees also makes it difficult for registered investment companies to tie fees to returns from specific governance activities that may – in principle – generate high returns. For example, an activist engagement in a firm may significantly enhance its value, and thus the portfolio value of a fund that holds a significant fraction of the firm's shares. Yet, if only the AUM fee fraction of the value enhancement (typically between 1% and 2% for mutual funds) can be earned

by the fund, then there is little incentive to pay the significant costs associated with the activist engagement (Bebchuk *et al.*, 2017).

Legal scholars suggest that a second manner in which the law affects governance incentives of investment companies is via the requirements for the transparent and unconflicted use of voice. Rock (2018) argues that these disclosure requirements have fostered "compliance type" incentives in governance. Referring to the 2003 SEC amendments to the ICA and IAA, Rock writes (p. 20-21):

"The SEC, in emphasizing money managers' fiduciary duties, and the extent to which conflicts of interest may breach those duties, created a compliance challenge. By then indicating reliance on guidelines or a predetermined policy of voting based on "the recommendations of an independent third party," the SEC gave a boost to "guideline based voting"... as well as to the proxy advisory industry."

It is plausible that investment companies and advisers treat governance as a compliance issue. In indirect support of such a link, Bebchuk *et al.* (2017) and Rock (2018) note that the large asset management families have very small governance departments. According to data in Bebchuk and Hirst (2019), BlackRock, for example, had a team of only 45 people responsible for 11,000 equity positions worldwide. Rock (2018) also notes that the senior staff within these governance departments are not compensated in a manner that relies on the value of assets under management. However, the finance literature suggests that we should be cautious in attributing compliance incentives to asset managers when it comes to governance. Iliev and Lowry (2015) demonstrate substantial heterogeneity amongst mutual fund families in the degree to which they rely on third-party recommendations for their voting, with only 25% of such fund families following such recommendations uniformly. (Further related papers are discussed in Section 5.) Similarly, an examination of the SEC 2003 amendments also suggests that the SEC considered, but did not implement, a proposed further change (SEC, 2003a, p. 12):

"The Commission has determined not to adopt the proposed requirement that a fund disclose in its annual and semi-annual reports to shareholders proxy votes (or failures to vote) that are inconsistent with the fund's proxy voting policies and procedures... Opponents of proxy voting record disclosure argued that the disclosure of inconsistent votes would be burdensome because it would require funds to analyze a large volume of proxy votes to determine whether any vote triggered the disclosure and then to provide a lengthy explanation to shareholders regarding each inconsistent vote..." As a result, in practice, funds may have significant latitude with regard how they vote on a particular proposal at a particular firm: since large fund families (as is clear from the extract above) vote in a huge number of proxy proposals in any given reporting period, and it would not be difficult in principle to (quietly) vote in a manner that selectively violates its declared policy in a relevant case. This could, of course, have the benefit of rendering the exercise of voice less of a "compliance" issue as feared by Rock (2018), but it could also enable fund families to favor the management view in instances where they have other connections (e.g., due to existing business ties) with portfolio firms.²¹ It is worth pointing out that the legal framework fosters a further difference between registered investment companies and hedge funds with respect to the exercise of voice: as Bebchuk *et al.* (2017) emphasize, since hedge funds cannot accept 401(k) investments, they are less tempted to pro-management voting.²²

A final manner in which the legal framework affects governance incentives is through the nature of institutional investors' internal governance. As discussed above, state pension funds are often governed by elected boards with political objectives and may well have differing incentives from those of private funds (Black, 1992; Woidtke, 2002).²³ Labor union pension funds, which are regulated under ERISA (like private pension funds), represent a particularly interesting case. As discussed in Section 3.1, the exercise of shareholder voice falls within the purview of ERISA fiduciary obligations: various communications from the DoL in the 1980s and 1990s established that proxy votes were to be treated as plan assets. However, the 2007 clarification issued by the DoL stated that ERISA treats the economic value of plan investments as primary – if there is a conflict between the obligation to vote and the obligation to maximize the economic value of plan assets, then voting is not recommended. Thus, a plan that votes in a manner that reduces the economic value of plan assets would surely be in violation of ERISA fiduciary obligations, and thus at clear risk of lawsuits by plan beneficiaries. Yet, it turns out that ERISA fiduciary obligations have limited bite for labor union pension funds, because the plan beneficiaries are workers whose interests may sometimes diverge from the economic value of the companies that employ them. For example, if a labor union pension funds takes proxy voting actions that seek to support workers' rights at a particular

²¹ As we discuss in Section 4, the finance literature documents evidence of such pro-management voting at the level of closely-contested individual proposals in individual firms with which large fund families have substantial 401(k) pensions management business ties (Cvijanovic *et al.*, 2016).

²² Morley (2019) argues that activism by big investment advisers is impeded due to internal conflicts among their clients with regard to activism.

²³ Further complications for public pension and labor union pension funds arise because of the specific governance rules on how current employees and retirees elect the funds' boards (Webber, 2014).

corporation inducing a potential loss in the value of that corporation, they would be *de jure* in violation of ERISA fiduciary obligations, but their *de facto* risk of a lawsuit would be minimal because the actions are benefiting exactly the people who have the right to sue them. Indeed, the literature has documented evidence of such behavior: Agrawal (2012) shows that some labor union pension funds pursue worker's rights at firms in which they are significantly represented by aggressively voting against management recommendations in director elections.

4. Theoretical Literature on Institutional Investors and Corporate Governance

4.1 Classical Theoretical Literature on Blockholder Governance

The theoretical literature on the corporate governance role of institutional investors stems from the blockholder monitoring and engagement literature. The underlying need for blockholder monitoring arises from the separation of ownership and control in publicly traded firms coupled with the failure of compensation contracts and/or board monitoring to overcome the negative effects of such separation.²⁴ The classical blockholder monitoring literature has proposed three distinct (but interrelated) channels by which blockholders may govern firms.

At the *passive* end of the monitoring spectrum, blockholders may monitor firms via (the threat of) trade, also often referred to as "governance via exit" (Admati and Pfleiderer, 2009; Edmans, 2009). In these models corporate executives take actions of varying desirability from the perspective of firm value, and blockholders trade on the basis of the observation of such choices, impounding the impact of managerial actions into prices and delivering market discipline.

By way of comparatively *active* monitoring, blockholders may take actions – behind-the-scenes engagement, public interventions, shareholder proposals, voting on proposals, or proxy fights – in order to directly enhance firm value. Such activities are collectively referred to as blockholder "voice." The extensive theoretical literature (beginning with Shleifer and Vishny, 1986) on "governance via voice" has

²⁴ For insightful recent surveys of the benefits and limitations of boards and executive compensation see Adams *et al.* (2010) and Edmans *et al.* (2017), respectively.

focussed on issues such as the role of block size, the number and size distribution of blocks (Winton, 1993; Edmans and Manso, 2011), and the costs of external monitoring (Burkart *et al.*, 1997).

Finally, at the most active end of the intervention spectrum, blockholders may "govern via takeovers" (Grossman and Hart, 1980). A few papers allow for blockholder trade while focussing on active monitoring via either takeovers (Kyle and Vila, 1991) or voice (Kahn and Winton, 1998; Maug, 1998; Faure-Grimaud and Gromb, 2004), though in none of these papers is trade inherently a governance mechanism.²⁵

The canonical models of blockholder monitoring are non-specific with respect to the *identity* of the blockholder: they apply equally well to (wealthy) private individuals and institutional investors. This is a consequence of the fact that such models feature governance actions that are readily accessible to any blockholder regardless of their identity. For example, for a given (qualifying) level of ownership, an individual blockholder is as well able as an institutional blockholder to place shareholder proposals on a firm's proxy statement, to vote on such proposals, to gain information about management's actions, and to trade on such information. Indeed, the early history of shareholder engagement, which formed the backdrop for the original theoretical literature on blockholder activism in the US, Gillan and Starks (2007) highlight the early role both of prominent individuals (e.g., Evelyn Davis, Lewis and John Gilbert, and T. Boone Pickens in the 1970s and 1980s) and of institutions (e.g., the public pension funds CalPERS and CalSTRS in the 1980s).

The non-specificity of the classical theoretical literature with respect to the identity of blockholders stands in contrast to our specific interest in the role of institutional investors. As a result, we do not dwell in detail on this broader set of papers. For an excellent and comprehensive survey of this broader literature, we refer readers to Edmans and Holderness (2017). For our purposes, we focus on contributions that study institutional investors specifically. In other words, we trace the *response* of the theoretical blockholder monitoring literature to the significant growth of the asset management industry documented above.

²⁵ Recent papers have generalized models that incorporate trading and voice (Back *et al.*, 2018), modeled the choice between voice and takeovers (Burkart and Lee, 2021; Corum and Levit, 2019) and also considered the complementarity of governance via exit and voice (Dasgupta and Piacentino, 2015; Levit, 2019).

There are two conceptual ways in which this newer literature has focussed on institutions. First, this literature has identified institutions by focussing on "what they can do," that is, by conditioning on abilities specific to institutional investors. Second, this literature has identified institutions by focussing on "who they are," that is, by conditioning on the identity of institutional investors as delegated blockholders acting as agents of end-investors who provide them capital, which in turn defines their incentives to govern. We discuss these two strands of the literature in Sections 4.2 and 4.3. Finally, the theoretical literature has recently focussed in on the effect of proxy voting advisors, who have gained prominence in the voting process given the obligations of institutional investors to vote. We discuss this in Section 4.4. Overall, though they are presented in an order reflecting the chronological development of the literature, by focussing on the ability to govern, the incentives to govern, and the obligations to govern, Sections 4.2, 4.3, and 4.4 form conceptual parallels to Sections 3.2, 3.3, and 3.1, respectively. Finally, there are significant – albeit imperfect – links between the theoretical (Section 4) and empirical (Section 5) literatures in finance on the governance role of institutional investors. For expositional ease, we examine links between the theory and empirics – and discuss areas in which these can be enhanced – at appropriate points in Section 5. We also return to some key themes with regard to linking theory and empirics in our concluding Section 6.

4.2 Theories Centered on Institution-Specific Ability

Chronologically, the first set of theoretical papers that modelled factors specific to institutional investors in corporate governance focused on their ability as sophisticated traders. This strand was motivated by the growth of hedge funds, who are nimble and sophisticated traders, able to utilize trading strategies usually unavailable to individual investors. Early in the century, regulators, and practitioners expressed concerns that hedge funds were exploiting their trading ability, including access to sophisticated derivatives markets, to acquire voting power in excess of their economic interest thus nullifying the "one-share, one-vote" structure usually embodied in public firm charters (Grossman and Hart, 1988). Legal scholars (e.g., Hu and Black, 2007) argued that this phenomenon, often referred to as "empty voting," could lead to perverse outcomes.²⁶ For example, if hedge funds could first acquire an excess of votes by borrowing shares on the record date of an important vote, but then skilfully trade to a short position in

²⁶ The term "empty voting" is not meant to be taken literally, as in the situations under consideration hedge funds do not literally have zero economic interest in the firm.
the period by the time the voting date arrived, then they could vote against a proposal that would otherwise enhance firm value in order to profit from their short position.

Brav and Mathews (2011) consider the economics of empty voting. They recognize the underlying trade-off that while hedge funds may have the incentive and ability to manipulate outcomes, they may also be better informed about the value inherent in individual proposals. Due to this information advantage, when other shareholders are less likely to be informed or vote in the correct manner, empty voting by hedge funds may be value enhancing. However, Brav and Mathews (2011) also argue that if it is relatively inexpensive to separate voting power from economic ownership at the record date (as a result of being able to borrow shares easily and cheaply), or if the existing shareholder base is sufficiently likely to vote in the correct manner, then the negative aspect of hedge funds' manipulation incentive may dominate, making empty voting a net negative.

Other researchers study a phenomenon closely related to empty voting, namely a situation in which a hedge fund can acquire a short position in a firm's equity while holding a long position in its debt close to a restructuring. In this case a hedge fund could again obtain trading profits by voting – as debt holder – against a restructuring proposal that preserves or enhances firm value. This issue is particularly relevant since hedge funds are known to be involved in the vast majority of large debt-restructurings in the US (Jiang *et al.*, 2012). Of course, restructuring proposals are endogenous, and corporate executives may adjust their proposals to nullify such strategic manipulation. Zachariadis and Olaru (2017) consider when such manipulation may be possible despite the endogeneity of restructuring proposals. They show that such manipulation is feasible when non-hedge fund traders are net buyers in the equity market (providing cover to hedge fund short-selling) and debt and equity markets are informationally segregated (so that hedge funds purchases in debt markets do not affect their ability to short sell in equity markets) and that it is precisely under such circumstances that corporate executives are most tempted to risk extracting higher value for shareholders by making aggressive restructuring proposals.

Brav and Mathews (2011) and Zachariadis and Olaru (2017) apply to institutional investors specifically as the mechanisms rely on the ability of hedge funds to trade across markets, borrowing and short-selling shares, trading simultaneously in debt and equity markets, and hedging economic exposure in derivatives markets. In a similar vein, Edmans *et al.* (2019) focus on another distinctive aspect of institutional investors – that, by virtue of their size, they are *common* owners across large numbers of firms

as a result of their size.²⁷ Their insight is to recognize that such common ownership can potentially enhance governance via both exit and voice. This is because, when investors choose *which* of many potential firms to sell, their sale is more informative than if they held only a single firm. This increases the price impact of their sale which improves information aggregation (enhancing the exit governance mechanism) and simultaneously increases the cost to "walk away" from the firm (enhancing incentives to engage via voice). Interestingly, the positive characterization of the governance impact of common ownership in Edmans *et al.* (2019) stands in contrast to some of the empirical literature which focuses on the anti-competitive effects of common ownership (see Schmalz 2018 for a recent survey). Building on the ability of institutional investors to hold many shares – but simultaneously recognizing that there are limits to the attention that such investors can devote to each firm in a large portfolio – Gilje *et al.* (2020) develop a model to quantify the impact of ownership overlap. They show that the impact of common ownership on managerial actions depends simultaneously on the degree to which the manager cares about institutional investor preferences, the weight the institution places on the relevant cross-firm externality, and the likelihood that the institutional investor is well informed about the portfolio value impact of the manager's choices.

4.3 Theories of Blockholders as Agents

4.3.1 Dual-Layered Agency Structures

Institutional investors are – by definition – asset managers, that is, they manage other people's money. This endows their role in corporate governance with a unique set of characteristics. When institutional investors hold equity blocks they participate in a dual-layered agency relationship. On the one hand, as equity holders in the firms in which they hold blocks, institutional investors act as principals who must monitor the actions of their agents (corporate executives). On the other, institutional investors are themselves agents, acting on behalf of their own clients (fund investors) and are monitored by them. Thus, institutional blockholders are simultaneously principals and agents. The second strand of the theoretical literature takes this dual-layered agency relationship as a starting point to examine the role of institutional investors in corporate governance.

The essence of how dual-layered agency structures may affect the governance role of institutional investors stems from incentives arising from the agency relationship between asset managers and their

²⁷ Another contributory factor, particularly for mutual funds, are limits on holding concentrated positions in individual firms, which leads to a spreading of holdings over many firms.

investors. Any aspect of incentives arising from such a relationship that leads institutional investors to behave differently from the stylized profit maximizing agents of the traditional blockholder monitoring literature (see Section 4.1) can affect the nature of corporate governance.

One key aspect of differences in incentives is canonical: almost by definition, an asset manager holding a block of shares will have a lowered level of direct interest, i.e., lowered "skin in the game" than a proprietary blockholder (who owns 100% of the block). The degree to which such skin in the game is reduced may vary across types of institutions. At one extreme, retail-facing, large, long-only asset managers such as mutual funds tend to have very little skin in the game. According to Khorana *et al.* (2007) over half of mutual funds have zero managerial ownership. In contrast, smaller, less regulated vehicles such as hedge funds that cater to more sophisticated clients can have significantly higher skin in the game. The literature suggests that hedge funds' skin in the game can vary from around 7% (Agarwal *et al.*, 2009) to 20% (He and Krishnamurthy, 2013); recent data by Gupta and Sachdeva (2019) also documents large cross-sectional variation in hedge funds' skin in the game.

A second key aspect of the agency relationship between asset managers and their investors is shaped by regulation. As discussed in Section 3.3, in the US the IAA forbids registered investment advisers from the charging of performance fees to clients based purely on fund profits unless these fees are symmetric in gains and losses (the so-called "fulcrum fee rule"). This leads the vast majority of mutual funds to charge flat assets under management fees (Elton *et al.*, 2003). As discussed above this in turn fosters a temptation to increase assets under management and thus induces registered investment companies to "compete for flow." Such competition for flow is central to the incentives of mutual fund managers (Brown *et al.*, 1996; Chevalier and Ellison 1997). However, even hedge funds – which are exempt from registering as investment companies under Clause 3(c)(7) of the ICA and thus have complex contingent compensation contracts (see Section 3.3) – gain a majority of their overall compensation from future flows rather than from their explicit compensation (Lim *et al.*, 2016).²⁸

²⁸ As Lim *et al.* (2016) find, flow considerations are more relevant for younger, growing hedge funds as older, successful hedge funds are often closed to new capital.

4.3.2 Dual-Layered Agency Models and the Exit Mechanism

The theoretical literature focussing on the dual-layered agency problem has considered both the effects of skin in the game and the role of competition for flow in affecting the governance role of institutional investors. Dasgupta and Piacentino (2015) study the role of competition for flow and skin in the game in a dual-layered agency model of the exit mechanism. In their model, funds (institutional investors) are differentially skilled stock pickers who must choose amongst firms on the basis of firms' future cash flows. These cash flows in turn are determined by the quality of firms' governance, about which funds have private information. In poorly governed firms, executives may take value destroying actions. Funds observe such choices and then decide whether to sell or not. The noisy market reacts imperfectly to such trades.

Exit governance works best if blockholders sell whenever executives make value destroying choices, thus lowering prices and punishing the executives. A credible threat of exit makes corporate executives take value destroying choices less often. In the canonical exit model of Admati and Pfleiderer (2009), which features proprietary blockholders, the threat of exit is always credible, because a blockholder who observes executives making value destroying choices would rather sell now in a noisy market than hold and feel the full negative value impact. Dasgupta and Piacentino (2015) show that competition for flow reduces the credibility of the exit threat, weakening exit as a governance mechanism. Funds who execute on a threat of exit convey information about their own *ex ante* choices to their investors. By exiting, the fund highlights that it chose a poorly governed firm to hold a block in, which reduces investors' confidence in the fund's stock picking ability. This leads to outflows which is costly for the fund, making the fund reluctant to exit and reducing the credibility of the exit threat.

Such competition for flow is complemented by the skin in the game effect. Exit governance works better the more keenly blockholders feel the value reducing impact of executives' choices. The lower the skin the game, the less the fund will suffer directly from value reduction. Dasgupta and Piacentino (2015) also show that when blockholder voice is non-binding on corporate executives (as is the case for shareholder proposals in the US), then blockholders who compete for flow will also be less likely to intervene. This is because the blockholder's threat to exit if his intervention is ignored is less credible, making it likely that executives will ignore that blockholder's voice, disincentivizing him from expending effort on interventions.

4.3.3 Dual-Layered Agency Models and Short-Termism

At the broadest level, competition for flow is a particularly salient manifestation of short-termism. For example, in the discussion above, when corporate executives make poor choices, institutional blockholders who compete for flow are hesitant to take the action (exit) that maximizes long-term payoffs, instead endogenously giving priority to the short-term gains and losses arising from retaining or losing assets under management. In an early contribution, Goldman and Strobl (2013) examine the implications of exogenously specified blockholder short-termism on the complexity of firm investments. In their paper, it is assumed that the horizon of a blockholder – interpreted as an institutional investor – is shorter than the amount of time required for the market to correctly value corporate investments. This tempts the blockholder to manipulate share prices, which executives try to counter by increasing the complexity of their investments. The overall effect is to reduce the informativeness of share prices, thus weakening monitoring via trade.

While Goldman and Strobl (2013) and Dasgupta and Piacentino (2015) both consider *passive* monitoring via trade, Burkart and Dasgupta (2021) consider whether competition for flow can induce short-termism in *active* monitoring as well. In particular, they show that activist funds who compete for investor flows may be sub-optimally myopic in their engagement strategy, taking actions that shift returns to their investors forward in time at the expense of long-term cash flows. In particular, activist funds excessively leverage their target firms, risking subsequent debt overhang if economic conditions deteriorate, significantly reducing the overall cash-flow generating potential of activists.

In a related but distinct context, Piacentino (2019) examines the governance impact of competition for investor capital by institutional investors who provide financing in *primary* markets. Such investors – venture capitalists – are differentially informed in Piacentino (2019). Uninformed venture capitalists are reluctant to back start-up firms because they fear being revealed as being uninformed down the road, curtailing their ability to obtain additional or follow-on investor capital. This improves the average quality of venture capital backed firms in primary markets, providing a "certification effect" and improving the allocation of financing to projects.

4.3.4 Dual-Layered Agency Models and Multiple (Interacting) Blockholders

A number of papers consider the role of competition for flow in *multiple* blockholder contexts. These papers demonstrate the nuanced impact of institutional investor incentives in a multiple blockholder setting. Song (2017) considers the interaction of a flow-motivated blockholder (a fund) with a second blockholder who is not flow-motivated (a proprietary blockholder). This interaction between dissimilar blockholders gives rise to two effects. First, as a result of the disincentive of the flow-motivated blockholder to intervene (as shown in Dasgupta and Piacentino, 2015) the proprietary blockholder recognizes that his effort is essential and may intervene more, that is, free-riding is reduced. In turn, Song (2017) argues that the fear that the flow-insensitive blockholder may exit and make an inactive flow-motivated blockholder look bad in the eyes of his clients may actually induce him to intervene directly.

In contrast to Song (2017)'s focus on interactions between dissimilar blockholders, Brav et al. (2021) study the possibility of parallel engagements by *similar* blockholders in the form of so-called "wolf packs" of several activist hedge funds who hold relatively small blocks – a phenomenon that has attracted significant recent attention from practitioners and lawmakers. Such multi-blockholder parallel engagements are salient precisely because of the preponderance of small institutional blockholders (as we document in Section 2): while each blockholder is unlikely to have much individual governance impact, collectively their joint engagement has significant potential. Yet, phenomena such as wolf packs are puzzling from a theoretical angle: When engagement is costly, small blockholders are likely to underengage relative to the social optimum. This is because of the public goods problem inherent in blockholder engagement: successful engagement efforts benefit even those blockholders who do not engage. While a social planner would account for gains to the full set of shareholders, individual blockholders only take into account private returns. A potential solution to such public goods problems takes the form of transfers across agents: rewarding those that provide the public good at the expense of those who do not. In this context, Brav et al. (2021) consider the impact of a dual-layered agency relationship in fostering parallel engagement. They argue that while reduced skin in the game arising from delegation worsens the public goods problem, competition for flow can ameliorate it. The reason is that skilled funds enjoy an advantage over unskilled funds with respect to costs or information in activist engagements, which means that participation in a successful campaign signals ability to fund investors. Such fund investors then gain reallocatable investor capital from those funds who do not participate, because fund investors wish to invest with skilled funds. Thus, effectively, investor flows form an endogenous set of transfers that reward those blockholders who engage at the expense of those who do not, ameliorating the public goods problem and enhancing engagement incentives across small blockholders.

Finally, like Song (2017), Cvijanovic *et al.* (2019) also focus on the interaction between different types of blockholders. They show that flow-sensitive blockholders may rush to sell out of a firm following the exit of an informed blockholder – potentially even ignoring their own information – fearing that they will otherwise be left invested in a firm that is likely to underperform, leading to outflow. This in turn enhances the price impact from the informed blockholder's exit, increasing his influence over management, enhancing governance via exit.

A key theme that emerges from the literature on dual-layered agency models is institutional investor heterogeneity. In Dasgupta and Piacentino (2015), Song (2017), and Cvijanovic *et al.* (2019) blockholders with differing incentives behave differently. Such differences in behaviour can lead to relevant interactions in governance as discussed above. This literature strongly suggests that empiricists should not expect institutional investors of different kinds to act identically as company stewards: the heterogeneity across blockholders provides a new focus for empirical analysis. We discuss empirical approaches to blockholder heterogeneity in Section 5.4, where we also outline specific ways in which the empirical literature has engaged with the theoretical predictions outlined here.

It is also worth noting that while the theoretical literature has, to date, only considered duallayered agency relationships, in reality the investment choices of institutional investors may create ownership chains characterized by *multi*-layered agency relationships. For example, as Figure 9 shows, US pension funds invest significantly in mutual funds. When they do so, they create (at least) a *triple-layered* agency relationship. Similar multi-layered agency relationships are created by investment in funds-offunds. The study of multi-layered agency relationships in the context of corporate governance represents a potential growth area for the theoretical literature.

4.3.5 Active and Passive Management

At a theoretical level, Corum *et al.* (2020) consider a different source of heterogeneity across institutional investors, namely, whether they are active or passive investors. Their paper sheds light on the empirical debate about the governance impact of the rise of passive investment vehicles (see Section 5). In contrast to the theoretical literature discussed above, which focuses on the incentives of active funds

to compete for flow, they look at reallocation of capital across three investment *modes*: private (nonintermediated) investment, investment in passive funds, and investment in active funds. The size of each sector and the fees paid to funds is endogenously determined. Corum *et al.* (2020) note that both the fund size (which determines exposure to the firm) and fund fees (which determines the fund's skin in the game) affect incentives to engage in the costly monitoring of portfolio firms. In this context, the expansion of low fee passive managers does not necessarily worsen governance. If, for example, the growth of low-fee passive funds occurs at the expense of private (non-intermediated) investment without affecting the size of the actively traded sector, then governance can improve because – despite their lower fees – the increasing size of passive funds may induce monitoring. If, on the other hand, the growth of low-fee passive funds occurs at the expense of active funds, then governance can decline, particularly so for undervalued firms in which actively managed funds try to invest.

Further subtle effects of the growth of indexing are examined by Levit *et al.* (2021). Their model explores the interaction of shareholder democracy (the determination of corporate policy by proxy voting) with share trading in a model with heterogeneous preferences. A key difference between political democracy and shareholder democracy is that in the latter the voters also trade, and trading affects the voter base. Because trading anticipates voting outcomes, but the outcome of trades affects the voting outcome, a feedback loop emerges that can introduce indeterminacy, and drive a wedge between prices and shareholder welfare (by making the marginal voter more extreme in their preferences than the average post-trade shareholder). In this context, Levit *et al.* (2021) argue that indexing – which reduces the degree of unrestricted share trading and thus the intensity of the feedback loop – can reduce indeterminacy, while having a non-monotone effect on firm value.

4.4 Theories on Institutional Investors and Proxy Voting Advisors

As noted in Section 3, the services of proxy voting advisors play a key role in the governance efforts of institutional investors. This role is unique to institutional investors as a result of at least two factors. The first factor stems from scale. Large institutional investors such as mutual fund companies hold shares in many companies, and must – by law – vote in a manner consistent with the best interest of their investors. This is a challenging task and has engendered a market for voting advice, leading to the rise of proxy voting advisors such as ISS and Glass Lewis. The second factor is stems from regulation. As discussed in Section 3, the mandated public reporting of both voting policy and votes actually cast by registered investment

companies since 2003 has increased scrutiny of institutional voting. This has fostered what legal scholars sometimes refer to as a "compliance challenge" for such institutional investors: it may simply be "safer" to outsource voting choices to external parties such as proxy voting advisors. In Section 5, we provide a careful discussion of the empirical evidence on the degree to which institutional investors rely on and are influenced by proxy voting advice.

The theoretical literature has lagged behind its empirical counterpart with regard to analysing the role of proxy advisors. However, the recent contribution of Malenko and Malenko (2019) provides a key analysis of several aspects of the economics of such information intermediaries. In their model, shareholders vote on a proxy proposal that may raise or lower firm value and can acquire information on the worth of such a proposal either by conducting (at some cost) private research or by paying a fee to a monopolistic proxy advisor. Shareholders then vote on the basis of this information and their votes determine the outcome and thus firm value. Therefore, in Malenko and Malenko's model, the availability of good information across shareholders translates into higher firm value. In this context, the presence of the proxy advisor fosters a key trade-off: on the upside, it provides the option to purchase an informative signal; on the downside, it reduces incentives to produce (independent) private signals about proposal value by conducting (independent) research, a "crowding out" effect. This matters because when shareholders undertake their own research, errors in their assessment of proposal value are uncorrelated with each other, and will wash out over a large number of shareholders. In contrast, if all or many shareholders purchase the same signal from a proxy voting advisor, any errors in the proxy advisor's assessment will be correlated across (all or many) shareholders, leading to a higher probability of incorrect voting outcomes. In other words, as Malenko and Malenko (2019) insightfully observe, for any given fee charged by the proxy advisor, there is a minimal level of precision of the information produced by the proxy voting advisor below which its presence is actually detrimental to firm value because the crowding out effect dominates. Further, if the proxy voting advisor strategically sets its fee to maximize profits, a second inefficiency arises. A proxy voting advisor with low quality information is likely to charge a low fee, enhancing the crowding out effect precisely when the quality of advice is poor. In contrast, a proxy voting advisor with high quality information may set a fee high enough to even ration access to its information, precisely when it is most valuable.

Malenko and Malenko (2019) also sheds light on two issues that arise prominently in our discussion in Sections 3 and 5. First, the "compliance challenge" created by the post-2003 regulatory

framework – as discussed in Section 3 – may make things worse: when institutional investors are motivated by a fear of litigation of regulatory censure, they are all the more likely to underinvest in private information production, exacerbating the crowding out problem. Second, the empirical literature discussed in Section 5 has focussed on the potential benefits of competition in the market for voting advice. However, holding the quality of advice constant, if competition lowers fees, it is again possible that the crowding out effect is enhanced.

As discussed in Section 3.2.1.2, there is growing concern amongst both institutional investors and regulators about the potential for conflicts of interest when proxy voting advisors make voting recommendations for proxy proposals at firms with which they have separate advisory relationships. The recent theoretical work of Levit and Tsoy (2021) speaks to this issue. They consider a setting in which an expert makes recommendations to multiple recipients. It is commonly known that the expert is biased because he prefers one (of several) recipient to act in a particular manner; but it is unknown *which* recipient forms the focal point of his bias. Such an expert is tempted to make one-size-fits-all recommendations – i.e., excessively correlated recommendations across disparate recipients – precisely in order to hide the identity of the recipient generating his bias. Since the advisory relationships of proxy voting advisors is not publicly known, such a model provides a lens through which to interpret the oft-criticized one-size-fits-all recommendations of proxy advisors. Empirical evidence on such one-size-fits-all recommendations is discussed in Section 5.5.3.

5. Empirical Literature on Institutional Investors and Corporate Governance

In this section, we provide a thematic overview of the empirical literature on institutional investors and corporate governance. We focus on two key ways in which institutional investors feature in the empirical literature. First, institutional investor data have been used to examine a number of classical theories of corporate governance. While such underlying theories – as noted in Section 4.1 – are not specific to institutional investors (and predate the institution-specific theoretical literature) the ready availability of data on institutional investors have meant that such theories have often been tested using institutional data. This literature is reviewed in Section 5.3. Following this, we focus on the strands of the literature that deal with institution-specific aspects of corporate governance. We discuss the role of institutional heterogeneity in Section 5.4, and subsequently discuss the role of institutional voting and proxy voting advisors in Section 5.5. As a key preliminary to all of this, we begin our review of the empirical literature by discussing in Section 5.1 the data sources and in Section 5.2 the methodologies that have been employed to examine governance effects of institutional investors.

5.1 Institutional Investor Data

5.1.1 Archival Data on Institutional Investors

Most papers relating institutional ownership to governance outcomes use Schedule 13F filings (see Section 3.2.1.3). These data are publicly available in the SEC's EDGAR database and compiled by Thomson Reuters.²⁹ While having been a valuable source of data for many contributions, a disadvantage of the data is that Schedule 13F reports are filed on a quarterly basis only, and that investors can request confidential treatment for certain holdings (Agarwal *et al.*, 2013). Further, the data can be filed up to 45 days after the quarter-end, which implies that the filings can reflect investments made more than four months ago. The limited frequency of 13F filings can be overcome with data from outside the US. For example, the Australian Portfolio Analytics Database contains information on Australian institutions' holdings on a *monthly* basis (it also contains daily data on their trading) – these data are used in Gallagher *et al.* (2013). Studies from outside the US typically use data from national registries to identify institutional holdings (e.g., Achleitner *et al.*, 2010 for Germany). A limitation of such studies is that these datasets often only contain ownership stakes exceeding a certain threshold. German law, for instance, only requires that owners holding 3% or more in a firm need to disclose their stakes (see Section 3.2.2.2).

Ferreira and Matos (2008) provide the first comprehensive analysis on institutional ownership around the world covering 27 countries and firms representing 40% of the world's market capitalization. The authors obtain their data from FactSet Ownership (previously FactSet/LionShares).³⁰ Their work triggered a series of other contributions using these data to examine the governance role of institutional

²⁹ Iliev *et al.* (2021) use data on investor views of EDGAR filings to show that many investors devote significant amounts of time reading governance topics in EDGAR filings.

³⁰ Data coverage in FactSet increased substantially in 1999 since the data provider enhanced the ownership information it collects from 35,000 stocks to 85,000 stocks internationally. FactSet collects these data from firm proxies, annual reports, stock exchange announcements, national regulatory agencies, local and offshore mutual funds, mutual fund industry directories, or institutions' webpages.

investors (e.g., Ferreira *et al.*, 2010; Aggarwal *et al.*, 2011; Erkens *et al.*, 2012; Bena *et al.*, 2017). Data on international institutional ownership is also available in Bureau van Dijk's Orbis database.³¹

Some studies examine daily trading data to capture how changes in institutional ownership affect voting outcomes (Bethel *et al.*, 2009) or hedge fund activism (Gantchev and Jotikasthira, 2018) – these data are obtained from the ANcerno database, which is used widely in asset pricing research (see Hu *et al.*, 2018 for an overview). Similar data are provided by Campbell *et al.* (2009) who infer daily institutional trading behavior from the Transactions and Quotes database of the NYSE.

US studies that focus on *specific* institutional investor types – notably hedge funds – typically use data from Schedule 13D filings in the EDGAR database (e.g., Holderness and Sheehan, 1985; Brav *et al.*, 2008; Klein and Zur, 2009 and Clifford, 2008). As explained in Section 3.2.1.3, investors owning more than 5% of a publicly traded firm have to file a Schedule 13D with the SEC within 10 days if they intend to influence a firm. Most studies examine Schedule 13D data on the institutions' ownership stakes (Item 5), the identity of the filers (Item 2), and the purposes for acquiring shares (Item 4). However, other details in Schedule 13D filings have also provided useful insights. First, Item 5(c) requires investors to report the quantity, price, and date of all trades in a target firm during the 60 days that precede the filing date. Collin-Dufresne and Fos (2015) and Gantchev and Jotikasthira (2018) use this information to study trading behavior of Schedule 13D filers. These data are particularly useful as they provide high-frequency information about the block-formation process and on the relation between trading and governance. Second, Item 6 requires the disclosure of any derivative contracts that have been entered by filers. Collin-Dufresne *et al.* (2021) show that derivatives are rarely used by 13D filers, suggesting that such products are unattractive for such informed investors and play a minor role in the trading strategies of activists.

As discussed in Section 3.2.1.3, investors acquiring more than 5% of a publicly traded firm who intend to remain *passive* with regard to governance must file a Schedule 13G with the SEC. Edmans *et al.* (2013) use Schedule 13D and Schedule 13G filings to study how stock liquidity affects the choice to file Schedule 13D versus Schedule 13G. More recently, Albuquerque *et al.* (2020b) model the choice of an investor between filing Schedules 13D and 13G and then use the model to estimate the expected returns to activist and passive investing.

³¹ These data are used relatively rarely in governance research (e.g., in Ginglinger *et al.*, 2017).

An alternative data source are proxy statements in which firms disclose their "principal" shareholders, that is, entities that are the beneficial owner of more than 5%. Proxy statements are usually a less relevant source of blockholder information, as information about principal shareholders is typically obtained from Schedule 13G and 13D filings (and amendments to these filings). However, proxy statements can become relevant as an additional data source when a group of shareholders ("dissidents") initiates a proxy contest and solicits votes in support of their director nominees or shareholder proposals excluded from the proxy statement filed by management. Since proxy contests are often initiated by shareholders who do not file Schedule 13Fs and own less than 5%, proxy statements can also provide information on the background and ownership of activists shareholders. In their proxy statement, dissidents need to disclosure their identities, intentions, and the number of shares they beneficially own. Fos (2017) uses these data to describe which shareholders stand behind proxy contests.³²

Another data source is ISS Voting Analytics, which is based on Form N-PX filings. US registered investment companies must file these forms with the SEC to disclose proxy votes and procedures in portfolio firms around the world (see Section 3.1.1). A useful feature of ISS Voting Analytics is that it contains, in additional to the voting records, the voting recommendations of the firm's management and of the proxy voting advisor ISS.³³

A benefit of archival data is that they allow researchers to establish broad evidence on whether institutional ownership, trading, or voting is associated with governance outcomes. The data also allow for the analysis of how dimensions of institutional investor heterogeneity – investor origins, horizon, or type – affect such associations. A limitation is that archival data do not necessarily capture *specific* channels through which institutional investors achieve beneficial governance outcomes. As a result, some researchers have turned to case study or survey data to fill this gap.

5.1.2 Case Study Data for Specific Institutional Investors

Case study-based research, often based on proprietary data, started in the mid-1990s with a contribution by Smith (1996) investigating 51 firms targeted by CalPERS, the largest US public pension

³² He shows that a particular class of institutional investors – activist hedge funds – sponsor more than 50% of proxy contests. In contrast, pension funds and mutual funds rarely sponsor a proxy contest, suggesting heterogeneous incentives to engage in hostile corporate governance.

³³ A comprehensive database on voting by non-US institutional investors around the world does not yet exist.

fund. The key benefit of such data is that they include details on the institution's private engagements (e.g., internal memos and reports) unobservable in archival data. Though only one institution is examined, the analysis is still insightful as CalPERS is considered a leader in shareholder activism.³⁴ A similar approach is employed in Carleton *et al.* (1998) using data from TIAA-CREF, another major US pension fund, to study private negotiations between the investor and 45 portfolio firms.

More than 20 years later, Becht *et al.* (2009) revived case-study based governance research by examining private interventions in 30 targets by the Hermes UK Focus Fund (HUKFF). The analysis is enriched by three cases studies that precisely illustrate how the investor engages portfolio firms (one is cited in Section 3.2.2.1). Becht *et al.* (2009) have access to HUKFF's correspondence with the target firms as well as to the internal details on organization's engagement processes. This enables them to identify the connections between the activist's engagement actions and subsequent stock return reactions – thereby mitigating a common concern in the literature, namely that firm value changes are not the result of the engagement, but instead reflect the stock picking ability of the activists.

Over the last years, the engagement activities of institutional investors branched out to incorporate also environmental and social topics, and this trend affected the academic literature. A first analysis using proprietary ESG engagement data is offered by Dimson *et al.* (2015) who examine engagement in 613 US firms – similar data is used in Barko *et al.* (2018) and Pezier (2020). While a focus of these studies is the stock return reactions to engagement, Hoepner *et al.* (2021) use data provided by Federated Hermes to examine how engagement affects downside risk.³⁵

While the papers in this section consider engagement by a single institutions, Doidge *et al.* (2019) combine public and proprietary information on *collective* activism by institutional investors that form part of the Canadian Coalition for Good Governance (CCGG) – a benefit of their data is the measurement of how investors coordinate engagement. A similar analysis if offered by Dimson *et al.* (2020) using coordinated engagement data from the Principles for Responsible Investment (PRI), an institutional investor coalition founded in 2006 with support from the United Nations.

³⁴ Another beneficial feature of the data in Smith (1996) is that he can use internal estimates by CalPERS of the costs of engagement to calculate a measure of the net benefit of activism.

³⁵ Federated Hermes is considered the most influential activist in terms of promoting ESG standards, and the authors use proprietary information on engagement activities and the investor's engagement success.

Case study data has benefits and costs. Apart from identifying the exact channels through which engagement affects firm value, an additional benefit is that they include the investor's measures of engagement success, usually in the form of some internally-tracked milestones.³⁶ A limitation is that general conclusions about the success rates of activism are harder to make when the data originate from one institution only. One reason is that the institutions that make their data available may be relatively successful activists; poorly performing institutions may be less willing to share their data.

5.1.3 Survey Data on Institutional Investors

A handful of papers use questionnaires among institutional investors to answer governance questions that are hard to address using archival or case study data.³⁷ McCahery *et al.* (2016) use a survey to understand how institutional investors engage with firms. A benefit of their survey is that it allows for a better understanding of governance activities that take place behind the scenes as opposed to publicly. Further, they provide evidence on specific predictions related to the (threat of) exit that cannot easily be tested using other forms of data. A related approach is used in Krueger *et al.* (2020) to understand whether and how institutional investors internally address climate risks.

The PRI now provides a publicly available survey-based database on institutional investors. Institutional investors that are signatories of the PRI are required to provide answers to an annual survey on various governance questions (e.g., proxy voting policies, shareholder engagement, ESG integration). As the responses are not anonymous, one can link the responses to investor characteristics. Gibson *et al.* (2020) provide a first analysis of these data. The richness of data suggests that more papers will follow.

A challenge in survey-based research is that the respondents may not be representative, which thus requires careful consideration of how response bias may affect the results. The reason is that investors with a high awareness of governance issues – and possibly with relatively high credentials in addressing them – are more likely to participate in such surveys. Nevertheless, understanding the preferences of such institutions is insightful as they are more likely to shape governance practices. To evaluate response bias, it is helpful to compare characteristics of responding and non-responding

³⁶ Researchers using archival data instead need to make assumptions on when an engagement is considered successful (e.g., if a shareholder proposal passes).

³⁷ This follows a trend in other areas of finance where surveys provided new insights into topics such as corporate financing (Graham and Harvey, 2001), capital allocation (Graham *et al.*, 2015), investor relations (Karolyi *et al.*, 2020), or ESG investing (Amel-Zadeh and Serafeim, 2018).

institutions. Another concern is that the respondents' answers are strategic or untruthful – to mitigate this concern, surveys usually guarantee that the answers are treated as being confidential.

5.2 Empirical Methods to Identify Governance Effects of Institutional Investors

A challenge in studying effects of institutional ownership on corporate governance outcomes is that both variables are affected by other variables that cannot be controlled for. This endogeneity makes it difficult to establish causal effects. In this section we discuss approaches used to tackle endogeneity.

5.2.1 Activism Campaign Announcements

A popular approach to study effects of institutional investors on firm value is the analysis of activism campaign announcements, with the goal of measuring the information effect of these event on the market prices of firms. Holderness and Sheehan (1985) adopt such an event study methodology to evaluate stock price reactions to Schedule 13D filings by six activist investors, Brav *et al.* (2008) and Klein and Zur (2009) study price reactions to Schedule 13D filings by activist hedge funds, and Fos (2017) evaluates the effects of proxy contest announcements. All these papers document positive stock price reactions to activism campaign announcements.

The literature considers two explanations for why activism campaign announcements are associated with positive announcement returns: i) market participants anticipate that activist investors will increase future cash flows by influencing the firm's corporate policies; or ii) activist investors are better at picking undervalued stocks, that is, at least part of these announcement returns are unrelated to the activist campaign. To distinguish between these explanations, Brav *et al.* (2015b) explore the stock market reactions when activists switch at the *same* target firm from a passive Schedule 13G filing to an active Schedule 13D filing. The idea is that the value attributable to stock picking should be obtained when a Schedule 13G is filed, while the value of a governance treatment is identified with the stock return around the announcement when the investor switches from a Schedule 13G to a Schedule 13D. Albuquerque *et al.* (2020b) rely on structural estimation to separate value creation and stock picking – they model the

choice of an investor between filing a Schedule 13D or 13G in order to estimate expected returns to activist and passive investing.³⁸

5.2.2 Discontinuities in Voting Outcomes

Institutional investors often sponsor shareholder proposals to influence firms' governance practices. If these proposals enhance firm value, one would expect a positive stock price reaction to the passage of such proposal. An obvious endogeneity concern related to such an analysis is that the passage of a proposal is anticipated by markets and in turn should not move prices. Cuñat *et al.* (2012) overcome this concern by comparing the market reaction to proposals on governance topics that pass by a small margin to those that fail by a small margin – the identification assumption is that for "close" proposals, the outcome is unanticipated by the market, and thus price reactions capture causal value enhancement.³⁹ Bach and Metzger (2019) cast some doubts on the validity of this approach by showing that close votes are disproportionately more likely to be won by management than by activist shareholders – 75% more shareholder proposals were rejected by a thin margin of 1% of outstanding shares than proposals approved by the same margin.

5.2.3 Index Decompositions

The idea to use index inclusions to obtain some level of exogenous variation in institutional ownership originates from Clay (2000, 2002). He exploits the addition of firms to the S&P 500 as a shock to institutional ownership to then estimate how the ownership changes affect CEO pay and firm value. Subsequent papers using the MSCI All-Country World Index and the S&P 500 include Ferreira and Matos (2008) and Aghion *et al.* (2013), respectively. The appealing idea of this approach is that institutions that track an index are forced to invest in a firm that gets added to the index, which in turn causes an increase in institutional ownership ("relevance condition"). This comes with the assumption that a firm's index inclusion affects governance outcomes *only through* its effect on institutional ownership ("exclusion restriction"). A violation of this restriction may arise if the inclusion is based on a firm's stock market performance, which in itself could either reflect or affect governance outcomes. Further, index inclusion

³⁸ Albuquerque *et al.* (2020b) find that more than 75% of Schedule 13D announcement returns can be attributed to value creation. When they consider a subsample of filings by activist hedge funds, more than 92% of the Schedule 13D announcement returns can be attributed to value creation.

³⁹ Cuñat *et al.* (2012) show that governance proposal by shareholders that pass yield an abnormal return of 1.3% relative to those that fail. Flammer (2015) uses a similar approach for E&S proposals.

announcements – in particular for the S&P 500 – likely increase media and investor recognition, which possibly confounds the governance treatment of the inclusions.

Starting with Chang *et al.* (2015), a series of papers expanded and improved on this early work by exploiting variation in institutional ownership around the size *cutoffs* used to calculate whether or not a firm is an index member. These contributions focus on the Russell indexes and exploit the fact that portfolio weights within the Russell 1000 and the Russell 2000 indexes are value-weighted, so that the smallest Russell 1000 stocks have small index weights, while the largest Russell 2000 stocks have large index weights (they are about ten times larger). This implies that, for every dollar invested in the two indexes, much is invested in stocks at the top of the Russell 2000, and little is invested in stocks at the bottom of the Russell 1000. Thus, such differences in index weights lead to higher institutional ownership in firms at the top of the Russell 2000, and lower institutional ownership in firms at the bottom of the Russell 2000, and lower institutional ownership in firms at the bottom of the Russell 2000, and lower institutional ownership in firms at the bottom of the Russell 2000, and lower institutional ownership in firms at the bottom of the Russell 1000. In this approach it is less likely that the exclusion restriction is violated, as size or performance differences above versus below the Russell 1000/2000 threshold are close to random.

Chang *et al.* (2015) do not consider governance outcomes, but a series of studies have built on their approach by empirically testing how governance outcomes are affected by Russell-related variation in institutional ownership. These papers use either a regression-discontinuity design (RDD) or an instrumental variables (IV) framework. RDD studies exploit the differences in institutional ownership of firms just above (bottom of the Russell 1000) and below the threshold (top of the Russell 2000) (Boone and White, 2015). Somewhat differently, IV studies instrument institutional ownership with an indicator that captures whether a firm i) switches between the Russell 1000 and the Russell 2000 (Fich *et al.*, 2015 or Schmidt and Fahlenbrach, 2017); or ii) is assigned to the Russell 2000 (Appel *et al.*, 2016 or Crane *et al.*, 2016).⁴⁰

5.2.4 Unexpected Mutual Fund Outflows

Index inclusions are less suitable in studies that use high frequency data on ownership changes by institutional investors as inclusions take place very infrequently. Gantchev and Jotikasthira (2018) therefore take a different approach to address the concern that firm-level changes in institutional

⁴⁰ Wei and Young (2019) criticize the use of Russell index reconstitution arguing that a problem with prior studies is that they exhibit selection bias. Notably, they show that firms close to, but on opposite sides of, the Russell 1000/2000 cutoffs already have different levels of institutional holdings before the index reconstitution takes place.

ownership are endogenous. They use trading in other stocks, outside of the firm's industry, as an instrument for institutional ownership. This approach builds on the idea that institutions experiencing large in- or outflows often scale their stock positions proportionally up or down (see Coval and Stafford, 2007; Edmans *et al.*, 2012). Relatedly, Michaely and Vincent (2013) use changes in institutional ownership that originate from mutual fund outflows to obtain exogenous variation.⁴¹

Two studies raise concerns about this approach. Berger (2019) argues that selection bias drives changes in corporate policies for firms affected by mutual fund outflows. She shows that firms affected by outflows alter financial policies compared to control firms even when no mispricing event occurs. Wardlaw (2020) shows that the standard approach to computing outflow-induced price pressure, which scales flow by the dollar volume of trading in a stock, produces a measure that is a direct function of a stock's realized return. He argues that this raises concerns about the exogeneity of the outflow measure as it violates the exclusion restriction for outcome variables such as firm valuation and performance.⁴²

5.3 Institutional Investors as Blockholders: Classical Evidence on Voice and Exit

We set the stage for our discussion of empirical results on institutional investor heterogeneity by briefly reviewing the evidence related to classical theories of blockholder governance. Guided by the theoretical literature reviewed in Section 4.1, we divide the discussion into evidence on "governance via voice" and "governance via exit." Results that speak directly to heterogeneity across institutional investors are discussed in Section 5.4.

5.3.1 Evidence on Governance via Voice

Empirical research testing classical theories of blockholder governance by institutional investors typically examine shareholder proposals or direct engagement. Interestingly, the initial evidence on shareholder proposals paints a rather disappointing picture of institutional investors' ability to increase firm value through this channel. Notably, Karpoff *et al.* (1996) document that shareholders proposals –

⁴¹ An alternative approach is used by Kempf *et al.* (2017), who consider a shock to the monitoring abilities of institutional investors instead of exploiting variation in their holdings. They thereby exploit the distraction of an institutional investor owning a specific firm that arises from unrelated industry performance shocks in other portfolio firms. A similar setting is used in Liu *et al.* (2020) to show that board oversight weakens when institutional investor are distracted.

⁴² A way to address Wardlaw (2020)'s concerns is by scaling flows by market capitalization instead of trading volume.

while triggered by poor performance – do not improve firm performance or valuations (even if proposals get a lot of support). Similarly, Gillan and Starks (2000) demonstrate that proposals sponsored by institutional investors appear to have no effects on firm value, though such proposals gain substantially more support than proposals sponsored by individuals.

While these (and other) early contributions provide little evidence for value increases, studies using more recent or larger data provide a brighter picture.⁴³ An influential results is the finding in Cuñat *et al.* (2012) that proposals that pass by a small margin earn an abnormal announcement return compared to those that fail by a small margin (see Section 5.2.3). These returns are higher among firms with larger institutional ownership, and for proposals sponsored by institutions. Flammer (2015) demonstrates that the adoption of "close call" E&S proposals generates positive announcement returns and higher firm performance. He *et al.* (2020) find that even though E&S proposals nearly always fail, investor support for these proposals contains information regarding future firm risks.

Evidence on *direct* engagement comes from three broad categories: hedge fund activism, individual institutions, and surveys. Within the first category, Brav *et al.* (2008) document positive abnormal returns of around 7% around the announcement of Schedule 13D filings by hedge funds (with no evidence of a reversal). The largest returns arise if the objective of the activist is the sale of the target or a change in business strategy. Klein and Zur (2009) provide similar evidence with a focus on confrontational campaigns and Greenwood and Schor (2009) confirm that returns are largely driven by firms that are eventually taken over.⁴⁴ Becht *et al.* (2017) show that the target stock price increase most in response to activism if engagements yield positive real outcomes. Further, they find that activism is generally highest among firms with high institutional ownership, particularly for US institutions. Bebchuk *et al.* (2020) study settlements achieved by hedge fund activists with target firms to document that such arrangements are accompanied by positive stock price reactions.

In the second category, Smith (1996) demonstrates that firm value increases when firms targeted by CalPERS agree to the investor's demands and decreases if they resist. Carleton *et al.* (1998) find high

⁴³ This is possibly the result of several factors, including a broader understanding of the importance of blockholder governance, the rise in institutional ownership, innovation in the shareholder activism space, and advances in empirical methodology. Denes *et al.* (2017) provide a review of this literature.

⁴⁴ A comprehensive survey of the literature on hedge fund activism is provided in Brav *et al.* (2010).

success rates of engagements by TIAA-CREF when proposing changes in targeted firms. Similarly positive results come from Becht *et al.* (2009) who find that most private engagements by HUKFF increase firm value. Related evidence on ESG engagement is provided by Dimson *et al.* (2015) and Barko *et al.* (2018) for returns, and by Hoepner *et al.* (2021) for downside risk. Two key conclusions emerge from these papers: first, engagement by individual institutions increases value, and second, public engagement channels – such as shareholder proposals – may not be required if private engagements are successful.

In the third category, McCahery *et al.* (2016) document widespread use of behind-the-scenes engagement among the respondents to their survey, highlighting the usefulness of surveys or case studies to understand the governance by institutions. Their findings also confirm that institutions first try to engage privately with portfolio firms and only take a public route when such private engagements fail.

Several papers examine the channels through which the documented value increases arise. Two salient channels appears to be executive pay and turnover. Hartzell and Starks (2003) and Mullins (2014) document that institutional ownership relates positively to incentive pay and negatively to pay levels, while Ertimur *et al.* (2011) find that shareholder activists tend to target firms with high CEO pay. Further, voting support for activists' shareholders proposals on pay is higher if CEO pay was more excessive or less related to performance.⁴⁵ Del Guercio *et al.* (2008) find that firms experience a rise in forced CEO turnover when institutional investors withhold votes towards directors' elections. Further, Kang *et al.* (2018) demonstrate that the number of blocks held by firm's large institutions positively relates to the sensitivity of forced CEO turnover to firm performance.

A number of papers relate institutional ownership to payout, financing and investment policies. Grinstein and Michaely (2005) cannot detect that higher institutional ownership leads to increases in dividends, repurchases or total payout. However, using Russell index reconstitutions, Crane *et al.* (2016) revisit this finding to show that firms pay out more when they institutional ownership rises.⁴⁶ Higher institutional ownership is also related to lower leverage (Michaely and Vincent, 2013), higher innovation

⁴⁵ However, Fernandes *et al.* (2013) find that an increase in institutional ownership, specifically of foreign-based institutions, is associated with higher pay and higher equity compensation for CEOs. It also leads to a convergence of pay to US level.

⁴⁶ Brav *et al.* (2008) find for hedge funds that activism leads to an increase in target firms' payouts.

(Aghion *et al.*, 2013), and better M&A (Fich *et al.*, 2015). Further, Brav *et al.* (2015b) show that, following interventions by activist hedge funds, the plants of targets experience an increase in productivity.

Two recent papers show that firms change corporate policies also when they realize that the *likelihood* of intervention increases. Fos (2017) shows that when the probability of a proxy contest increases, firms reduce cash holdings, invest less, increase leverage, and increase payouts. Gantchev *et al.* (2019) demonstrate that the effects of hedge activism reach beyond actual targets, as non-targeted peer firms make improvements similar to those of the target firms under the threat of activism (they increase leverage and payout, and reduce investments and cash). Thus, threats of a proxy contest or of hedge fund activism appear to reduce agency costs of free cash flow (Jensen, 1986).

5.3.2 Evidence on Governance via Exit

The evidence supports existing theories modelling (the threat of) exit as an important governance mechanism utilized by institutional investors. A challenge for establishing such evidence is the difficulty of empirically capturing the exit *threat* which is inherently unobservable. To deal with this challenge, researchers have attempted to identify circumstances under which the threat is likely to be stronger.

The first empirical analysis of the importance of the exit mechanism dates back to Parrino *et al.* (2003) who show that CEO turnover increases after selling by investors.⁴⁷ The work of Parrino *et al.* (2003) predated the emergence of the theoretical literature on exit, which began with the work of Admati and Pfleiderer (2009) and Edmans (2009).The emergence of the theoretical literature on exit sparked a series of new papers taking specific predictions of these models to the data.

Edmans *et al.* (2013) document for hedge funds that governance through the exit threat increases when the liquidity of a stock is higher – this effect is stronger, as predicted by theory, when executive pay is more sensitive to the stock price. Edmans *et al.* (2013) measure the threat of governance through exit based on whether an institution has filed a Schedule 13G, that is, a filing that explicitly signals that an investor will not use direct activism (it would otherwise file a Schedule 13D), leaving the exit threat as the only remaining governance tool. Similar evidence is provided by Bharath *et al.* (2013) who document that

⁴⁷ Helwege *et al.* (2012) show that the importance of exit as a driver of CEO turnover decreased over time, while activism by institutional investors instead emerged as a driver of turnover.

firms with larger blockholdings experience increases in value when stock liquidity increases. Again, this effect is strongest if executive pay is more sensitive to the stock price, a situation in which the exit threat becomes more credible. Dimmock *et al.* (2018) exploit the role of a potential tax liability on capital gains accrued on a stock. Such liability makes it more costly for an institutional investor to sell a stake in a firm, which plausibly reduces the credibility of an exit threat. Indeed, greater capital gain lock-ins reduce the likelihood of exit being used as a governance tool (intervention becomes more likely).

McCahery *et al.* (2016) use their survey to directly ask institutional investors how plausible they consider the exit mechanism to be. They provide four key findings. First, the investors consider exit as a practically useful strategy, with 49% (39%) responding that they used the exit mechanism in the past. Second, 42% believe that the *threat* of exit is effective when it comes to disciplining firm management. Third, the respondents state that exit threat effectiveness depends on whether other investors also exit for the same reason, the equity stake size of the investor, managerial ownership, and whether other large shareholders are also present – the importance of the exit decision by others is consistent with Cvijanovic *et al.* (2019). Fourth, exit is seen as a governance mechanisms that complements voice (rather than substituting it), and intervention typically happens prior to a potential exit.

Three papers empirically examine the interaction of voice and exit as well as how the interplay between different institutions affects the exit effectiveness. Duan and Yao (2016) consider mutual funds' choice between exit and voice, showing that funds with smaller stakes and shorter horizons prefer exit, and that exit occurs more frequently among small, liquid firms with large inside ownership. Becht *et al.* (2019) use proprietary data from a UK asset manager to document that "Against" votes at the shareholder meeting by the asset manager (voice) leads to stock selling (exit), and even more so when the asset managers internal analysts downgraded the stock. Building on their own model (see Section 4.3.4), Cvijanovic *et al.* (2019) examine how institutional investors react to each other's exit. They document first evidence for correlated exit, that is, the result that following an exit by an activist hedge fund, (flow-motivated) mutual funds sell out of the target firms significantly more than other institutional investors.

5.4 Institutional Investor Heterogeneity

Having evaluated the broad evidence on the role of institutional investors for exit and voice, we next highlight findings indicating differences across institutional investor types in the governance of firms,

building on the key themes emerging from Sections 2, 3, and 4 above. We focus on differences in governance incentives, location, horizon, passive holdings, and business ties and relate – whenever possible – the evidence to the theoretical models in Section 4.2 and 4.3.

5.4.1 Institutional Investor Governance Incentives

The theoretical literature in Section 4.2 shows how heterogeneity across institutional investors – in terms of skin in the game and incentives to compete for flow – affect the governance role of institutional investors. Some early evidence in support of specific model predictions of this literature is provided by McCahery *et al.* (2016) who document that institutional investors are concerned that clients make negative inferences about their stock selection ability if an institution uses governance via exit. This directly supports the predictions of Dasgupta and Piacentino (2015) who show theoretically that the fear of outflows weakens the credibility of the exit threat for sufficiently flow motivated institutions.

Another contribution is Israelsen *et al.* (2019) who examine heterogeneity in incentives and skills across different block ownership forms. Specifically, they compare "committed blockholders" who likely have skin in the game (individuals or private equity firms) with "financial blockholders" who care mostly about fund flows (institutions that file Schedule 13Fs). Israelsen *et al.* (2019) show that committed blockholders are significantly more likely to "self-identify" as activists by filing a Schedule 13D in comparison to financial blockholders. Thus, consistent with the predictions of Dasgupta and Piacentino (2015), reduced skin in the game and enhanced incentive to compete for flow may weaken incentives to govern. Interestingly, they also find that firms with committed blocks appear to have *worse* governance structures. However, this result may arise due to an equilibrium effect: firms with committed shareholders have lower monitoring needs to begin with. External governance mechanisms beyond committed ownership may not be needed to the same extent as in firms with financial blockholders.

Lewellen and Lewellen (2021) provide an important building block to study flow motivations in governance by quantifying institutional investors' financial incentives to be engaged shareholders. They measure an institution's incentive to engage as the increase in the investor's cash flow (i.e., management fees) when a firm's stock increases by 1% in value. Notably, they consider the *direct* effect on assets under management as well as the *indirect* effect on subsequent fund flows – the latter in turn depends on relative performance across funds and thus on the degree to which a stock is over- versus underweight in the

institution's portfolio relative to peers. This measure should be helpful for more papers to shed light on why differences in incentive structures across institutional investors are important. While the quantification of flow-based incentives is an important step, it is noteworthy that the theoretical literature suggests that flow motivations have a nuanced effect on governance incentives, which can be either negative by inducing short-termism (Dasgupta and Piacentino, 2015; Burkart and Dasgupta, 2021) or positive by inducing intervention, particularly in multiple blockholder contexts (Song, 2017; Brav *et al.*, 2021). Thus, the quantification of flow-based incentives represents only a first step towards evaluating the role of institutional incentives to govern.

5.4.2 Conflicted versus Non-Conflicted Institutional Investors

While the theoretical literature has focused on heterogeneity in incentives arising from skin in the game and competition for flow, a key additional source of heterogeneity arises from the degree to which institutional investors have direct conflicts of interest arising from business ties to portfolio firms (see Section 3.3). To identify the role of such conflicts of interests, existing studies either classify institutional investor very broadly or use data on their actual business ties with portfolio firms.

Using the first approach, Ferreira and Matos (2008) define mutual funds and investment advisers as "independent institutions" and compare them with bank trusts, insurance companies, and other institutions, which they classify as "grey institutions." The idea is that independent institutions have fewer potential business ties with portfolio firms and, as a result, are involved in monitoring firm management. Using this classification, Ferreira and Matos (2008) as well as Aggarwal *et al.* (2011) demonstrate that independent institutional investors are more active in positively shaping firms' governance structures.

Using the second approach, Davis and Kim (2007) use data on mutual fund proxy votes to show that fund voting is not influenced by business ties for any given shareholder proposal, though they do find that fund families that have higher business ties overall tend to vote in a more pro-management manner across all portfolio firms irrespective of individual business ties. However, they use only six types of shareholder proposals. Ashraf *et al.* (2012) use data on executive pay proposals and reach similar conclusions. Using a much larger data set of proposals, Cvijanovic *et al.* (2016) challenge these results by showing that business relationships with portfolio firms are associated with pro-management proxy voting by mutual fund families. The richness of their dataset makes it possible to robustly control for unobserved heterogeneity, enabling them to link pro-management voting with business ties at the level of given pairs of firms and fund families and for individual proposals. They also find that the effect is driven by shareholder-sponsored – as opposed to management-sponsored – proposals, thus suggesting that conflicts of interest prevalent among a particularly important class of institutional investors may actually impede, rather than facilitate, good corporate governance. Finally, the effect is strongest for proposals that pass or fail by modest margins, that is, in cases where the voting of individual fund families may potentially have decisive impact. In a similar vein, Hamdani and Yafeh (2009) find – using Israeli data – suggest that institutional investors owned by public companies provide greater support to management-sponsored proposals for those companies.⁴⁸ Bolstering the narrative of conflicted voting by mutual fund families captured in these papers, Cohen and Schmidt (2009) find that trustee fund families overweight their portfolio holdings in sponsor firms and are reluctant to exit in response to negative shocks.

As discussed above, a useful analysis is also offered by Agrawal (2012) who studies proxy votes of AFL-CIO union funds to show that some investors pursue other interests (here workers), rather than shareholder value alone. Specifically, he finds that AFL-CIO-affiliated investors are less opposed to directors when the AFL-CIO stops representing a firm's workers. Further, similar voting patterns cannot be found for public pension funds as well as mutual funds.

5.4.3 Long-term versus Short-term Institutional Investors

The empirical analysis of how institutional investor horizon affects firm outcomes is complicated by several factors. On a theoretical (or conceptual) dimension, the challenge is that investor horizons are not fundamental characteristics of institutional investors, but rather endogenous consequences of their incentives. As discussed in Section 4.3.3, the incentive to compete for flow may endogenously induce short-termism. Hence, the degree to which institutions have locked up capital (e.g., hedge funds may have more locked-up capital than mutual funds), patient final investors (e.g., institutional funds may differ in their responsive to short-term performance than retail funds), and skin in the game may lead to different degrees of short-termism. Investor horizons are also endogenous consequences of firm performance.

⁴⁸ The recent work of Dressler (2020) provides further backing for the "agency conflicts" view of institutional voting in the Israeli context, documenting that the greater the voting power of institutional investors, the lower the likelihood that they vote against management-sponsored proposals. The finding of promanagement voting behavior amongst Israeli institutional investors is reinforced in a subsample of delayed votes, suggesting that pre-vote negotiations may play a meaningful role.

Whether institutions are short term or long term often depends on how well the investment in a firm performs – if the firm does well, institutional investors will retain their stakes and become long-term investors, if the firm does poorly, they will sell their stakes and become short-term investors (see Edmans, 2009). A similar endogeneity holds for other firm outcomes, including corporate policies or governance practices. More broadly, there is not only an "influence effect" of investor horizon on firm outcomes but also a confounding "selection effect," that is, institutions with certain horizons may prefer to invest in firms with certain characteristics.⁴⁹

On a measurement dimension, the challenge is that one ideally wants to use an *ex-ante* measure reflecting how long an institution *intends to hold* a specific firm's equity stake, rather than an *ex-post* proxy reflecting the past (realized) investment horizon of the institution as whole. Specifically, the past overall trading frequency of an institution may not be informative of its intention for how long to hold a stake in a given firm. For how long an institutions plans to hold a stake likely depends on the factors mentioned above as well as its perception of the resolution of any mispricing in a stock. The considerations along both dimensions show that it is important to empirically account for selection effects and the endogeneity in investor horizon. If not possible, one should carefully interpret any established relations between investor horizon proxies and governance outcomes, especially when *ex-post* measures are used.

The early empirical papers on institutional investor horizons significantly predate the theoretical literature on blockholder incentives surveyed in Section 4 and focuses – due to obvious data constraints – on *ex-post* measures of investor horizons.⁵⁰ Specifically, this literature make use of Schedule 13F data in order to capture investor horizon. Bushee (1998) measures short-term institutional ownership using ownership by transient investors, which are institutions with high portfolio turnover and diversified portfolios. He differentiates these investors from "quasi-indexer" institutions with low turnover and diversified portfolios. Gaspar *et al.* (2005) instead measure investor horizon using the churn rate or portfolio turnover of an institutional investor. A higher churn rate reflects that an institution turns over its holdings more frequently. Firms with higher ownership by institutional investors with high churn rates (i.e., high "fund

⁴⁹ For example, Starks *et al.* (2021) shows that institutional investors with long horizons prefer high-ESG firms, while short-term investors prefer the opposite.

⁵⁰ More recent papers have to some extent inherited the definitions of short-termism used in early papers in the literature.

turnover") are in turn defined as having more short-term institutional owners. A benefit of the measure is that it facilitates the interpretation of results obtained with it, as it does not conflate turnover and portfolio diversification (both variables in theory affect governance outcomes).

Early evidence on the effects of short-term investors is provided by the finding in Bushee (1998) that firms with more short-term institutions spend less on R&D (especially in cases where cuts to R&D can reverse a decline in earnings). Recent evidence in support of this result comes from Cremers *et al.* (2020) who demonstrate that an increase in short-horizon institutional ownership due to a firm's addition to the Russell 2000 index leads to cuts to long-term investment. Related evidence is provided by the finding in Derrien *et al.* (2013) that greater long-term ownership (measured using fund turnover) is associated with higher corporate investments when firms are valued lower than predicted based on firm fundamentals.

Other contributions indicate that the effects of short-term investors extend beyond R&D. For example, firms with more short-term investors perform worse in corporate takeovers, both as targets and acquirers (Gaspar *et al.*, 2005; Chen *et al.*, 2007), exhibit more fraud and empire building (Harford *et al.*, 2018), and use share buybacks more frequently to payout out more to shareholders (Gaspar *et al.*, (2013). Using Bushee (1998)'s classification, Borochin and Yang (2017) find that higher ownership by dedicated institutions is associated lower executive pay (which they interpreted as being better from a governance perspective), while the opposite holds when transient investors is higher. A more positive assessment of short-term institutional ownership emerges from Giannetti and Yu (2020) who find that firms with more short-term owners are better able to adapt to changes in their business environments via innovation or management changes.

These results provide important insights into our understanding of the role of heterogeneity in investor horizons. Yet, the growth of the theoretical literature in this area offers potential new ways to formalize and measure institutional short-termism. It thereby represents a potential area of growth for this literature and offers opportunities to reinterpret existing empirical results.

5.4.4 Passive versus Active Institutional Investors

The role of passive investors has significantly increased over the last ten years due to a sharp increase in index investing and ETFs. This shift in institutional ownership has potentially major effects on the governance of firms, as index funds and ETFs cannot use the threat of exit and as some passive investors may be reluctant to engage in costly governance due to their low-fee business models. Despite these important considerations, relatively little empirical research exists to this date. In fact, the few studies that exist provide conflicting evidence on the governance effects of passive institutional ownership.

On the dark side, Schmidt and Fahlenbrach (2017) document that higher index fund or ETF ownership leads to more CEO power, fewer independent directors on the board, and worse corporate M&A. Heath *et al.* (2021) show that index funds rarely vote against management on contentious governance issues, and when dissatisfied with management, they do not use exit to express their dissatisfaction. They also find no evidence that index funds engage with firm management. Boone *et al.* (2020) document that the big passive fund families, BlackRock, State Street, and Vanguard, are increasingly likely to vote with management, especially in case of controversial proposals. Gutiérrez and Philippon (2018) show that firms owned by many passive investors (measured using ownership by quasi-indexers) tend to underinvest relative to their investment opportunities. In a recent paper, Bennett *et al.* (2020) demonstrate that the previously documented positive announcement returns of being added to the S&P 500 has disappeared, and argue that this is caused by the anticipation of negative governance effect due the increasing (decreasing) amount of passive (active) ownership after firms are added to the index.

In contrast and on the bright side, Appel *et al.* (2016) demonstrate that an increase in passive mutual fund ownership leads to more independent directors, the removal of antitakeover mechanisms, and more equal voting rights. Further, Appel *et al.* (2019) find that shareholder activism is generally more successful when a large fraction of the firm is owned by passively managed mutual funds. Schmidt and Fahlenbrach (2017) reconcile these findings with theirs by arguing that passive ownership has positive governance effects when low-cost governance activities are concerned (e.g., voting), while it has negative effects for high-cost governance (e.g., monitoring of corporate M&A activities). Filali Adib (2020) builds on this debate to zero in on the voting channel to better understand how index funds affect corporate governance. She finds that the market reaction to a proposal's passage (failure) is stronger if a more index funds vote to support (oppose) it and that higher ownership by index funds makes the passage of value-enhancing proposals more likely. This finding lends support to a positive role of index funds in low-cost governance activities (in this case voting).

One consequence of the increase in passive ownership is that holdings by a small number of investors, mostly providers of index funds, has increased significantly across most firms. This has caused a new phenomenon labelled "common ownership," whereby the same set of investors may have significant stakes in firms that compete on the product market.⁵¹ We do not delve into this literature as Schmalz (2018) provides a recent review of these papers.

In both the areas of active versus passive ownership and the effects of common ownership discussed in this subsection, there is a significant gap between the theoretical and empirical literatures. In the case of the former, the only paper that considers passive *and* active institutional investors simultaneously within the same model is Corum *et al.* (2020), which underscores the relatively subtle interaction between the governance roles of these classes of investors. In the case of the latter, a paper that considers common ownership by institutional investors – Edmans *et al.* (2019) – demonstrates the positive role of common ownership on voice and exit in general, but does not consider the product market interactions that have been the focus of the majority of empirical papers in this area. Further, Anton *et al.* (2020) show theoretically that common ownership, specifically between product market competitors that strategically interact, can in fact make bad governance optimal. Thus, there is significant scope for the interface of theory and empirics in this area.

5.4.5 Interplay between Institutional Investors

The stylized facts in Section 2 suggest the presence of multiple institutional blocks per firm, both within and across types of institutional investors. This suggests that the interplay between institutions, both within and across types, is key to corporate governance. A series of papers look at how the interplay between different institutional investors affects governance outcomes. Focusing within the class of mutual funds, Matvos and Ostrovsky (2010) consider voting in director elections. Apart from finding some that funds are consistently more management friendly than others, they provide evidence for peer effects, that is, funds tend to oppose management when other funds also do so. Bebchuk *et al.* (2020) hand collect

⁵¹ This has sparked an extensive debate about whether common ownership has anticompetitive effects. For example, using data on the airline industry, Azar *et al.* (2018) show that prices are higher and output is lower when firms have low incentives to compete due to common ownership. Dennis *et al.* (2019) challenge these results and argue that they are generated by the endogenous market share component of the common ownership measure. Lewellen and Lowry (2021) argue that many of the effects attributed to common ownership originate from other factors, such as differential responses of firms or industries to the global financial crisis, and Geng *et al.* (2021) find that common ownership can be beneficial by promoting innovation.

data on voting outcomes in proxy contests and look at the interplay between activist hedge funds and other institutional investors. They use this data to examine voting support by institutional investors when there is a proxy fight triggered by a dissident activist and find a positive relation between the propensity for activist targeting and pro-activist voting by mutual funds. Similar evidence on the importance of having a shareholder base of activism friendly institutions is provided by Brav *et al.* (2020). Kedia *et al.* (2021) demonstrate that investor dissatisfaction with management is associated with an increased probability of becoming an activism target. Crane *et al.* (2019) show that cliques of institutional investors (those connected through the network of institutional holdings) tend to vote together on proxy items, increasing the likelihood of votes against management. Fos (2017) finds that the presence of institutional and activist investors makes it more likely that a firm might be the target of proxy contests (which delivers positive abnormal returns upon its announcement).

A particularly prominent type of interplay between institutional investors arises in the form of socalled "wolf packs," a phenomenon by which several activist hedge funds engage in parallel at the same target firm, complementing each other's efforts while remaining sufficiently independent to avoid a joint 13D filing. A small number of recent empirical papers has examined the wolf pack phenomenon. Becht *et al.* (2017) document that as many as 20% of hedge fund activism events involve multiple activists intervening in parallel. Notably, wolf packs increase the probability of a successful engagement and they are associated with higher announcement returns when stakes are disclosed. More recently, Artiga Gonzalez and Caluzzo (2019) also show that wolf packs are more common in larger firms, where individual activists are unlikely to have sufficient influence.

Some recent evidence also highlights that blockholder diversity can have negative effects. Schwartz-Ziv and Volkova (2020) demonstrate that firms held by a blockholder base that is heterogeneous consistently perform worse than firms owned by homogeneous blockholders. They argue that this is the result of diverging views, preferences, and objectives among heterogeneous blockholders.

There is an emerging dialogue between the theoretical and empirical literatures in the area of the interplay between institutional investors – the relevant theoretical papers are discussed in Section 4.2. For example, the findings of both Becht *et al.* (2017) and Artiga Gonzalez and Caluzzo (2019) are broadly consistent with the model of Brav *et al.* (2021). Some recent theoretical papers have also started to consider the interaction between different types of institutional investors (Song, 2017 and Cvijanovic *et*

al., 2019). Given the subtlety of potential interactions between institutional investors, this is an area in which the dialogue between theoretical and empirical work is particularly salient.

5.4.6 Institutional Investor Location

Another dimension of heterogeneity relates to the origin of institutions, which can affect the cultural or social values promoted by institutions. Work in this area, pioneered by Miguel Ferreira and Pedro Matos in a series of articles (some of them with coauthors), demonstrates that the origin of investors matters strongly for governance outcomes. In Ferreira and Matos (2008), higher foreign ownership, particularly from the US, is associated with higher firm valuations and better performance, and these effects originate from foreign institutions' stronger monitor of corporate executives.⁵² Further, the evidence in Aggarwal *et al.* (2011) reveals that the steady increase in foreign ownership due to the globalization of capital markets leads to the adoption of better governance practices across the world. Bena *et al.* (2017) find that foreign institutional ownership significantly increases long-term investments (capital expenditures and R&D) and innovation output. This evidence rejects the view that foreign institutions are "locust" that reap short-term gains only. Luong *et al.* (2017) reach similar conclusions.

Further evidence on the role of foreign institutional investors is provided in Leuz *et al.* (2008) and their observation that foreign institutions leave firms that do not improve their governance. Dyck *et al.* (2019) find that institutional investors increase firms' E&S performance when they originate from countries with norms that emphasize the importance of E&S issues. Ilhan *et al.* (2021) demonstrate that institutions from countries with high norms and stewardship codes demand more climate-related disclosure from their portfolio firms. Finally, Gibson *et al.* (2020) document that institutional investors from Europe are more likely to be signatories of the PRI, which promotes high governance (and E&S) standards around the world. In countries other than the US, such PRI signatories have better portfolio-level ESG footprints than non-PRI signatories.

Kim *et al.* (2016) show that domestic institutional investors may also be good for governance in some settings, as they show that higher domestic ownership is associated with less earnings management. Yet, they also show that foreign institutions become more effective as monitors if the firm environment reflects greater agency conflicts or weaker governance controls. Ng *et al.* (2016) study the impact of

⁵² Similar evidence on the beneficial role of international institutional ownership is provided by Chan *et al.* (2009).

international investor origin on stock market liquidity, which affects governance as it makes the threat of exit more credible. They separate foreign ownership into foreign direct ownership (strategic foreign holdings larger than 5%) and foreign portfolio ownership (other institutional holdings). While strategic foreign ownership negatively relates to liquidity, foreign portfolio ownership seems to benefit liquidity. In their study, foreign institutional ownership is also positively associated with firm valuations.

Instead of using heterogeneity along the international origin of investors, Chhaochharia *et al.* (2012) look at the distance between a firm's headquarters and the location of the firm's institutional investors. They document that firms with more local institutional ownership exhibit better internal governance, are less likely to aggressively manage their earnings, and are genuinely more profitable.

5.4.7 Institutional Investor Ideology

Two recent papers examine heterogeneity across institutional investor along the political or governance spectrum. Bolton *et al.* (2020) map mutual funds and pension funds into the political left-right spectrum based on their proxy voting, using an innovative approach from political science. While far-left investors are characterized as "socially responsible" investors (they vote most consistently for social and environment shareholder proposals), far-right investors are characterized as "money-conscious" investors (they oppose proposals that might be financially costly to shareholders). They additionally also separate institutions along a governance dimension, based on the degree to which they believe discipline should be imposed on management. In the spirit of this second dimension, Bubb and Catan (2021) use voting data by mutual funds to classify them into three "parties" with different governance "philosophies": the Traditional Governance Party (the firm should be run by the board and management, not shareholders), the Shareholder Intervention Party (shareholders should intervene through proposals and proxy fights), and the Shareholder Veto Party (shareholders should monitor management and veto actions proposed by management when they have concerns).

5.5 Proxy Voting Advisors and Institutional Investors

5.5.1 Importance of Proxy Voting Advisors

As discussed in Section 3, proxy voting is important for regulatory and fiduciary duty, and many institutions are required (or voluntarily choose) to disclose their proxy votes, as well as their voting

policies. Proxy voting entails costs, particularly since many investors have to cast votes on thousands of securities. As a result, institutional investors increasingly make use of proxy voting advisors, along with information from sell-side and in-house analysts, when deciding on how they want their shares voted. The proxy advisory industry has therefore grown substantially over the past decade and proxy voting advisors provide a service to address many institutional investor's regulatory requirement to vote shares in portfolio firms in the best interest of their final beneficiaries. McCahery *et al.* (2016) provide survey evidence to examine the extent to which institutional investors use proxy voting advisors. They show that 60% of respondents to their survey make use of at least one proxy voting advisor, and almost half of these respondents use the proxy services of more than one advisor.

Two firms dominate the market for proxy voting advice. ISS, the market leader, was founded in 1985 and is majority owned by Deutsche Börse (since 2021). ISS employs 2,000 employees and covers about 44,000 annual meetings in 2020 worldwide. Its main competitor, Glass Lewis, is a portfolio firm of the Alberta Investment Management Corporation and the Ontario Teachers' Pension Plan Board and was founded in 2003. It employs more than 360 employees. Both firms make voting recommendations based on general voting guidelines, which are usually for specific countries.⁵³ Some early evidence on how firms try to game the specific rule that are described in these guidelines is provided in Ishida and Kochiyama (2020). Other firms offering proxy voting advice include US firms Egan-Jones Proxy Services, Segal Marco Advisors, and ProxyVote Plus; Swiss firms SWIPA and Ethos; UK firm Minerva; and the French firm Proxinvest. All of these "boutique" firms are substantially smaller than the "Big Two."

Shu (2020) provides the first estimates of the market shares of proxy advisory firms. He estimates that in 2017, ISS controlled 63% of the market for mutual funds in the US and Glass Lewis controlled 28%. Interestingly, Shu's estimates indicate that the proxy advice market has become less concentrated over the last decade, with ISS gradually losing market share to Glass Lewis and other smaller proxy advisors.

⁵³ See <u>https://www.issgovernance.com/policy-gateway/voting-policies/</u> and

<u>https://www.glasslewis.com/guidelines/</u>. The proxy voting advisors ask and receive feedback from firms and investors on their drafts for these proxy voting guidelines.

5.5.2 Influence of Voting Recommendations

A key question relevant to investors, firms, and regulators is whether and when recommendations of proxy voting advisors have an effect on voting outcomes. Some of the first evidence on the role of proxy voting advisors is provided by Choi *et al.* (2008, 2010) who study the effects of voting recommendations for uncontested director election at S&P 1500 firms. Choi *et al.* (2010) document that ISS is more influential than Glass Lewis in terms of effects on voting outcomes (their analysis also includes other proxy voting advisors, namely Proxy Governance and Egan Jones). Further, they estimate that an ISS recommendation shifts 6 to 10% of shareholder votes. The estimates originate from the years 2005 and 2006, so they are likely to provide a lower bound, since the services of proxy voting advisors have been used by more institutional investors since then. Further early empirical evidence is provided by Cotter *et al.* (2010) who study voting during the 2003 to 2008 proxy season. A benefit of their analysis is that they study a broad range of management and shareholder proposals. They find that mutual funds appear to vote consistently with recommendations by ISS, and more so than other shareholders. Brav *et al.* (2020) show how ISS recommendations serve as a coordination for concentrating withholding on the focal proxy voting candidate, which effectively helps dissidents win without confronting management.

Ertimur *et al.* (2013) use hand-collected data to study the role of proxy voting advisors for "say on pay" votes in the year 2011. Say-on-pay votes are nonbinding votes on executive pay that started to be required as a result of the *Dodd–Frank Act* in 2011. In their sample, ISS issued "Against" recommendations for 11% of the firms and Glass Lewis for 22%. Though say-on-pay plans are rarely voted down (only at 2% of firms), a negative ISS recommendation is associated with 24.7% (12.9%) more against votes (12.9% in case of Glass Lewis). They further find that the sensitivity of shareholder votes to "against" recommendations is higher if block ownership by institutional investors (ownership by institutions that hold more than 5%) is larger. Larcker *et al.* (2015) also study the role of proxy voting advisors for say-on-pay votes. Consistent with Ertimur *et al.* (2013), they find that the recommendations of proxy voting advisors are positively related to voting outcomes. Their evidence also indicates that several firms change their compensation plans *before* the annual meeting in a way that is consistent with the preferences of the proxy voting advisors. Shu (2020) also confirms such influence using very recent data.

Regressing the votes cast by institutions on the recommendations of proxy voting advisors may overstate the effect of proxy voting advice, as voters and advisors may independently come to the same

conclusion on how to vote. In other words, the same unobservable firm characteristics that lead proxy voting advisors to give a certain voting recommendation can affect institutions' actual voting, leading to an upward bias in a regression estimate. Thus, while the studies reviewed above show a positive correlation between proxy voting advisors' recommendations and voting outcomes, it remains unclear whether this means that institutions follow ISS recommendations or, instead, whether proxy voting advisors to institutions' voting preferences.

Two studies have made progress towards better understanding whether recommendations cause voting outcomes or coincide with investors preferences. Iliev and Lowry's (2015) analysis exploits the fact that different funds have different incentives to "actively" vote, i.e., to independently assess the issues up for vote. Incentives are higher if, for example, an investor has invested a higher percentage of fund assets in a portfolio firm. Consistent with variations in incentives for active voting across investors, they find also high variation in investors' reliance on proxy voting advisors. While more than 25% of mutual funds rely almost entirely on recommendations by ISS, others put only little weight on them. ISS recommendations have little predictive power in explaining the voting by investors that have strong incentives to actively vote but high predictive power when these incentives are low.

Malenko and Shen (2016) use a RDD framework to gauge how ISS recommendations effect sayon-pay voting outcomes. Specifically, to identify a causal effect, they exploit variation in ISS recommendations due to a rule in ISS's voting guidelines on say-on-pay proposals in the years 2010-2011. Under this rule, the ISS conducted an initial screen of firms focusing on their one- and three-year shareholder returns relative to certain cutoffs, and it then performed deeper analyses of the firms' pay practices only among firms ranked below the cutoff. Hence, firms below the cutoff were subject to greater scrutiny to achieve a favorable ISS recommendation (compared to firms above the cutoff). Malenko and Shen (2016) show a strong effect of ISS recommendations: relative to positive ISS recommendations, negative recommendations by ISS lead to a 25 percentage point decrease in voting support for say-on-pay proposals. Hence, they conclude that ISS moves about 25% of the votes. There is also evidence that voting recommendations vary significantly across the big two proxy voting advisors, as shown in Ertimur *et al.* (2013) for say-on-pay votes. While most papers focus on the effect of proxy voting advisors on US firms, Hitz and Lehmann (2018) study European firms. Their findings suggest that proxy voting advisors findings suggest also play an important role at European firms.
Two recent papers provide new evidence on how the role of proxy voting advisors has evolved over time. Aggarwal *et al.* (2014) find that investor voting on shareholder proposals has become more independent of ISS recommendations over time (2004-2010), but only in proposals where ISS recommends a vote against the proposal. Boone *et al.* (2020) examine the time period from 2005 to 2018 and also find that ISS has become less influential as mutual funds are less likely to follow the ISS recommendations.

Overall, these results suggest that the role of proxy voting advisors is not simply in aggregating shareholder preferences or in coinciding with them. Instead, the evidence suggest that proxy voting advisors actively influence voting decisions of institutional investors. However, there is significant heterogeneity in the degree to which institutions are influenced by proxy voting advice.

5.5.3 Determinants and Quality of Voting Recommendations

The evidence suggests that proxy voting advisors do have some influence on voting outcomes. This raises the question of how their recommendations are determined and how good they are. On this topic, views are divided. One view holds that proxy voting advisors offer institutional investors informed and reliable voting advice. The services of proxy advisors in turn reduce voting costs of institutional investors, which should lead to overall better voting decisions. To the contrary, the alternative view holds that the proxy voting advisors' recommendations ignore firm-specific circumstances and are too standardized. Further, their recommendation criteria entail a lack of transparency, which in turns makes it difficult to assess the quality of voting recommendation. This raises the question of what actually drives the recommendations made by proxy voting advisors.

lliev and Lowry (2015) provide some evidence that indicates that ISS uses a one-size-fits-all approach (blanket recommendations) during their sample period from 2006 to 2010. The reason is that on many governance and pay proposals, ISS almost always recommends voting against firm management without accounting for firm-specifics. To the contrary, actively voting funds seem to be more likely to vote in a firm-specific manner. A more positive picture emerges from Ertimur *et al.* (2013) who find that proxy voting advisors are more likely to issue an "against" recommendation on say-on-pay proposals at firms with poor performance and higher levels of CEO pay, indicating that firm-specific circumstances are taken into account. The mixed view is also reflected in the survey evidence of McCahery *et al.* (2016). While they document that 55% of their survey respondents either agree or strongly agree that proxy voting advisors

are useful by helping them vote in a more informed way, they also find that 30% agree or strongly agree that proxy voting advisors' advice is too standardized.

Albuquerque *et al.* (2020a) examine compensation proposals to show that the quality of the voting advice by ISS depends on whether ISS faces a heavy workload or not; they find that high quality recommendations – defined as those for which ISS releases "Against" recommendations and negative assessments are associated with lower subsequent firm performance – occur only during the off season (that is, for firms where the fiscal year does not end in December).

Apart from firm-specific determinants of vote recommendations, public opinion also appears to play a role. For example, Aggarwal *et al.* (2014) show that ISS voting recommendations are associated with public confidence in banks and (negative) newspaper articles on executive compensation. They argue that this finding reflects that ISS uses a proxy policy formulation process that incorporates feedback from an annual survey of market participants.

Heinen *et al.* (2018) examine whether US-based proxy voting advisors "export" US corporate governance by not sufficiently considering local settings when making recommendations for non-US firms. They document that the voting recommendations diverge more between foreign and local proxy voting advisors than among foreign proxy advisors. Further, "Against" recommendations by local proxy voting advisors have an incremental effect on voting outcomes beyond the recommendations by foreign proxy voting advisors.

5.5.4 Firm and Market Reactions to Voting Recommendations

Several studies also examine how firms and stock prices react to recommendations by proxy voting advisors. As most proxy votes are non-binding, it is not *ex ante* obvious that the effect of proxy voting advisors on voting outcomes translates into real changes at firms. Alexander *et al.* (2010) study market reactions around recommendations by ISS issued during proxy contests involving board seats between 1992 and 2005. They find that the announcement of ISS recommendations creates abnormal stock returns at firms, which indicates that their views bring new information to the market. Ertimur *et al.* (2013) find that firms that receive a negative say-on-pay vote that was trigger by a negative recommendation from proxy voting advisors engage with investors and make changes to their compensation plans. They also find

negative abnormal returns (-0.5% to -0.7%) in the case of "Against" recommendations by ISS for firms where such a recommendation was unexpected by the market.

Larcker *et al.* (2015) find negative announcement returns for compensation plans that were adjusted before a shareholder vote to reflect the preferences of proxy voting advisors. This indicates that proxy voting advisors may trigger firms to make choices that are not beneficial to shareholders. Calluzzo and Dudley (2019) provide evidence that proxy voting advisors influence real outcomes through their recommendations, notably, director behavior, executive retention, and executive compensation practices. Guest and Nerino (2020) find that ISS has an effect beyond its influence via proxy recommendations and subsequent voting outcomes. Specifically, they show that downgrades in the governance ratings provided by ISS are associated with negative announcement returns of -1.14% over a three-day window.

5.5.5 Conflicts of Interest

Another issue surrounding the use of proxy voting advisors is whether they are exposed to conflicts of interests when developing their recommendations. Notably, ISS not only sells advice about how investors in portfolio firms should vote, but it also consults firms on how to improve their corporate governance practices – a concern is that this dual role may lead to voting recommendations tilted by conflicts of interest (e.g., Yermack, 2010). Glass Lewis does not provide such consulting services to firms. The empirical analysis of these potential conflicts of interest is still in its infancy. Yet, the results in McCahery *et al.* (2016) indicate that institutional investors are concerned to some degree about conflicts of interests, as 52% of their respondents agree or strongly agreeing that proxy voting advisors are exposed to conflicts of interest. Li (2018) shows that the entrance of Glass Lewis into the market for proxy voting advice in early 2003 has reduced the scope for conflicts of interests (i.e., favoritism by ISS). Specifically, he shows that ISS started to issue more pro-shareholder recommendations after the market entry by Glass Lewis, especially when Glass Lewis did not support the shareholder proponents.

6. Conclusions

We provide a comprehensive overview of the role of institutional investors in corporate governance. The first component of our analysis traces the emergence of institutional investors as concentrated owners of public firms in modern economies, using a wide variety of data sources to establish new stylized facts. Subsequently, we provide a detailed characterization of key aspects of the current legal and regulatory setting within which institutional investors operate with respect to the governance of their portfolio firms. Finally, our review synthesizes the recent academic literature in finance on the role of institutional investors in corporate governance, attempting to link theoretical predictions to empirical findings.

In this concluding section, we reflect on the state of the literature on the governance role of institutional investors and outline a few thoughts on areas for potential research. There is now a significant literature, both theoretical and empirical, focusing directly on the role of institutional investors in corporate governance. The empirical literature is larger, and much of it predates the relatively new theoretical literature. But the theoretical literature is now growing and beginning to gain significant ground. In the process, it is providing foundations for economically relevant metrics and tests in empirical analysis via predictions regarding the role of institutional incentives. This in turn presents rich opportunities for further research in this area and guidance for how to examine the effects of institutional ownership.

At the broadest level, the heterogeneity that we document and highlight with regard to institutions' governance obligations, abilities, and incentives suggests that empirical research should reconsider how to account for institutional ownership. In the majority of governance studies, it is conventional to simply control for a catch-all category of total institutional ownership. Our examination of the data, the law, and the emerging academic literature suggests that it may be more informative to employ alternative metrics reflecting specific institutional characteristics, building on theoretical predictions. For example, a given degree of concentrated ownership by regulated, flow sensitive retail mutual funds is likely to differ significantly in its governance impact relative to a similar degree of ownership by unregulated hedge funds with sophisticated or locked-in investors. Similarly, focus should not be limited to sizeable (5%+) institutional blockholders: smaller blockholders may have incentives – deriving from the delegated nature of institutional blockholding – to bolster each other's engagement efforts. In essence, institutional heterogeneity matters, and must be accounted for.

Turning to specifics, our discussion throughout Section 5 suggests that there is significant variation across different strands within the literature with respect to the integration of theoretical and empirical analyses. In certain areas, there is a significant and growing dialogue between the theoretical and empirical literatures. For example, on the fundamental issue of how institutional incentives arising from competition for flow and skin in the game affect governance actions (as discussed in Section 5.4.1) there is both established and emerging empirical evidence in support of theoretical predictions. There are also emerging empirical techniques that provide the prospect of deeper investigations in this area, and also call for richer theoretical models. Similarly, in the area of the interplay between multiple institutional blockholders in determining governance outcomes in a given firm (as discussed in Section 5.4.5) there are significant points of contact between the theoretical and empirical literatures. This is true both in the context of engagements by multiple blockholders of the same type, for example, wolf packs of activist hedge funds and in the context of interactions across different types of institutional investors. Further, the emerging empirical evidence on how institutional investors interact may guide further theoretical development in this area.

In other areas, there is less dialogue between the theoretical and empirical literatures, representing significant potential for further work. For example, the significant empirical literature on the horizons of institutional investors treats such horizons as a primitive characteristic of institutions. The beginnings of this literature significantly predate the existence of theory in this area. Partly as a result, the empirical literature has arrived at several parallel ways of classifying institutions according to their horizons. Ultimately, the degree to which institutional investors are short-term or long-term, however, must depend on the nature of their incentives – for example, the degree to (and the frequency at) which they must compete for investor flow as well as the extent to which their skin in the game fosters long-term incentives. The growing theoretical literature on such incentives, discussed in Section 4.3, provides opportunities for better founded measures of investor horizons and thus provides an opportunity to significantly enrich this area of the literature.

In contrast, in some other areas, the theoretical literature is sparse and there is significant guidance in the empirical literature to expand and extend theoretical analyses. This is particularly evident in the area of active versus passive management, where the theoretical literature is in its infancy, while the empirical literature is engaged in a lively debate. A similarly lively debate exists in the empirical literature on common ownership, but there is currently little by way of a systematic approach within the theoretical literature to speak to this debate. The dynamic rise of passive institutional owners also raises the question of whether "established" empirical facts from prior studies on the relation between institutional ownership and firm outcomes remain valid in the current environment. Finally, while there is

some theoretical analysis of the incentives of proxy voting advisors, the empirical literature that addresses these issues is very significant and provides opportunities for further theoretical work.

The phenomenal growth of the asset management industry has transformed the nature of equity ownership across developed economies. A majority of blockholders in publicly traded firms are today institutional investors who are themselves agents acting on behalf of their capital providers. This places a multi-layered agency problem at the heart of modern corporate governance and calls for an enriched academic approach to its analysis. Academic researchers have responded to this significant change, leading to the development of both a theoretical and empirical literatures in this area. The academic literature on the role of institutional investors in corporate governance is poised at a critical stage of development. In our view, the key to further progress is an active two-way dialogue between the theoretical and empirical literatures. The emergence and enrichment of such dialogue holds the promise to transform the way in which economists view corporate governance in the modern economy.

References

- Achleitner, A.-K., A. Betzer, and J. Gider. 2010. "Do Corporate Governance Motives Drive Hedge Fund and Private Equity Fund Activities?". *European Financial Management*. 16(5): 805–828.
- Adams, R.B., B.E. Hermalin, M.S. Weisbach. 2010. "The Role of Boards of Directors in Corporate Governance: A Conceptual Framework and Survey". *Journal of Economic Literature*. 48 (1): 58-107
- Admati, A. R., and P. Pfleiderer. 2009. "The "Wall Street Walk" and Shareholder Activism: Exit as a Form of Voice". *Review of Financial Studies*. 22(7): 2645–2685.
- Agarwal, V., N. D. Daniel, and N. Y. Naik. 2009. "Role of Managerial Incentives and Discretion in Hedge Fund Performance". *Journal of Finance*. 64(5): 2221–2256.
- Agarwal, V., V. Fos, and W. Jiang. 2013. "Inferring Reporting-Related Biases in Hedge Fund Databases from Hedge Fund Equity Holdings". *Management Science*. 59(6): 1271–1289.
- Agarwal, V., W. Jiang, Y. Tang, and B. Yang. 2013. "Uncovering Hedge Fund Skill from the Portfolio Holdings They Hide". *Journal of Finance*. 68(2): 739–783.
- Aggarwal, R., I. Erel, M. Ferreira, and P. Matos. 2011. "Does Governance Travel around the World? Evidence from Institutional Investors". *Journal of Financial Economics*. 100(1): 154–181.
- Aggarwal, R., I. Erel, and L. T. Starks. 2014. "Influence of Public Opinion on Investor Voting and Proxy Advisors". *Working Paper*.
- Aghion, P., J. Van Reenen, and L. Zingales. 2013. "Innovation and Institutional Ownership". *American Economic Review*. 103(1): 277–304.
- Agrawal, A. K. 2012. "Corporate Governance Objectives of Labor Union Shareholders: Evidence from Proxy Voting". *Review of Financial Studies*. 25(1): 187–226.
- Albuquerque, A. M., M. E. Carter, and S. Gallani. 2020a. "Are ISS Recommendations Informative? Evidence from Assessments of Compensation Practices". *Working Paper*.
- Albuquerque, R. A., V. Fos, and E. J. Schroth. 2020b. "Value Creation in Shareholder Activism: A Structural Approach". *Working Paper*.
- Alexander, C. R., M. A. Chen, D. J. Seppi, and C. S. Spatt. 2010. "Interim News and the Role of Proxy Voting Advice". *Review of Financial Studies*. 23(12): 4419–4454.
- Amel-Zadeh, A., and G. Serafeim. 2018. "Why and How Investors Use ESG Information: Evidence from a Global Survey". *Financial Analysts Journal*. 74(3): 87–103.
- Andonov, A., Y. V. Hochberg, and J. D. Rauh. 2018. "Political Representation and Governance: Evidence from the Investment Decisions of Public Pension Funds". *Journal of Finance*. 73(5): 2041–2086.

- Anton, M., F. Ederer, M. Giné, and M. C. Schmalz. 2020. "Common Ownership, Competition, and Top Management Incentives". *Working Paper*.
- Appel, I. R., T. Gormley, and D. Keim. 2016. "Passive Investors, Not Passive Owners". *Journal of Financial Economics* 121(1): 111–141.
- Appel, I. T. Gormley, and. D. Keim. 2019. "Standing on the Shoulders of Giants: The Effect of Passive Investors on Activism". *Review of Financial Studies*. 32(7): 2720–2774.
- Artiga Gonzalez, T., and P. Calluzzo. 2019. "Clustered shareholder activism". *Corporate Governance: An International Review*. 27(3): 210-225.
- Ashraf, R., N. Jayaraman, and H. E. Ryan. 2012. "Do Pension-Related Business Ties Influence Mutual Fund Proxy Voting? Evidence from Shareholder Proposals on Executive Compensation". *Journal of Financial and Quantitative Analysis.* 47(3): 567–588.
- Azar, J., M. C. Schmalz, and I. Tecu. 2018. "Anticompetitive Effects of Common Ownership". *Journal of Finance* 73(4): 1513–1565.
- Bach, L., and D. Metzger. 2019. "How Close Are Close Shareholder Votes?". *Review of Financial Studies*. 32(8): 3183–3214.
- Back, K., P. Collin-Dufresne, V. Fos, and T. Li. 2018. "Activism, Strategic Trading, and Liquidity". *Econometrica*. 86(4): 1431-1463.
- Backus, M., C. Conlon, and M. Sinkinson. 2021. "Common Ownership in America: 1980 2017". American Economic Journal: Microeconomics. forthcoming.
- Barko, T., M. Cremers, and L. Renneboog. 2018. "Shareholder Engagement on Environmental, Social, and Governance Performance". *Working Paper*.
- Bebchuk, L. A., and S. Hirst. 2019. "Index Funds and the Future of Corporate Governance: Theory, Evidence, and Policy". *Columbia Law Review*. 119(8): 2029–2146.
- Bebchuk, L. A., A. Brav, W. Jiang, and T. Keusch. 2020. "Dancing with Activists". *Journal of Financial Economics.* 137(1): 1–41.
- Bebchuk, L. A., A. Cohen, and S. Hirst. 2017. "The Agency Problems of Institutional Investors". *Journal of Economic Perspectives*. 31(3): 89–102.
- Becht, M., J. Franks, J. Grant, and H. F. Wagner. 2017. "Returns to Hedge Fund Activism: An International Study". *Review of Financial Studies.* 30(9): 2933–2971.
- Becht, M., J. Franks, C. Mayer, and S. Rossi. 2009. "Returns to Shareholder Activism: Evidence from a Clinical Study of the Hermes UK Focus Fund". *Review of Financial Studies*. 22(8): 3093–3129.

- Becht, M., J. R. Franks, and H. F. Wagner. 2019. "Corporate Governance Through Exit and Voice". *Working Paper*.
- Bena, J., M. A. Ferreira, P. Matos, and P. Pires. 2017. "Are Foreign Investors Locusts? The Long-Term Effects of Foreign Institutional Ownership". *Journal of Financial Economics*. 126(1): 122–146.
- Bennet, B., R. M. Stulz, and Z. Wang. 2020. "Does Joining the S&P 500 Index Hurt Firms?". Working Paper.
- Berger, E. 2019. "Selection Bias in Mutual Fund Flow-Induced Fire Sales: Causes and Consequences". *Working Paper*.
- Bethel, J. E., G. Hu, and Q. Wang. 2009. "The Market for Shareholder Voting Rights around Mergers and Acquisitions: Evidence from Institutional Daily Trading and Voting". *Journal of Corporate Finance*. 15(1): 129–145.
- Bharath, S. T., S. Jayaraman, and V. Nagar. 2013. "Exit as Governance: An Empirical Analysis". *Journal of Finance*. 68(6): 2515–2547.
- Black, B. S. 1990. "Shareholder Passivity Reexamined". Michigan Law Review 89(3): 520-608.
- — . 1992. "Agents Watching Agents: The Promise of Institutional Investor Voice". UCLA Law Review.
 39(4): 811-893.
- Bolton, P., T. Li, E. Ravina, and H. Rosenthal. 2020. "Investor Ideology". *Journal of Financial Economics*. 137(2): 320–352.
- Boone, A. L., S. Gillan, and M. Towner. 2020. "The Role of Proxy Advisors and Large Passive Funds in Shareholder Voting: Lions or Lambs?". *Working Paper*.
- Boone, A. L., and J. T. White. 2015. "The Effect of Institutional Ownership on Firm Transparency and Information Production". *Journal of Financial Economics*. 117(3): 508–533.
- Borochin, P., and J. Yang. 2017. "The Effects of Institutional Investor Objectives on Firm Valuation and Governance". *Journal of Financial Economics.* 126(1): 171–199.
- Bradley, M., A. Brav, I. Goldstein, and W. Jiang. 2010. "Activist Arbitrage: A Study of Open-Ending Attempts of Closed-End Funds". *Journal of Financial Economics*. 95(1): 1–19.
- Brav, A., A. Dasgupta, and R. D. Mathews. 2021. "Wolf Pack Activism". Management Science. forthcoming.
- Brav, A., W. Jiang, and H. Kim. 2010. "Hedge Fund Activism: A Review". *Foundations and Trends in Finance.* 4: 185-246.
- Brav, A., W. Jiang, and H. Kim. 2015a. "Recent Advances in Research on Hedge Fund Activism: Value Creation and Identification". *Annual Review of Financial Economics.* 7: 579–595.
- ———. 2015b. "The Real Effects of Hedge Fund Activism: Productivity, Asset Allocation, and Labor Outcomes". *Review of Financial Studies*. 28(10): 2723–2769.

- Brav, A., W. Jiang, F. Partnoy, and R. Thomas. 2008. "Hedge Fund Activism, Corporate Governance, and Firm Performance". *Journal of Finance*. 63(4): 1729–1775.
- Brav, A., and R. D. Mathews. 2011. "Empty Voting and the Efficiency of Corporate Governance". *Journal of Financial Economics*. 99(2): 289–307.
- Brav, Al., W. Jiang, T. Li, and J. Pinnington. 2020. "Picking Friends Before Picking (Proxy) Fights: How Mutual Fund Voting Shapes Proxy Contests". *Working Paper*.
- Brown, K. C., W. V. Harlow, and L. T. Starks. 1996. "Of Tournaments and Temptations: An Analysis of Managerial Incentives in the Mutual Fund Industry". *Journal of Finance*. 51(1): 85–110.
- Bubb, R., and E. Catan. 2021. "The Party Structure of Mutual Funds". *Review of Financial Studies.* forthcoming.
- Buchanan, B., J. Netter, A. Poulsen, and Y. Tina. 2012, "Shareholder Proposal Rules and Practice: Evidence from a Comparison of the United States and United Kingdom". *American Business Law Journal*. 49(4): 739–803.
- Burkart, M., and A. Dasgupta. 2021. "Competition for Flow and Short-Termism in Activism". *Review of Corporate Finance Studies*. 10(1): 44-81.
- Burkart, M., D. Gromb, and F. Panunzi. 1997. "Large Shareholders, Monitoring and the Value of the Firm". *Journal of Corporate Finance*. 16(4): 443–455.
- Burkart, M., and S. Lee. 2021. "Activism and Takeovers". Review of Financial Studies. forthcoming.
- Bushee, B. J. 1998. "The Influence of Institutional Investors on Myopic R&D Investment Behavior". *The Accounting Review.* 73(3): 305–333.
- Calio, J. E., and R. X. Zahralddin. 1994. "The Securities and Exchange Commission's 1992 Proxy Amendments: Questions of Accountability". *Pace Law Review*. 14(2): 459.
- Calluzzo, P., and E. Dudley. 2019. "The Real Effects of Proxy Advisors on the Firm". *Financial Management*. 48(3): 917–943.
- Campbell, J. Y., T. Ramadorai, and A. Schwartz. 2009. "Caught on Tape: Institutional Trading, Stock Returns, and Earnings Announcements". *Journal of Financial Economics*. 92(1): 66–91.
- Carleton, W. T., J. M. Nelson, and M. S. Weisbach. 1998. "The Influence of Institutions on Corporate Governance through Private Negotiations: Evidence from TIAA-CREF". *Journal of Finance*. 53(4): 1335–1362.
- Champ, N. 2012. "Speech by SEC Staff: What SEC Registration Means for Hedge Fund Advisers". New York City.

- Chan, K., V. Covrig, and L. Ng. 2009. "Does Home Bias Affect Firm Value? Evidence from Holdings of Mutual Funds Worldwide". *Journal of International Economics.* 78(2): 230–241.
- Chang, Y.-C., H. Hong, and I. Liskovich. 2015. "Regression Discontinuity and the Price Effects of Stock Market Indexing". *Review of Financial Studies*. 28(1): 212–246.
- Chen, X., J. Harford, and K. Li. 2007. "Monitoring: Which Institutions Matter?". *Journal of Financial Economics.* 86(2): 279–305.
- Chevalier, J., and G. Ellison. 1997. "Risk Taking by Mutual Funds as a Response to Incentives". *Journal of Political Economy*. 105(6): 1167–1200.
- Chhaochharia, V., A. Kumar, and A. Niessen-Ruenzi. 2012. "Local Investors and Corporate Governance". *Journal of Accounting and Economics.* 54(1): 42–67.
- Choi, S. 2000. "Proxy Issue Proposals: Impact of the 1992 SEC Proxy Reforms". *Journal of Law, Economics, and Organization.* 16(1): 233–268.
- Choi, S. J., J. E. Fisch, and M. Kahan. 2008. "Director Elections and the Influence of Proxy Advisors". *Working Paper*.
- ———. 2010. "The Power of Proxy Advisors: Myth or Reality?". *Emory Law Journal*. 59(4): 869–918.
- Clay, D. G. 2000. "The Effects of Institutional Investment on CEO Compensation". Working Paper.
- Clay, D. G. 2002. "Institutional Ownership and Firm Value". Working Paper.
- Clifford, C. P. 2008. "Value Creation or Destruction? Hedge Funds as Shareholder Activists". *Journal of Corporate Finance*. 14(4): 323–336.
- Collin-Dufresne, P., and V. Fos. 2015. "Do Prices Reveal the Presence of Informed Trading?". The Journal of Finance. 70(4): 1555–1582.
- Collin-Dufresne, P., V. Fos, and D. Muravyev. 2021. "Informed Trading and Option Prices: Theory and Evidence from Activist Trading". *Journal of Financial and Quantitative Analysis*. forthcoming.
- Cohen, L., and B. Schmidt. 2009, "Attracting Flows by Attracting Big Clients". *The Journal of Finance*. 64(5): 2125–2151.
- Corum, A. A., and D. Levit. 2019. "Corporate Control Activism". *Journal of Financial Economics*. 133(1): 1– 17.
- Corum, A. A., A. Malenko, and N. Malenko. 2020. "Corporate Governance in the Presence of Active and Passive Delegated Investment". *Working Paper*.
- Cotter, J., A. Palmiter, and R. Thomas. 2010. "ISS Recommendations and Mutual Fund Voting on Proxy Proposals". *Villanova Law Review.* 55(1): 1–56.

- Coval, J., and E. Stafford. 2007. "Asset Fire Sales (and Purchases) in Equity Markets". *Journal of Financial Economics.* 86(2): 479–512.
- Crane, A. D., A. Koch, and S. Michenaud. 2019. "Institutional Investor Cliques and Governance". *Journal of Financial Economics.* 133(1): 175–197.
- Crane, A. D., S. Michenaud, and J. P. Weston. 2016. "The Effect of Institutional Ownership on Payout Policy: Evidence from Index Thresholds". *Review of Financial Studies*. 29(6): 1377–1408.
- Cremers, M., A. Pareek, and Z. Sautner. 2020. "Short-Term Institutions, Analyst Recommendations, and Mispricing". *Management Science*. 66(10): 4535–4551.
- Cuñat, V., M. Gine, and M. Guadalupe. 2012. "The Vote Is Cast: The Effect of Corporate Governance on Shareholder Value". *Journal of Finance*. 67(5): 1943–1977.
- Cvijanović, D., A. Dasgupta, and K. E. Zachariadis. 2016. "Ties That Bind: How Business Connections Affect Mutual Fund Activism". *Journal of Finance*. 71(6): 2933–2966.
- Cvijanovic, D., A. Dasgupta, and K. E. Zachariadis. 2019. "The Wall Street Stampede: Exit as Governance with Interacting Blockholders". *Working Paper*.
- Dasgupta, A., and G. Piacentino. 2015. "The Wall Street Walk When Blockholders Compete for Flows". *Journal of Finance.* 70(6): 2853–2896.
- Davis, G. F., and E. H. Kim. 2007. "Business Ties and Proxy Voting by Mutual Funds". *Journal of Financial Economics*. 85(2): 552–570.
- Del Guercio, D., L. Seery, and T. Woidtke. 2008. "Do Boards Pay Attention When Institutional Investor Activists "Just Vote No"?". *Journal of Financial Economics.* 90(1): 84–103.
- Denes, M. R., J. M. Karpoff, and V. B. McWilliams. 2017. "Thirty Years of Shareholder Activism: A Survey of Empirical Research". *Journal of Corporate Finance*. 44: 405–424.
- Dennis, P. J., K. Gerardi, and C. Schenone. 2019. "Common Ownership Does Not Have Anti-Competitive Effects in the Airline Industry". *Working Paper*.
- Derrien, F., A. Kecskés, and D. Thesmar. 2013. "Investor Horizons and Corporate Policies". *Journal of Financial and Quantitative Analysis.* 48(6): 1755–1780.
- Dimmock, S. G., W. C. Gerken, Z. Ivković, and S. J. Weisbenner. 2018. "Capital Gains Lock-in and Governance Choices". *Journal of Financial Economics*. 127(1): 113–135.
- Dimson, E., O. Karakaş, and X. Li. 2015. "Active Ownership". Review of Financial Studies 28(12): 3225–3268.
- Dimson, E., O. Karakaş, and X. Li. 2020. "Coordinated Engagements". Working Paper.
- Doidge, C., A. Dyck, H. Mahmudi, and A. Virani. 2019. "Collective Action and Governance Activism". *Review* of Finance. 23(5): 893–933.

- DoL. 1988. "Letter from the Department of Labor to Helmut Fandl, Chairman of the Retirement Board of Avon Products, Inc.".
- ———. 1990. "Letter from the Department of Labor to Robert A.G. Monks, Institutional Shareholder Services, Inc.".
- ----. 1994. "Interpretive Bulletins Relating to the Employee Retirement Income Security Act of 1974".
- ----. 2007. "US Department of Labor Opinion: Response to Thomas J. Donohue".
- Dressler, E. 2020. "Voice and Power: Do Institutional Shareholders Make Use of Their Voting Power?". Journal of Corporate Finance. 65(1): 101716.
- Duan, Y., and Y. Jiao. 2016. "The Role of Mutual Funds in Corporate Governance: Evidence from Mutual Funds' Proxy Voting and Trading Behavior". *Journal of Financial and Quantitative Analysis.* 51(2) 489–513.
- Dyck, A., K. V. Lins, L. Roth, and H. F. Wagner. 2019. "Do Institutional Investors Drive Corporate Social Responsibility? International Evidence". *Journal of Financial Economics*. 131(3): 693–714.
- Edmans, A. 2009. "Blockholder Trading, Market Efficiency, and Managerial Myopia". *Journal of Finance*. 64(6): 2481–2513.
- — . 2014. "Blockholders and Corporate Governance". Annual Review of Financial Economics. 6(1): 23– 50.
- Edmans, A., I. Goldstein, and W. Jiang. 2012. "The Real Effects of Financial Markets: The Impact of Prices on Takeovers". *Journal of Finance*. 67(3): 933–971.
- Edmans, A., V. W. Fang, and E. Zur. 2013. "The Effect of Liquidity on Governance". *Review of Financial Studies*. 26(6): 1443–1482.
- Edmans, A., X. Gabaix, and D. Jenter. 2017. "Executive Compensation: A Survey of Theory and Evidence". *Handbook of the Economics of Corporate Governance*. 1: 383–539.
- Edmans, A., and C. G. Holderness. 2017. "Blockholders: A Survey of Theory and Evidence". *Handbook of the Economics of Corporate Governance*. 1: 541–636.
- Edmans, A., D. Levit, and D. Reilly. 2019. "Governance under Common Ownership". *Review of Financial Studies*. 32 (7): 2673–2719.
- Edmans, A., and G. Manso. 2011. "Governance through Trading and Intervention: A Theory of Multiple Blockholders". *Review of Financial Studies*. 24(7): 2395–2428.
- Elton, E. J., M. J. Gruber, and C. R. Blake. 2003. "Incentive Fees and Mutual Funds". *Journal of Finance*. 58(2): 779–804.

- Erkens, D. H., M. Hung, and P. Matos. 2012. "Corporate Governance in the 2007–2008 Financial Crisis: Evidence from Financial Institutions Worldwide". *Journal of Corporate Finance*. 18(2): 389–411.
- Ertimur, Y., F. Ferri, and V. Muslu. 2011. "Shareholder Activism and CEO Pay". *Review of Financial Studies*. 24(2): 535–592.
- Ertimur, Y., F. Ferri, and D. Oesch. 2013. "Shareholder Votes and Proxy Advisors: Evidence from Say on Pay". *Journal of Accounting Research*. 51(5): 951–996.
- Faure-Grimaud, A., and D. Gromb. 2004. "Public Trading and Private Incentives". *Review of Financial Studies*. 17(4): 985–1014.
- Fernandes, N., M. A. Ferreira, P. Matos, and K. J. Murphy. 2013. "Are US CEOs Paid More? New International Evidence". *Review of Financial Studies*. 26(2): 323–367.
- Ferreira, M. A., M. Massa, and P. Matos. 2010. "Shareholders at the Gate? Institutional Investors and Cross-Border Mergers and Acquisitions". *Review of Financial Studies*. 23(2): 601–644.
- Ferreira, M. A., and P. Matos. 2008. "The Colors of Investors' Money: The Role of Institutional Investors around the World". *Journal of Financial Economics.* 88(3): 499–533.
- Fich, E. M., J. Harford, and A. L. Tran. 2015. "Motivated Monitors: The Importance of Institutional Investors' Portfolio Weights". *Journal of Financial Economics*. 118(1): 21–48.
- Filali Adib, F. Z. 2020. "Passive Aggressive: How Index Funds Vote on Corporate Governance Proposals". Working Paper.
- Flammer, C. 2015. "Does Corporate Social Responsibility Lead to Superior Financial Performance? A Regression Discontinuity Approach". *Management Science*. 61(11): 2549–2568.
- Fos, V. 2017. "The Disciplinary Effects of Proxy Contests". *Management Science*. 63(3): 655–671.
- Franks, J. 2020. "Institutional Ownership and Governance". *Oxford Review of Economic Policy*. 36(2): 258–274.
- Fung, W., and D. A. Hsieh. 1999. "A Primer on Hedge Funds". Journal of Empirical Finance. 6(3): 309–331.
- Gallagher, D. R., P. A. Gardner, and P. L. Swan. 2013. "Governance through Trading: Institutional Swing Trades and Subsequent Firm Performance". *Journal of Financial and Quantitative Analysis.* 48(2): 427–458.
- Gantchev, N. 2013. "The Costs of Shareholder Activism: Evidence from a Sequential Decision Model". Journal of Financial Economics. 107(3): 610–631.
- Gantchev, N., O. Gredil, and P. Jotikasthira. 2019. "Governance under the Gun: Spillover Effects of Hedge Fund Activism". *Review of Finance*. 23(6): 1031–1068.

- Gantchev, N., and C. Jotikasthira. 2018. "Institutional Trading and Hedge Fund Activism". *Management Science*. 64(6): 2930–2950.
- Gaspar, J.-M., M. Massa, and P. Matos. 2005. "Shareholder Investment Horizons and the Market for Corporate Control". *Journal of Financial Economics*. 76(1): 135–165.
- Gaspar, J.-M., M. Massa, P. Matos, R. Patgiri, and Z. Rehman. 2013. "Payout Policy Choices and Shareholder Investment Horizons". *Review of Finance*. 17(1): 261–320.
- Geng, H., H. Hau, and S. Lai. "Does Shareholder Overlap Alleviate Patent Holdup?". Working Paper.
- Giannetti, M., and X. Yu. 2020. "Adapting to Radical Change: The Benefits of Short-Horizon Investors". *Management Science*. forthcoming.
- Gibson, R., S. Glossner, P. Krueger, P. Matos, and T. Steffen. 2020. "Responsible Institutional Investing around the World". *Working Paper*.
- Gilje, E.P., T.A. Gormley, and D. Levit. 2020. "Who's Paying Attention? Measuring Common Ownership and Its Impact on Managerial Incentives". *Journal of financial Economics.* 137(1): 152–178.
- Gillan, S. L., and L. T. Starks. 2000. "Corporate Governance Proposals and Shareholder Activism: The Role of Institutional Investors". *Journal of financial Economics.* 57(2): 275–305.
- ———. 2007. "The Evolution of Shareholder Activism in the United States". *Journal of Applied Corporate Finance*. 19(1): 55–73.
- Ginglinger, E., C. Hebert, and L. Renneboog. 2017. "Connected Firms and Investor Myopia". Working Paper.
- Goldman, E., and G. Strobl. 2013. "Large Shareholder Trading and the Complexity of Corporate Investments". *Journal of Financial Intermediation*. 22(1): 106–122.
- Graham, J. R., and C. R. Harvey. 2001. "The Theory and Practice of Corporate Finance: Evidence from the Field". *Journal of Financial Economics*. 60(2–3): 187–243.
- Graham, J. R., C. R. Harvey, and M. Puri. 2015. "Capital Allocation and Delegation of Decision-Making Authority within Firms". *Journal of Financial Economics.* 115(3): 449–470.
- Greenwood, R., and M. Schor. 2009. "Investor activism and takeovers". *Journal of Financial Economics*. 92(3): 362-375.
- Grinstein, Y., and R. Michaely. 2005. "Institutional Holdings and Payout Policy". *Journal of Finance*. 60(3): 1389–1426.
- Grossman, S. J., and O. D. Hart. 1980. "Takeover Bids, the Free-Rider Problem, and the Theory of the Corporation". *Bell Journal of Economics*. 11(1) 42-64.
- ———. 1988. "One Share-One Vote and the Market for Corporate Control". Journal of Financial Economics. 20(1-2): 175–202.

- Guest, P. M., and M. Nerino. 2020. "Do Corporate Governance Ratings Change Investor Expectations? Evidence from Announcements by Institutional Shareholder Services". *Review of Finance.* 24 (4): 891–928.
- Gupta, A., and K. Sachdeva. 2019. "Skin or Skim? Inside Investment and Hedge Fund Performance". *Working Paper.*
- Gutiérrez, G., and T. Philippon. 2018. "Ownership, Concentration, and Investment". In American Economic Review Papers and Proceedings. 108: 432–37.
- Hamdani, A., and Y. Yafeh. 2013. "Institutional Investors as Minority Shareholders". *Review of Finance*. 17(2): 691–725.
- Harford, J., A. Kecskés, and S. Mansi. 2018. "Do Long-Term Investors Improve Corporate Decision Making?". *Journal of Corporate Finance*. 50: 424–452.
- Hartzell, J. C., and L. T. Starks. 2003. "Institutional Investors and Executive Compensation". *Journal of Finance*. 58 (6): 2351–2374.
- He, Y., B. Kahraman, and M. Lowry. 2020. "ES Risks and Shareholder Voice". Working Paper.
- He, Z., and A. Krishnamurthy. 2013. "Intermediary Asset Pricing". *American Economic Review*. 103(2): 732–770.
- Heath, D., D. Macciocchi, R. Michaely, and M. Ringgenberg. 2021. "Do Index Funds Monitor?". *Review of Financial Studies*. forthcoming.
- Heinen, V., C. Koch, and M. Scharfbillig. 2018. "Exporting Corporate Governance: Do Foreign and Local Proxy Advisors Differ?". *Working Paper*.
- Helwege, J., V. J. Intintoli, and A. Zhang. 2012. "Voting with Their Feet or Activism? Institutional Investors' Impact on CEO Turnover". *Journal of Corporate Finance*. 18(1): 22–37.
- Hitz, J.-M., and N. Lehmann. 2018. "Empirical Evidence on the Role of Proxy Advisors in European Capital Markets". *European Accounting Review*. 27(4): 713–745.
- Hoepner, A. G. F., I. Oikonomou, Z. Sautner, L. T. Starks, X. Zhou, Xiaoyan. 2021. "ESG Shareholder Engagement and Downside Risk". *Working Paper*.
- Holderness, C. G., and D. P. Sheehan. 1985. "Raiders or saviors? The Evidence on Six Controversial Investors". *Journal of Financial Economics.* 14(4): 555–579.
- Hornstein, J. A. 1993. "Proxy Solicitation Redefined: The SEC Takes an Incremental Step Toward Effective Corporate Governance". *Washington University Law Quarterly* 71(4): 1129–1163.

- Hu, H. T. C., and B. Black. 2007. "Hedge Funds, Insiders, and the Decoupling of Economic and Voting Ownership: Empty Voting and Hidden (Morphable) Ownership". *Journal of Corporate Finance*. 13(2-3): 343–367.
- Hu, G., K. M. Jo, Y. A. Wang, and J. Xie. 2018. "Institutional Trading and Abel Noser Data". *Journal of Corporate Finance*. 52: 143–167.
- Ilhan, E., P. Krueger, Z. Sautner, and L. T. Starks. 2021. "Climate Risk Disclosure and Institutional Investors". *Working Paper*.
- Iliev, P., and M. Lowry. 2015. "Are Mutual Funds Active Voters?". *Review of Financial Studies*. 28(2): 446–485.
- Iliev, P., J. Kalodimos, and M. Lowry. 2021. "Investors' Attention to Corporate Governance". *Review of Financial Studies*. forthcoming.
- Ishida, S., and T. Kochiyama. 2020. "ISS's Proxy Voting Guidelines and ROE Management". Working Paper.
- Israelsen, R. D., M. Schwartz-Ziv, and J. Watson. 2019. "Ownership and Governance Style: New Evidence from Nonfinancial Blockholders". *Working Paper.*
- Jensen, M. C. 1986. "Agency Costs of Free Cash Flow, Corporate Finance, and Takeovers". *American Economic Review Papers and Proceedings*. 76(2): 323–329.
- Jiang, W., K. L. Li, and W. Wang. 2012. "Hedge Funds and Chapter 11". Journal of Finance. 67(2): 513–560.
- Kahn, C., and A. Winton. 1998. "Ownership Structure, Speculation, and Shareholder Intervention?". Journal of Finance. 53(1): 99–129.
- Kang, J.-K., J. Luo, and H. S. Na. 2018. "Are Institutional Investors with Multiple Blockholdings Effective Monitors?". *Journal of Financial Economics*. 128(3): 576–602.
- Karolyi, G. A., D. Kim, and R. Liao. 2020. "The Theory and Practive of Investor Relations: A Global Perspective". *Management Science* 66(10): 4359–4919.
- Karpoff, J. M., P. H. Malatesta, and R. A. Walkling. 1996. "Corporate Governance and Shareholder Initiatives: Empirical Evidence". *Journal of Financial Economics.* 42 (3): 365–395.
- Katelouzou, D. and M. Siems. 2020. "The Global Diffusion of Stewardship Codes". Working Paper.
- Kedia, S., L. T. Starks, and X. Wang. 2021. "Institutional Investors and Hedge Fund Activism". *Review of Corporate Finance Studies*. 10(1): 1–43.
- Kempf, E., A. Manconi, and O. Spalt. 2017. "Distracted Shareholders and Corporate Actions". *Review of Financial Studies*. 30(5): 1660–1695.
- Khorona, A., H. Servaes, and L. Wedge. 2007. "Portfolio Manager Ownership and Fund Performance". *Journal of Financial Economics*. 85(1): 179–204.

- Kim, I., S. Miller, H. Wan, and B. Wang. 2016. "Drivers behind the Monitoring Effectiveness of Global Institutional Investors: Evidence from Earnings Management". *Journal of Corporate Finance*. 40: 24–46.
- Klein, A., and E. Zur. 2009. "Entrepreneurial Shareholder Activism: Hedge Funds and Other Private Investors". *Journal of Finance*. 64(1): 187–229.
- Krueger, P., Z. Sautner, and L. T. Starks. 2020. "The Importance of Climate Risks for Institutional Investors". *Review of Financial Studies* 33(3): 1067–1111.
- Kyle, A. S., and J. L. Vila. 1991. "Noise Trading and Takeovers". *RAND Journal of Economics*. 22(1): 54–71.
- Larcker, D. F., A. L. McCall, and G. Ormazabal. 2015. "Outsourcing Shareholder Voting to Proxy Advisory Firms". *Journal of Law and Economics*. 58(1): 173–204.
- Levit, D. 2019. "Soft Shareholder Activism". Review of Financial Studies. 32(7): 2775–2808.
- Levit, D., N. Malenko, and E. Maug. 2021. "Trading and Shareholder Democracy". Working paper.
- Levit, D., and A. Tsoy. 2021. "A Theory of One-Size-Fits-All Recommendations". *American Economic Journal: Microeconomics*. forthcoming.
- Leuz, C., K. V. Lins, and F. E. Warnock. 2008. "Do Foreigners Invest Less in Poorly Governed Firms?". *Review* of Financial Studies. 22(8): 3245–3285.
- Lewellen, J., and K. Lewellen. 2021. "Institutional Investors and Corporate Governance: The Incentive to Be Engaged". *Journal of Finance*. forthcoming.
- Lewellen, K., and M. Lowry. 2021. "Does Common Ownership Really Increase Firm Coordination". *Journal* of Financial Economics. forthcoming.
- Li, T. 2018. "Outsourcing Corporate Governance: Conflicts of Interest within the Proxy Advisory Industry". Management Science. 64(6): 2951–2971.
- Lim, J., B. A. Sensoy, and M. S. Weisbach. 2016. "Indirect Incentives of Hedge Fund Managers". *Journal of Finance*. 71(2): 871–918.
- Liu, C., A. Low, R. Masulis, and L. Zhang. 2020. "Monitoring the Monitor: Distracted Institutional Investors and Board Governance". *Review of Financial Studies*. 33(10): 4489–4531.
- Luong, H., F. Moshirian, L. Nguyen, X. Tian, and B. Zhang. 2017. "How Do Foreign Institutional Investors Enhance Firm Innovation?". *Journal of Financial and Quantitative Analysis.* 52(4): 1449–1490.
- Macey, J. R., and J. M. Netter. 1987. "Regulation 13D and the Regulatory Process". *Washington University Law Quarterly*. 65(1): 131–161.
- Malenko, A., and N. Malenko. 2019. "Proxy Advisory Firms: The Economics of Selling Information to Voters". *Journal of Finance*. 74(5): 2441–2490.

- Malenko, N., and Y. Shen. 2016. "The Role of Proxy Advisory Firms: Evidence from a Regression-Discontinuity Design". *Review of Financial Studies*. 29(12): 3394–3427.
- Matos, P. 2020. ESG and Responsible Institutional Investing Around the World: A Critical Review. Working Paper.
- Matvos, G., and M. Ostrovsky. 2010. "Heterogeneity and Peer Effects in Mutual Fund Proxy Voting". *Journal of Financial Economics*. 98(1): 90–112.
- Maug, E. 1998. "Large Shareholders as Monitors: Is there a trade-off between liquidity and control?". *Journal of Finance*. 53(1): 65–98.
- McCahery, J. A., Z. Sautner, and L. T. Starks. 2016. "Behind the Scenes: The Corporate Governance Preferences of Institutional Investors". *Journal of Finance*. 71(6): 2905–2932.
- McRitchie, J. 2014. "Fiduciary Duty to Announce Votes (Part 2): Historical Background". Retrieved from https://www.corpgov.net/2014/05/fiduciary-duty-announce-votes-part-2-historical-background/
- Michaely, R., and C. Vincent. 2013. "Do Institutional Investors Influence Capital Structure Decisions?". *Working Paper.*
- Miyajima, H., and T. Hoda. 2015. "Ownership Structure and Corporate Governance: Has an Increase in Institutional Investors' Ownership Improved Business Performance?". *Policy Research Institute, Ministry of Finance, Japan, Public Policy Review.* 11(3): 361–393.
- Morley, J. 2019. "Too Big to Be Activist". Southern California Law Review. 92(6): 1407–1454.
- Mullins, W. 2014. "The Governance Impact of Index Funds: Evidence from Regression Discontinuity". *Working Paper*.
- Ng, L., F. Wu, J. Yu, and B. Zhang. 2016. "Foreign Investor Heterogeneity and Stock Liquidity around the World". *Review of Finance*. 20(5): 1867–1910.
- Parrino, R., R. W. Sias, and L. T. Starks. 2003. "Voting with Their Feet: Institutional Ownership Changes around Forced CEO Turnover". *Journal of Financial Economics*. 68(1): 3–46.
- Pezier, E. 2020. "Estimating the Long-Term Effects of Activism Using a Natural Experiment in UK Small-Cap Stocks". *Working Paper*.
- Piacentino, G. 2019. "Venture Capital and Capital Allocation". Journal of Finance. 74(3): 1261–1314.
- Ringe, W. G. 2020. "Stewardship and Shareholder Engagement in Germany". *European Business* Organization Law Review. forthcoming.
- Rock, E. 2018. "Institutional Investors in Corporate Governance". Oxford Handbook of Corporate Law and Governance.
- Roe, M. J. 1991. "A Political Theory of American Corporate Finance". Columbia Law Review. 91(1): 10–67.

- Schiffman, H. 1976. "The Relationship Between the Investment Advisor and the Mutual Fund: Too Close for Comfort". *Fordham Law Review*. 45: 183-202.
- Schmalz, M. C. 2018. "Common-Ownership Concentration and Corporate Conduct". Annual Review of Financial Economics. 10(1): 413–448.
- Schmidt, C., and R. Fahlenbrach. 2017. "Do Exogenous Changes in Passive Institutional Ownership Affect Corporate Governance and Firm Value?". *Journal of Financial Economics*. 124(2): 285–306.
- Schwartz-Ziv, M., and E. Volkova. 2020. "Is Blockholder Diversity Detrimental?". Working Paper.
- SEC. 2003a. "Disclosure of Proxy Voting Policies and Proxy Voting Records by Registered Management Investment Companies". *17 CFR Parts 239, 249, 270, and 274*.
- ----. 2003b. "Proxy Voting by Investment Advisers". 17 CFR Part 275.
- ———. 2019a. "Commission Interpretation and Guidance Regarding the Applicability of the Proxy Rules to Proxy Voting Advice (a)". *17 CFR Part 241*.
- — 2019b. "Commission Guidance Regarding Proxy Voting Responsibilities of Investment Advisers (b)". 17 CFR Parts 271 and 276.
- Shadab, H. B. 2009. "The Law and Economics of Hedge Funds: Financial Innovation and Investor Protection". *Berkeley Business Law Journal* 6(2): 240–297.
- Shleifer, A., and R. W. Vishny. 1986. "Large Shareholders and Corporate Control". *Journal of Political Economy.* 94(3): 461–488.
- Shu, C. 2020. "Proxy Advice Industry and Its Growing Influence". Working Paper.
- Smith, M. P. 1996. "Shareholder Activism by Institutional Investors: Evidence from CalPERS". *Journal of Finance*. 51(1): 227–252.
- Song, F. 2017. "Blockholder short-term incentives, structures, and governance". *Working Paper*.
- Starks, L. T., P. Venkat, and Q. Zhu. 2021. "Corporate ESG Profiles and Investor Horizons". Working Paper.
- Tett, G. "Proxy Advisers Must Embrace Calls for Reform". *Financial Times*, retrieved from: https://www.ft.com/content/d6c79acc-3039-11e9-ba00-0251022932c8.
- Villalonga, B., R. Amit, M. A. Trujillo and A. Guzmán. 2015. "Governance of family firms". *Annual Review* of Financial Economics. 7(1): 635-654.
- Wardlaw, M. 2020. "Measuring mutual fund flow pressure as shock to stock returns". *Journal of Finance*. 75(6): 3221–3243.
- Webber, D. 2014. "The Use and Abuse of Labor's Capital". *New York University Law Review*. 89(6): 2106–2189.

- Wei, W., and A. Young. 2019. "Selection Bias or Treatment Effect? A Re-Examination of Russell 1000/2000 Index Reconstitution". *Working Paper*.
- Winton, A. 1993. "Limitation of Liability and the Ownership Structure of the Firm". *Journal of Finance*. 48(2): 487–512.
- Woidtke, T. 2002. "Agents Watching Agents?: Evidence from Pension Fund Ownership and Firm Value". Journal of Financial Economics. 63(1): 99–131.
- Yermack, D. 2010. "Shareholder Voting and Corporate Governance". *Annual Review of Financial Economics*. 2(1): 103–125.
- Zachariadis, K. E., and I. F. Olaru. 2017. "The Impact of Security Trading on Corporate Restructuring". *Review of Finance*. 21(2): 667–718.

| Sector | 1950 | 1970 | 1990 | 2000 | 2010 | 2020 |
|---|------|------|------|------|------|------|
| Private Pension Funds | 0.0 | 8.1 | 16.2 | 11.2 | 7.8 | 5.4 |
| Federal, State and Government Pension Funds | 0.0 | 1.2 | 8.1 | 7.7 | 8.5 | 5.3 |
| Insurance Companies | 2.6 | 3.3 | 4.1 | 6.2 | 6.7 | 1.9 |
| Mutual Funds | 1.6 | 4.8 | 7.1 | 18.3 | 20.2 | 20.8 |
| Closed-End Funds | 0.9 | 0.5 | 0.5 | 0.2 | 0.4 | 0.2 |
| Exchange-Traded Funds | 0.0 | 0.0 | 0.0 | 0.4 | 3.6 | 6.6 |
| Foreign Sector | 1.6 | 3.3 | 6.9 | 9.3 | 13.7 | 16.4 |
| Household Sector | 92.8 | 78.2 | 56.5 | 45.6 | 37.2 | 38.3 |
| Other | 0.4 | 0.6 | 0.7 | 1.1 | 1.8 | 5.1 |

Table 1: Ownership Shares of the US Stock Market in Percent

Source: Federal Reserve Statistical Release Data: Flow of Funds Data United States. Exchange-Traded Funds are first listed in December 7, 2001. The Household Sector includes Bank Personal Trusts.

| | Ν | Mean | STD | 10 th | Median | 90 th |
|--|-----------------|-------|-------|------------------|--------|------------------|
| | | | | percentile | | percentile |
| Panel A: Full Sample | | | | | | |
| Number of 1%-blocks | 353,729 | 11.19 | 11.11 | 1.00 | 9.00 | 23.00 |
| Number of 3%-blocks | 353,729 | 4.33 | 6.26 | 0.00 | 3.00 | 9.00 |
| Number of 5%-blocks | 353,729 | 2.40 | 4.77 | 0.00 | 1.00 | 5.00 |
| Number of 10%-blocks | 353,729 | 0.90 | 3.21 | 0.00 | 0.00 | 2.00 |
| Panel B: Hedge Fund Blocks | | | | | | |
| Number of 1%-blocks | 353,729 | 2.64 | 2.71 | 0.00 | 2.00 | 6.00 |
| Number of 3%-blocks | 353,729 | 1.15 | 1.67 | 0.00 | 1.00 | 3.00 |
| Number of 5%-blocks | 353,729 | 0.68 | 1.31 | 0.00 | 0.00 | 2.00 |
| Number of 10%-blocks | 353,729 | 0.23 | 0.86 | 0.00 | 0.00 | 1.00 |
| Panel C: Mutual Fund Blocks | (company-level) | | | | | |
| Number of 1%-blocks | 865,963 | 2.48 | 3.25 | 0.00 | 1.00 | 7.00 |
| Number of 3%-blocks | 865,963 | 0.64 | 1.13 | 0.00 | 0.00 | 2.00 |
| Number of 5%-blocks | 865,963 | 0.29 | 0.70 | 0.00 | 0.00 | 1.00 |
| Number of 10%-blocks | 865,963 | 0.03 | 0.19 | 0.00 | 0.00 | 0.00 |
| Panel D: Mutual Fund Blocks (fund-level) | | | | | | |
| Number of 1%-blocks | 865,963 | 2.32 | 3.16 | 0.00 | 1.00 | 7.00 |
| Number of 3%-blocks | 865,963 | 0.31 | 0.67 | 0.00 | 0.00 | 1.00 |
| Number of 5%-blocks | 865,963 | 0.11 | 0.36 | 0.00 | 0.00 | 0.00 |
| Number of 10%- blocks | 865,963 | 0.01 | 0.09 | 0.00 | 0.00 | 0.00 |

Table 2: Institutional Blockholders in the US

Source: In Panels A and B, the Schedule 13F ownership data that is underlying the measures in this table covers the 1999-2017 period and is obtained from Michael Sinkinson's website (see Backus *et al.*, 2021). In Panel B, the analysis is limited to Schedule 13F filers who are identified as hedge funds in Agarwal *et al.* (2013). In Panels C and D, mutual fund ownership data covers the 1999-2017 period and is obtained from the CRSP Survivor-Bias-Free US Mutual Fund Database. Panel C reports the results at mutual fund company level and Panel D at the mutual fund level. The unit of observation is at the security-reporting quarter.

Table 3: Governance Obligations of Institutional Investors Globally

| | | Setting of Voting Policy | Disclosure of Voting Policy | Disclosure of Voting record | Setting Policy for Conflict of Interest | Disclosure for Conflict of Interest | Monitoring | Dialogue with Investee | Maintain supervision even if outsourced |
|-------------------|--|--------------------------------|--------------------------------------|-----------------------------------|--|---|------------|------------------------------|--|
| US | Investment Companies Private Pension Fund | Law | Law | Law | Law | Law | Law | | Law Law |
| | Investment Advisers | Law | Law | Law | Law | Law | Law | | Law |
| China | National Social Security Funds | Industry Assoc. | Code | Code | | | | | |
| | Pension Funds | Industry Assoc. | Code | Code | | | | | |
| | Insurance Funds | Industry Assoc. | Code | Code | | | | | |
| | Public Offering Funds | Industry Assoc. | Code | Code | | | | | |
| Japan | Institutional Investors | Code | Code | Code | Code | Code | Code | Code | Code |
| | Proxy Advisor | Code | Code | Code | Code | Code | Code | Code | Code |
| Germany | Investment Funds/Asset Managers | Code | Law, Code | | Law <i>,</i> Code | | Code | Code | Law, Code |
| United Kingdom | Institutional Investors | Code | Code | Code | Code | Code | Code | Code | Code |
| | Service Provider | Code | Code | Code | Code | Code | Code | Code | Code |
| India | Mutual Funds and Asset Managers | Law | Law | Law | Law | | | | |
| | Insurers | Code | Code | Code | Code | Code | Code | Code | Code |
| | Pension Funds | Law | Law | Law | Law | Law | Law | Law | Law |

Source: OECD Corporate Governance Factbook 2019. "Code" may also refer to codes developed by private initiatives.

Table 4: Proxy Access Rules Globally

| | China | Japan | Germany | UK | India |
|--------------|----------------|----------------|----------------|------------------|---------------|
| Minimum | 3% | 1% or 300 | 5% or €500,000 | 5% or 100 | 10% of voting |
| Holding | | voting rights | | shareholders | share |
| Required | | with 6 months | | holding | |
| | | holdings | | together > | |
| | | | | GBP10,000 | |
| Deadline for | 10 days before | 8 weeks before | 10 days before | 7 weeks before | 21-45 days |
| Request | the meeting | meeting | the meeting | meeting | before the |
| | | | | | meeting |
| Deadline for | Two days to | | 14 days for | Company has to | 21 days |
| Company to | accept and | | accepting and | publish at same | |
| Respond | publish | | publishing | time as it gives | |
| | | | | notice of the | |
| | | | | meeting | |
| | | | | (Source: UK | |
| | | | | Companies Act | |
| | | | | 2006, 314 (2), | |
| | | | | 315 (1) b) | |

Source: OECD Corporate Governance Factbook 2019, unless otherwise stated.

Table 5: Disclosure Rules Globally

| | Rules to Disclose Majority Shareholdings |
|---------|---|
| China | Only for issuer listed on Shanghai or Shenzhen stock exchange Disclosure triggered when a shareholding reaches or exceeds 5% of total issued shares of an issuer, then: 5%-20%: basic disclosure 20%-30%: detailed disclosure >30%: tender offer disclosure report Lock-in periods for shareholders: 5% or more shareholders restricted from selling shares for period of 6 months from purchase and from purchasing shares in issuer for period of 6 months from its last sale of such shares Additional restriction selling shares: Aggregate divestment on open market within 3 months cannot exceed 1% of total shares of issuer Aggregate divestment through block trading within 90 days cannot exceed 2% of total shares of the issuer |
| Japan | Only for companies that are listed on stock exchange in Japan Disclosure concerned with "ownership" of "Target Securities" and disclosure triggered when: "Ratio of Shareholdings" (calculated by specific formula) of a person exceeds 5% or, Ratio of Shareholdings of large shareholder changes by 1% or more, or There is material change to matters described in previously filed report Special Reporting with lower frequency for certain institutional investors trading securities without control intent Need to report within 5 business days from every base date (occurs at least bimonthly) |
| Germany | Only for issuer trading on EEA regulated market and whose home member state is Germany (separate disclosure obligation of holdings of ≥25% in shares of German stock corporation which is not listed) Disclosure thresholds met from holding either voting shares or financial instruments referenced to voting shares or a combination Disclosure triggered for: 3% (only for shares not financial instruments), 5%, 10%, 15%, 20%, 25%, 30%, 50% and 75% |

| | ○ If holder of previously reported financial instruments exercises and acquires underlying shares and shares threshold is met but overall % held is unchanged → new nature of holdings needs to be disclosed |
|-------|--|
| UK | Only for UK incorporated or non-UK incorporated issuer whose shares admitted to trading on EEA regulated market and whose member state is the UK or UK incorporated issuer whose shares admitted to trading on UK prescribed market Disclosure thresholds met either voting shares or financial instruments referenced to voting shares or a combination Disclosure triggered for: UK-incorporated company: 3% of total voting rights and each whole percentage point after that Non-UK incorporated issuer: 5%, 10%, 15%, 20%, 25%, 30%, 50% and 75% of total voting rights If holder of previously reported financial instruments exercises and acquires underlying shares and shares threshold is met but overall % held is unchanged → new nature of holdings needs to be disclosed |
| India | Only for issuer listed on recognized stock exchange in India Disclosure triggered for: Person (or group) acquiring ≥5% of total shares or voting rights of company Disclosure required if change in holding of ≥5% holder and change exceeds 2% of total shares or voting rights |

Source: AOSphere (Allen & Overy)





Source: The Schedule 13F ownership data that is underlying the measures in this figure covers the 1999-2017 period and is obtained from Michael Sinkinson's website (Backus *et al.*, 2021). The blue line represents 1%-blocks (i.e., cases when an institution owns at least 1% of shares outstanding), the orange line represents 3%-blocks, the gray line represents 5%-blocks, and the yellow line represents 10%-blocks.



Figure 2: Time-series Evolution of Hedge Fund block Ownership in the US

Source: The list of Schedule 13F filers who are identified as hedge funds is from Agarwal *et al.* (2013). The blue line represents 1%-blocks (i.e., cases when a hedge fund owns at least 1% of shares outstanding), the orange line represents 3%-blocks, the gray line represents 5%-blocks, and the yellow line represents 10%-blocks.





Source: Mutual fund holdings data are from the CRSP Survivor-Bias-Free US Mutual Fund Database. The blue line represents 1%-locks by mutual funds (i.e., cases when a mutual fund owns at least 1% of shares outstanding), the orange line represents 3%-blocks, the gray line represents 5%-blocks, and the yellow line represents 10%-blocks.



Figure 4: Time-series Evolution of Five Largest Institutional Blockholders

Source: The Schedule 13F ownership data that is underlying the measure in this figure covers the 1999-2017 period and is obtained from Michael Sinkinson's website (Backus *et al.*, 2021). The dark line represents the percentage of shares outstanding owned by five largest institutional blockholders in a firm.



Figure 5: Time-series Evolution of Big Three Institutional Ownership

Source: The Schedule 13F ownership data that is underlying the measure in this figure covers the 1999-2017 period and is obtained from Michael Sinkinson's website (Backus *et al.*, 2021). The dark line represents the percentage of shares outstanding owned by the "Big Three," that is, BlackRock, State Street, and Vanguard.



Figure 6: Time-series Evolution of Institutional Ownership around the World

Source: FactSet Ownership. The blue line represents institutional ownership in UK firms, the grey line represents institutional ownership in German firms, the green line represents institutional ownership in Indian firms, the yellow line represents institutional ownership in Japanese firms, and the orange line represents institutional ownership in Chinese firms.



Figure 7: Time-series Evolution of Foreign Institutional Ownership around the World

Source: FactSet Ownership. The blue line represents foreign institutional ownership in UK firms, the grey line represents foreign institutional ownership in German firms, the green line represents foreign institutional ownership in Indian firms, the yellow line represents foreign institutional ownership in Japanese firms, and the orange line represents foreign institutional ownership in Chinese firms.



Figure 8: International Data: Foreign to Total Institutional Ownership over Time

Source: FactSet Ownership. The blue line represents the fraction of foreign institutional ownership (relative to total institutional ownership) in UK firms, the grey line represents the same fraction n German firms, the green line represents the same fraction in Indian firms, the yellow line represents the same fraction in Japanese firms, and the orange line represents the same fraction in Chinese firms.



Figure 9: Investments of Pension Funds into Mutual Funds in the US

Source: Federal Reserve Statistical Release Data: Flow of Funds Data United States. Investments of pension funds into mutual funds are calculated by dividing the market value of mutual fund shares held by pension funds (FRED Source ID: LM593064205.Q) by the total amount of pension funds' (financial) assets (FRED Source ID: FL594090005.Q). For each calendar year, we use the end of year data.

european corporate governance institute

about ECGI

The European Corporate Governance Institute has been established to improve *corpo*rate governance through fostering independent scientific research and related activities.

The ECGI will produce and disseminate high quality research while remaining close to the concerns and interests of corporate, financial and public policy makers. It will draw on the expertise of scholars from numerous countries and bring together a critical mass of expertise and interest to bear on this important subject.

The views expressed in this working paper are those of the authors, not those of the ECGI or its members.

www.ecgi.global

european corporate governance institute

ECGI Working Paper Series in Finance

| Editorial Board | |
|---------------------|--|
| Editor | Mike Burkart, Professor of Finance, London School of Economics and Political Science |
| Consulting Editors | Franklin Allen, Nippon Life Professor of Finance, Professor of Economics, The Wharton School of the University of Pennsylvania |
| | Julian Franks, Professor of Finance, London Business School |
| | Marco Pagano, Professor of Economics, Facoltà di Economia |
| | Università di Napoli Federico II |
| | Xavier Vives, Professor of Economics and Financial Management, IESE Business School, University of Navarra |
| | Luigi Zingales, Robert C. McCormack Professor of Entrepreneurship and Finance, University of Chicago, Booth School of Business |
| Editorial Assistant | Úna Daly, Working Paper Series Manager |

www.ecgi.global/content/working-papers

Electronic copy available at: https://ssrn.com/abstract=3682800

european corporate governance institute

Electronic Access to the Working Paper Series

The full set of ECGI working papers can be accessed through the Institute's Web-site (www.ecgi.global/content/working-papers) or SSRN:

| Finance Paper Series | http://www.ssrn.com/link/ECGI-Fin.html |
|----------------------|--|
| Law Paper Series | http://www.ssrn.com/link/ECGI-Law.html |

www.ecgi.global/content/working-papers